Theoretical and Practical Research in Economic Fields

Biannually

Volume VI Issue 1(11) Summer 2015

ISSN 2068 - 7710 Journal **DOI** http://dx.doi.org/10.14505/tpref





is an advanced e-publisher struggling to bring further worldwide learning, knowledge and research. This transformative mission is realized through our commitment to innovation and enterprise, placing

us at the cutting-edge of electronic delivery in a world that increasingly considers the dominance of digital content and networked access not only to books and journals but to a whole range of other pedagogic services.

In both books and journals, **ASERS Publishing** is a hallmark of the finest scholarly publishing and cutting-edge research, maintained by our commitment to rigorous peer-review process.

Using pioneer developing technologies, **ASERS Publishing** keeps pace with the rapid changes in the e-publishing market.

ASERS Publishing is committed to providing customers with the information they want, when they want and how they want it. To serve this purpose, ASERS publishing offers digital Higher Education materials from its journals, courses and scientific books, in a proven way in order to engage the academic society from the entire world.

Summer 2014 Volume VI Issue 1(11)

Editor in Chief

PhD Laura UNGUREANU Spiru Haret University, Romania

Editor

PhD Ivan KITOV Russian Academy of Sciences, Russia Editorial Advisory Board

Monal Abdel-Baki American University in Cairo,Egypt

Mădălina Constantinescu SpiruHaret University and ASERS Publishing, Romania

Jean-Paul Gaertner Ecole de Management de Strasbourg, France

Shankar Gargh Editor in Chief of Advanced in Management, India

Piotr Misztal The Jan Kochanowski University in Kielce, Faculty of Management and Administration, Poland

Russell Pittman International Technical Assistance Economic Analysis Group Antitrust Division, USA

Rachel Price-Kreitz Ecole de Management de Strasbourg, France

Rena Ravinder Politechnic of Namibia, Namibia

Andy Ștefănescu University of Craiova, Romania

Laura Gavrilă (formerly Ștefănescu) *SpiruHaret* University Romania

Hans-Jürgen Weißbach University of Applied Sciences - Frankfurt am Main, Germany

ASERS Publishing

http://www.asers.eu/asers-publishing ISSN 2068 – 7710 Journal's Issue DOI http://dx.doi.org/10.14505/tpref.v6.1(11).00

Contents:

Spatial Analysis of Income Growth in the 1 Philippines: Evidence from Intra-Country Data

Dennis S. MAPA, Manuel Leonard F. ALBIS, Dorcas Mae P.	5
COMANDANTE, Josephine D. CURA, Maureen P. LADAO	

2 Bubbles, Bluffs and Greed

Taiji HARASHIMA21

The Whole Economy Approach of the Input Output Model

	Fengquan LIU	50
4	Networks in Context: the Concept of Network in Manuel Castells' Theory of the	

....65

Ari-Veikko ANTTIROIKO

Network Society

3

Call for Papers Volume VI, Issue 2(12), Winter 2015

Theoretical and Practical Research in Economic Fields

Many economists today are concerned by the proliferation of journals and the concomitant labyrinth of research to be conquered in order to reach the specific information they require. To combat this tendency, **Theoretical and Practical Research in Economic Fields** has been conceived and designed outside the realm of the traditional economics journal. It consists of concise communications that provide a means of rapid and efficient dissemination of new results, models and methods in all fields of economic research.

Theoretical and Practical Research in Economic Fields publishes original articles in all branches of economics – theoretical and empirical, abstract and applied, providing wide-ranging coverage across the subject area.

Journal promotes research that aim at the unification of the theoretical-quantitative and the empirical-quantitative approach to economic problems and that are penetrated by constructive and rigorous thinking. It explores a unique range of topics from the frontier of theoretical developments in many new and important areas, to research on current and applied economic problems, to methodologically innovative, theoretical and applied studies in economics. The interaction between empirical work and economic policy is an important feature of the journal.

Theoretical and Practical Research in Economic Fields, starting with its first issue, it is indexed in EconLit, RePEC, EBSCO, ProQuest, Cabell Directories and CEEOL databases.

The primary aim of the Journal has been and remains the provision of a forum for the dissemination of a variety of international issues, empirical research and other matters of interest to researchers and practitioners in a diversity of subject areas linked to the broad theme of economic sciences.

All the papers will be first considered by the Editors for general relevance, originality and significance. If accepted for review, papers will then be subject to double blind peer review.

Invited manuscripts will be due till November 25th, 2015, and shall go through the usual, albeit somewhat expedited, refereeing process.

Deadline for submission of proposals:	25 th November, 2015
Expected publication date:	December, 2015
Website:	www.asers.eu/journals/tpref/
E-mail:	tpref@asers.eu

To prepare your paper for submission, please see full author guidelines in the following file: <u>TPREF_Full_Paper_Template.doc</u>, then send it via email at <u>tpref@asers.eu</u>.



DOI: http://dx.doi.org/10.14505/tpref.v6.1(11).01

SPATIAL ANALYSIS OF INCOME GROWTH IN THE PHILIPPINES. EVIDENCE FROM INTRA-COUNTRY DATA

Manuel Leonard F. ALBIS School of Statistics, University of the Philippines, Philippines Dennis S. MAPA School of Statistics, University of the Philippines, Philippines <u>cdsmapa@yahoo.com</u> Dorcas Mae P. COMANDANTE School of Statistics, University of the Philippines, Philippines Josephine D. CURA School of Statistics, University of the Philippines, Philippines Maureen P. LADAO School of Statistics, University of the Philippines, Philippines

Suggested Citation:

Albis, M.L.F., Mappa, D.S. (2015). Spatial analysis of income growth in the Philippines. Evidence from intracountry data. *Theoretical and Practical Research in Economic Field*, (Volume VI, Summer), 1(11): 5-20. DOI:10.14505/tpref.v6.1(11).02.Available from:http://www.asers.eu/journals/tpref/curent-issue.

Article's History:

Received April, 2015; *Revised* June, 2015; *Accepted* August, 2015. 2015. ASERS Publishing. All rights reserved.

Abstract:

This paper looks at the spatial relationship of the average per capita income growth using provincial data from 1988 to 2009. The results from the study provide insights on the geographical dimensions of provincial income growth and showed evidence on the role of spatial effects in the formal econometric analysis of intra-country income growth models. Despite data limitations, the study provides a strong empirical evidence of the presence of positive spatial dependence or degree of similarity in the average per capita income growth of the provinces, albeit the degree of positive spatial dependence weakens in the latter periods. This positive spatial correlation suggests the provinces may be converging in terms of their income growth and they do so in movements similar to their neighbors. Moreover, the study shows that spatial dependence weakened in the latter periods (1994-2000 and 2000-2009). The weakening of spatial dependence may provide insights on the uneven provincial/regional income growth experienced in the country. One possible explanation of the weak spatial dependence is that two or more groups of neighboring provinces are growing at similar rates within the group, but at different rates across groups. This opens the possibility of having different convergence clubs (of provinces) within the country.

Keywords: spatial dependence, Moran's index, intra-country growth mode.

1. Introduction

The growth rates of provincial per capita income in the Philippines in the last 25 years have been generally diverse: there are provinces with average annual per capita income growths that can be considered as moderately high (more than 5%), while majority of the provinces have income growth

that is comparable to the poorest countries in the world (around 1%). There are several reasons why the economic performance of the Philippines, in general, had been disappointing relative to its more successful East Asian neighbors. As noted by Balisacan and Hill (2003), "the Philippines economic performance looks deficient partly because it is most often compared with its neighbors ... the world's fastest growing".¹ Several papers have come out in recent years, specifically explaining what went wrong in the case of the Philippines.²

The relatively moderate and inconsistent (compared to the country's East Asian neighbors) income growth over a longer period is one of the major reasons of the high povertyincidence in the country. The official statistics on headcount poverty compiled by the National Statistical Coordination Board (NSCB) showed that while the percentage of poor Filipinos increased to 26.5% in 2009 (the latest poverty data) from 24.9 in 2003. Moreover, the percentage of subsistence poor (or food poor)³ did not change much during the same period, only slightly decreasing from 11.1% in 2003 to 10.8% in 2009. However, the number of food poor in the population has increased to about 9.44 million in 2009 from 8.8 million in 2003.

Empirical analysis using cross-country data during the period 1975 to 2000 (Mapa and Balisacan, 2004) points to the country's rapid population growth as one of the reasons why the country is not one of the high-performing Asian economies. The Philippines has the second largest population in Southeast Asia (about 92 million in 2010), next only to Indonesia, and ranks among the countries with the highest population growth rates in Asia.⁴The authors' econometric models showed that demographic factors have strong and significant effects on economic growth. Using simulation analysis to compare the income growth paths of the Philippines and Thailand, the authors showed that differences in the population growth rates between the two countries account for about 0.768 % point of forgone growth for the Philippines. It implies that had the Philippines followed Thailand population growth path during the period 1975 to 2000 the country's growth in the average income per person would have been 0.768 percentage point higher every year. Using the country's provincial data,

Using intra-country (provincial) data, Mapa, Balisacan and Briones (2006) identified the factors constraining the income growth in the Philippines provinces during the period 1985 to 2003. Using robustness procedures, the authors found that the high level of inequality and high percentage of young dependents (percentage of those aged 0 to 14 relative to the total population) are contributing negatively to the average provincial per capita income growth. The authors also showed the two variables to be robust determinants of provincial income growth. Moreover, the models also showed that the long running conflict in the provinces of the Autonomous Region of Muslim Mindanao (ARMM)⁵

¹ During the period 1976-2000, for example, the average growth rates (in US\$ Purchasing Power Parity or PPP) for the Philippines, Thailand and Korea are 4.1%, 7.98%, and 9.90%, respectively.

² Balisacan and Hill (2003) and Balisacan and Hill (2007) provide a collection of such good papers.

³ The prevalence of subsistence poor refers to the proportion of families or individuals with per capita income less than the per capita food threshold. The food threshold is determined using regional one-day menus priced at the provincial level. These menus are determined using low-cost nutritionally adequate food items satisfying basic food requirements of 2,000 calories which are 100% adequate for the recommended energy and nutrient intake (RENI) for energy and protein and 80% adequate for the RENI for vitamins, minerals and other nutrients (NSCB 2010).

⁴ The country's annual population growth rate from 1975 to 2000 is 2.36%, although this has gone down to 1.90% for 2000-2010 based on the results of the 2010 Census of Population. In May 1, 2010 the population of the Philippines is at 92.34 million. Moreover, data shows that the total fertility rate (TFR) is highest among the poorest households, where the TFR is 5.20 for the poorest 20% of households against the national average of 3.30 (as of 2008).

⁵ The Autonomous Region in Muslim Mindanao is the region of the Philippines that is composed of all the Philippines' predominantly Muslim provinces, namely: Basilan (except Isabela City), Lanao del Sur, Maguindanao, Sulu and Tawi-Tawi, and the Philippines' only predominantly Muslim city, the Islamic City of Marawi. The regional capital is at Cotabato City, although this city is outside of its jurisdiction. The poverty incidence (percent of population) in the region is 45.9% in 2009 and three of its provinces are among the poorest provinces in the Philippines in 2009, namely: Sulu (poverty incidence: 44.1%), Maguindanao (53.7%) and Lanaodel Sur (44.8%).

results to a lower average per capita income growth in these provinces relative to the other provinces in the Philippines.

Other studies notably, Balisacan (2005, 2007) and Balisacan and Fuwa (2002), showed the level of human stock capital (using child mortality rate as the proxy variable in the studies) as a statistically significant determinant of provincial income growth rate. In addition, these studies also showed the literacy rate and access to infrastructure as positive and significant determinants of provincial income growth rate. The authors also find that increment in land reform implementation (CARP) have a positive and significant effect on the average provincial income growth rate. To capture the effects of politics on provincial income growth, the Balisacan papers utilized the initial political conditions defined as the extent of dynasty within a province – empirically measured as the proportion of provincial officials related by blood or affinity. Using different econometric model specifications, Balisacan and Fuwa (2002) showed that the extent of dynasty has a negative and significant effect on income growth, while Balisacan (2007) showed the same dynasty variable, while negatively affecting income growth, is statistically insignificant.

The motivation of this paper is to investigate the reasons for having unequal growth rates in per capita income in the provinces in the Philippines. This research question will be answered by looking at the effect of spatial dependence (or location dependence) in the empirical growth models. The paper aims to contribute empirical evidence explaining why growth rates in the provinces are uneven and provide policy handles to mitigate this problem.

The rest of the paper is presented as follows. Section II gives an overview of the spatial dependence, particularly in the context of the econometric growth models. This section also explains the process of computing the Spatial Weight Matrix and the concepts of the Moran's Index, the Moran's Scatterplot. Section III provides the empirical analysis of the spatial dependence of the provincial income growth of the provinces while section IV discusses the results of the econometric models. Lastly, section V concludes.

2. Spatial autocorrelation in the econometric growth model

Spatial autocorrelation is defined as the coincidence of value similarity with location similarity (Anselin and Bera; 1998). Spatial dependence occurs when the observations of one location depends on the values of the other locations. The presence of spatial structure in the quantitative data provides the information about the similarity of the characteristics (e.g. income growth of neighboring provinces) vis-à-vis the distance between the locations (provinces) and the spatial autocorrelation of the variable (e.g. income growth) explains how this variation is affected by the distance (Fortin and Dale; 2009). Positive spatial autocorrelation happens when similar values of the variable of interest (e.g. income growth) cluster together, while negative spatial autocorrelation appears when dissimilar values are clustered in space. In the economic growth literature, the possibility that space is a determinant of income growth has been studied widely in the context of geographical variables such as climate and location (Gallup, Sachs, Mellinger; 1999). The area of spatial-econometrics, a sub-field of econometrics, looks at the possibility of spatial interaction and spatial structure and has recently been incorporated into the study of empirical growth (Durlauf and Quah; 1999). The applications of spatial econometric has been traditionally carried out in the regional science applications (Abreu, De Groot and Florax; 2004), where politically unstable regions/countries may have negative externalities or spillover effects on the other regions/countries. Since the publication of the book on Spatial Econometrics: Methods and Models by Luc Anselin (Anselin; 1988), numerous studies on spatial econometric analysis of geographical spillovers and growth have been made. The basic premise of spatial econometrics in regional/provincial economic growth studies is that regional/provincial data can be spatially ordered since similar regions tend to cluster and that econometric models must take into account the fact that economic phenomenon may not be randomly distributed on an economically integrated regional space (Baumont, Ertur, Le Gallo; 2001).

To capture potential spatial/spillover effects which indicates how the average growth rate of per capita income in the Philippine provinces is affected by its neighboring provinces, after controlling for other factors affecting income growth, Mapa, Balisacan and Briones (2006) introduced a measure of neighborhood effect in their intra-country growth regression models. This variable is computed as the average growth rate of the neighboring provinces, where the neighbors (of a specific home province) are identified using a contiguity-nearest distance based measure.⁶ The inclusion of this spatial variable, the neighborhood effect, into the growth regression model, conforms to the spatial auto-regressive model discussed by Anselin (1988, 2009). By introducing a spatial variable, the dynamics of how the provinces' economic performance interacts with each other can be better understood. The empirical analysis of the authors showed a negative and significant influence of the neighborhood effect in the growth rate of per capita income of the home province increases, the average growth rate of per capita income of the neighboring provinces decrease, showing some sort of negative growth externalities.⁷

This paper incorporates the provincial spatial dependence in the intra-country growth regression model (Barro-type) by utilizing the commonly used spatial lag model to capture this location dependency.⁸ Substantive spatial dependence is incorporated into the unconditional growth regression specification through a spatially lagged dependent variable,

$$g_{y^0} = \mathbf{X}\boldsymbol{\beta} - \alpha \log(y^0_{T_1}) + \theta\left(\frac{YD}{P}\right) + \rho \mathbf{W}g_{y^0} + \epsilon$$
(1)

where g_{y^0} is the growth rate of per capita income, y_{T1}^0 is the initial per capita income, YD/P is the proportion of young dependents (aged 0 to 14) in the population and X is the vector of economic and political variables (or the Barro's core) that may have impact on the steady state growth.

The matrix W is the spatial weights matrix with elements w_{ij} corresponding to the province (i, j). The spatial weight matrix is an $(n \times n)$ positive matrix (W) that provides "neighborhood sets" for each observation (Anselin, 2002). Its elements are non-stochastic, non-negative, finite, and exogenous to the model. An observation is not a neighbor to itself; hence the diagonal elements, denoted by w_{ii} , are set to zero. The element w_{ij} ($i \neq j$), indicates the spatial connection between region *i* to region *j*(Baumont, Ertur, & Le Gallo, 2001). To facilitate comparison among different models, the weight matrix shall be standardized so that the sum of the elements in a row equals unity. The choice on the kind of matrix to be used depends on the researcher. An example is the simple binary contiguity matrix, which is a matrix with values of elements equal to 1, when regions *i* and *j* share a border; and 0, otherwise. The use of the contiguity measure is not appropriate if there is a relatively high degree of heterogeneity in the spatial distribution of points or in the areas of regions (Anselin, 2002). For locations with high degree of heterogeneity, a spatial weight matrix with a distance-based critical cut-off is preferred. This paper employs the distance-based spatial weight, using the inverse of the distance, defined as:

$$w_{ij} = \begin{cases} 0 & if \ i = j \\ 1/d_{ij} & if \ d_{ij} < D \\ 0 & if \ d_{ij} \ge D \end{cases}$$
(2)

⁶ Using the contiguity measure, two provinces are neighbors if they share a common border. To mitigate the problem associated with some island provinces (e.g. Province of Cebu) in the country, the authors used the nearest neighbor method for this group of provinces.

⁷ Similar studies using European regions (Baumont, Ertur and Le Gallo (2001)) and US States/Counties show that the neighborhood effect is positive.

⁸ The two other methods of capturing spatial dependence are the spatial error model and the spatial cross-regressive model.

where d_{ij} is the distance between provinces i and j, using the provincial capital city/municipality as the point of reference and D is the cut-off distance, usually equal but not restricted to the first quartile of the distances (Baumont, Ertur, & Le Gallo, 2001). For the purposes of this study, the 35th percentile, a distance of about 655 kilometers, was used to ensure the existence of at least one neighbor for each province. Our interest is the parameter ρ in the econometric model in (1), which provides information on the spatial relationship of the per capita income growth of the home provinces and their neighbors, controlling for other factors.

2.1 Moran's Index

Existence of spatial autocorrelation indicates that the values of variables in one location are affected by values of the variables in neighboring locations. This is one of the problems faced by researchers in executing regression analysis on spatial relationships. Several measures of spatial autocorrelation, like the Geary's C (Geary, 1954), have already been proposed. The most popular of which is the Moran's Index (Moran, 1950), which will be the focal measure of spatial dependence that will be used in this paper. The Moran's Ihave shown to be consistently more powerful than Geary's C (Cliff and Ord, 1975; 1981), tests for global spatial autocorrelation for continuous data. The statistic is expressed as,

$$I = \left[\frac{n}{S_o}\right] \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} w_{ij} x_i x_j}{\sum_{j=1}^{n} x_i^2}$$
(3)

where n is the number of observations, w_{ij} is the element of the spatial weight matrix W corresponding to the ith row and the jth column, x_i and x_j are deviations from the mean of a variable for regions i and j, respectively, and S_o is a normalizing factor equal to the sum of the elements of the weights matrix W, that is, $S_o = \sum_{i=1}^n \sum_{j=1}^n w_{ij}$. If the rows of the weight matrix are standardized, the Moran's Index reduces to,

$$I = \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} w_{ij} x_i x_j}{\sum_{j=1}^{n} x_i^2}$$
(4)

In matrix notation, this is,

$$I = \frac{x'Wx}{x'x}$$
(5)

where W is a spatial weights matrix and x is a vector of observed values x_i , in deviations from the mean (Lim, 2003). Based on the cross-products of the deviations from the mean, the Moran's Index is similar but not equivalent to a correlation coefficientbetween x at one location and the weighted average of the values of x of its neighbors. Its range of possible values is from -1 to 1, where a positive value indicates that across all geographic units, similar values are more likely than dissimilar values between neighbors, and vice versa. Under the assumption of no autocorrelation, the expected value of the Moran's Index is $-\frac{1}{n-1}$, regardless of the specified weight matrix. The expected value approaches to zero as n approaches infinity. In addition, the Moran's Index is equivalent to the slope coefficient in the linear regression of the spatial W_x on x.

2.2 Moran's Scatterplot

The Moran's I statistic gives us a single global result for the whole data set. However, it does not provide information on the characteristics of spatial clustering. To provide graphical analysis of local

spatial dependence, one may use the Moran's scatterplot. The scatterplot is helpful in identifying outlying provinces. The Moran's scatterplot has four main parts and providing four different kinds of spatial association between the home province and its neighboring provinces. For this paper, the four parts correspond the following classifications: (a) high per capita income growth rate for the home province with high per capita income growth rate for neighboring provinces (HH - Quadrant I), (b) low per capita income growth rate for the home province with high per capita income growth rate for neighboring provinces (LH - Quadrant II), (c) low per capita income growth rate for the home province with low per capita income growth rate for neighboring provinces (LL - Quadrant III), (d) high per capita income growth rate for the home province with low per capita income growth rate for neighboring provinces (HL - Quadrant IV). One can think of it as "the spatial lag of the variable on the vertical axis and the original variable on the horizontal axis (Anselin, 2002)." A more comprehensive presentation of the spatial dependence of the average per capita income growth rate can be done using scatter plot maps.

3. Empirical analysis of the provincial data (1988-2009)

The dataset used in this study consist of information collected from the 74 provinces in the Philippines. The provinces are those defined in 1985 for data consistency. Currently, there are 80 existing provinces. The provinces are listed in Table 1 below. Data on average per capita income were sourced from the Family Income and Expenditure Surveys (FIES), data on road density were sourced from the Department of Public Works and Highways (DPWH). Information on the proportion of governors and mayors affiliated to the President comes from the Commission on Elections (COMELEC). The Philippine Statistical Yearbook (PSY) provides the data on the land type of all the provinces in the Philippines. The data on the percentage of municipalities classified according to the degree of slope is taken from GIS maps. The data on the annual amount of rainfall (in millimeter) for each province was taken from the Philippine Atmospheric, Geophysical and Astronomical Services Administration (PAGASA). The variables and the data sources used in the econometric models are provided in Table 2.

Region I	Region IV ⁹	Region VI	Region X	Region XIII
llocos Norte	Aurora	Aklan	Bukidnon	Metro Manila
Ilocos Sur	Batangas	Antique	Camiguin	XIV
La Union	Cavite	Capiz	Misamis Occidental	Abra
Pangasinan	Laguna	lloilo ¹⁰	Misamis Oriental	Benguet
Region II	Marinduque	Negros Occidental	Region XI	Ifugao
Batanes	Mindoro Oriental	Region VII	Davao ¹¹	KalingaApayao ¹²
Cagayan	Quezon	Bohol	Davao del Sur	Mt. Province
Isabela	Rizal	Cebu	Davao Oriental	Region XV
Nueva Vizcaya	Romblon	Negros Oriental	South Cotabato ¹³	Lanao del Sur
Quirino	Mindoro Occidental	Siquijor	Region XII	Maguindanao14
Region III	Palawan	Region VIII	Cotabato	Sulu
Bataan	Region V	Eastern Samar	Lanao del Norte	Tawi-Tawi
Bulacan	Albay	Leyte ¹⁵	Sultan Kudarat	Region XVI

Table	1. List of	74 Philippine	provinces	included	in the study

⁹ Region IV is now divided into CALABARZON or Region IV-A and MIMAROPA or Region IV-B.

¹⁰ Guimaras is now separated from Iloilo.

¹¹ Compostela Valley is now separated from Davao. The name of Davao changed back to Davao del Norte.

¹² KalingaApayao is now two different provinces: Kalinga and Apayao.

¹³ Sarangani is now separated from South Cotabato.

¹⁴ ShariffKabunsuan is now separated from Maguindanao.

¹⁵ Biliran is now separated from Leyte.

Nueva Ecija	Camarines Norte	Northern Samar	Agusan del Norte
Pampanga	Camarines Sur	Samar	Agusan del Sur
Tarlac	Catanduanes	Southern Leyte	Surigao del Norte ¹⁶
Zambales	Masbate	Region IX	Surigao del Sur
	Sorsogon	Baslian ¹⁷	
		Zamboanga del Sur ¹⁸	
		Zamboanga del Norte	

Table 2. Variables and data sources in the econometric mode	nables and data sources in the econometric models
--	---

Variables	Description	Data Source
GROWTH	Average Annual Growth Rate of Nominal Per Capita Income	Authors Computation from FIES data of NSO
LOGINIINC	Initial per capita income (in natural logarithm); for 1988, 1994 and 2000	FIES, NSO
POP014_INIT	percentage of young dependents	Authors Computation from FIES, NSO
AVERAIN	Average Rainfall	PAGASA
AVEGOVMAY	Percentage of Mayor and Gov with the President's Party	COMELEC
AVEEGINI	Gini Coefficient (income)	Authors Computation from FIES, NSO
AVEEGINI_SQ	Square of Gini (income)	Authors Computation from FIES, NSO
GRELECTSH	Growth Rate - HHs with Electricity	Authors Computation from FIES, NSO
GRROADNAT	Growth Rate - National Road	Department of Public Works and Highways
GEOGRAPHY	Dummy Variable for Geography (Landlock)	Philippine Statistical Yearbook
SLOPE	Slope of the Land	Geographical Information System
MINERAL	Province with Mineral	DENR, Mines and Geosciences Bureau

Using the data set, the authors created three (3) pseudo-panel data, for years: (a) 1988-1994, (b) 1994-2000 and (c) 2000-2009. One of the objectives of the paper is to look the changes (if there any) of spatial correlation or spatial dependence of the growth rates of income in the Philippine provinces through the years. Another reason for the creation of the pseudo-panel data is to increase the sample size to minimize the error in the estimation of the econometric models.

The average per capita income growth for the provinces during the three (3) pseudo-panel periods are provided in table 3. The figures show that the average income growth per person has been erratic and inconsistent, growing at just about 0.51 percent during the period 1988 to 1994, increasing at a relatively impressive rate of 3.45% in 1994 to 2000, but dropped again to 0.36 for 2000 to 2009. The overall per capita income growth during the period 1988 to 2009, a period of 21 years, is just about 1.24%.

Variable	Mean	Std. Dev.	Min	Max
Annual Growth Per Capita Income (1988-1994)	0.47	3.41	-8.96	10.6
Annual Growth Per Capita Income (1994-2000)	3.45	3.28	-3.19	13.0
Annual Growth Per Capita Income (2000-2009)	0.36	1 73	_/ 20	5.26

Table 3. Average per capita income growth of the Philippine provinces

The existence of spatial autocorrelation in income growth is examined using the Moran's index and the results are given in table 4. The estimated Moran's indices are all positive for the three panels and for the overall period 1988 to 2009, implying positive spatial correlation on the provincial per capita

1.24

1.35

-2.49

Annual Growth Per Capita Income (1988-2009)

¹⁶ Dinagat Islands was separated from SurigaodelNorte in 2006.

¹⁷ Basilan now belongs to Region XV (ARMM), and Aurora in Region III.

¹⁸ ZamboangaSibugay is now separated from Zamboangadel Sur.

income growth. This means that the average income growth of the home and neighboring provinces are more likely to be similar (increasing together or decreasing together), rather than being dissimilar (one is increasing while the other is decreasing). All these values, although small, mean that a positive autocorrelation is present in the data indicating the clustering of provinces with similar average per capita income growth rates.

Testing the significance of the Moran's indices, however, shows that only the values for 1988 to 1994 and the overall period 1988 to 2009are significantly different from zero. The indices for the periods 1994 to 2000 and 2000 to 2009 are not significantly different from zero.

Year	Moran's Index	SD	z-stat	p-value*
1988-1994	0.194	0.052	4.037	0.0000
1994-2000	0.036	0.052	0.963	0.1680
2000-2009	0.030	0.050	0.891	0.1860
1988-2009	0.120	0.049	2.725	0.0030

	Manan'a	المعالم م	f	A				C
i able 4.	ivioran s	indices	tor the	Average	per C	apita	Income	Growin

Note: * one-sided p-value

The figures below present the Moran's scatter plots and the corresponding Philippine maps that will be used to describe the movement of the average income growth rate of the provinces through time. The Moran's scatterplots in figures 1 to 4 display the distribution of all the 74 provinces on the four quadrants, based on the home provinces'relation to the average income growth rate of the neighboring provinces. The provinces in quadrants 1 (High-High or HH) and 3 (low-Low or LL) represent the home provinces that experienced similar income growth rates to the neighboring provinces, whereas those included in quadrants 2 (Low-High or LH) and 4 (High-Low or HL) represent the home provinces with contrasting income growth rates to the neighboring provinces.

The provinces included in quadrant 1 (HH) are the provinces where the both the home province and the neighboring provinces (using the distance measure) had average per capita growth rates higher than the national average, while the provinces in quadrant 3 (LL) are the are the provinces where the both the home province and the neighboring provinces had average per capita growth rates lower than the national average. The provinces in quadrant 2 (LH) are the provinces where the home province had lower average income growth compared to the national average, while the neighboring provinces experienced higher income growth compared to the national average. The provinces in quadrant 4 (LH) are the provinces where the home province had higher average income growth rate than the national average whereas the neighboring provinces had lower average income growth compared to the national average income growth



Figure 1. Moran's Scatter Plot (1988-1994)

The scatter plot corresponding to the period 1988 to 1994 indicates that about 60% of all the home provinces have similar income growth rates (higher or lower than the national average) with the neighboring provinces. About 34% of the provinces are in the High-High quadrant (quadrant 1), while 26% of the provinces are in the Low-Low quadrant (quadrant 1).

The rest of the provinces, about 40%, showed contrasting (or dissimilar) income growth rates with the neighboring provinces, with 29% in the Low-High quadrant (quadrant 2) and the remaining 11% in the High-Low quadrant (quadrant 4).



Figure 2. Moran's Scatter Plot (1994-2000)

For the period 1994 to 2000, the percentage of provinces having similar growth rates in per capita income with the neighboring provinces (higher or lower than the national average) dropped to about 54% (from 60% for the period 1988 to 1994). Moreover, only 22% of the provinces are in the High-High quadrant (quadrant 1), while a higher 32% are in the Low-Low quadrant (quadrant 1). The percentage of the provinces showing dissimilar income growth rates with the neighboring provinces increase to 46% (from 40% for the period 1988 to 1994), with 27% in the Low-High quadrant (quadrant 2) and 19% in the High-Low quadrant (quadrant 4). The results suggest that the degree of similarity in the average per capita income growth of the home provinces with their neighbors weakened during the

second period, as shown by the Moran's Index (in Table 4), where the value decreased to 0.036 (for 1994 to 2000) from 0.196 (for 1988 to 1994).



Figure 3. Moran's Scatter Plot (2000-2009)

For the period 2000 to 2009, the percentage of provinces having similar growth rates in per capita income with the neighboring provinces is about 55% (with 32% in the High-High quadrant (quadrant 1) and 23% in the Low-Low quadrant (quadrant 1)). The percentage of the provinces showing dissimilar income growth rates with the neighboring provinces increase is about 45% (with 26% in the Low-High quadrant (quadrant 2) and 19% in the High-Low quadrant (quadrant 4). Similar to the period 1994 to 2000, the degree of similarity of the income growth rate between the home provinces and the neighbors is very weak as shown by the Moran's Index for this period which is about 0.030.



Figure 4. Moran's Scatter Plot (1988-2009)

For the overall period 1988 to 2009, the percentage of provinces having similar growth rates in per capita income with the neighboring provinces is about 64% (with 42% in the High-High quadrant (quadrant 1) and 22% in the Low-Low quadrant (quadrant 1)). The percentage of the provinces showing dissimilar income growth rates with the neighboring provinces increase is about 36% (with 24% in the

Low-High quadrant (quadrant 2) and 12% in the High-Low quadrant (quadrant 4). The degree of similarity of the income growth rate between the home provinces and the neighbors is positive and significant, with a Moran's Index of about 0.12.



Note: Quadrant 1 is High-High; Quadrant 2 is Low-High; Quadrant 3 is Low-Low and Quadrant 4 is High-Low Figure 5. Mapping of the Philippine Provinces Income Growth (1988-1994) using the Moran's Quadrant



Note: Quadrant 1 is High-High; Quadrant 2 is Low-High; Quadrant 3 is Low-Low and Quadrant 4 is High-Low



Figure 6. Mapping of the Philippine Provinces Income Growth (1994-2000) using the Moran's Quadrant

Figure 7. Mapping of the Philippine Provinces Income Growth (2000-2009) using the Moran's Quadrant *Note*: (Quadrant 1 is High-High; Quadrant 2 is Low-High; Quadrant 3 is Low-Low and Quadrant 4 is High-Low)



Note: (Quadrant 1 is High-High; Quadrant 2 is Low-High; Quadrant 3 is Low-Low and Quadrant 4 is High-Low) Figure 8. Mapping of the Philippine Provinces Income Growth (1988-2009) using the Moran's Quadrant

4. Econometric models

This section discusses the empirical results of the econometric model using country's intracountry data for the period 1988 to 2009. The econometric model for the average per capita income growth (the dependent variable) is a Barro-type (Barro and Xala-i-Martin, 2004) growth regression model, augmenting it to incorporate the provincial spatial dependence using the distance measure of identifying the neighboring provinces of a particular home province (using the maximum distance of 655 kilometers to identify the neighbor). The weight used is the inverse of the distance implying that the correlation between income growth of the two provinces decreases with distance. The spatial dependence is referred to as the neighborhood effect. The explanatory variables include the initial per capita income of the province (to capture the conditional convergence or catching-up effect), the proportion of young dependents (aged 0 to 14) in the population (demographic factor), measure of inequality (using the expenditure Gini and its square), geographical variables such as average amount of rainfall, average slope of the land and an indicator variables if the province is landlocked or not. An indicator variable if the province has mineral resources is also included in the model. Measures of infrastructure are also included in the model using the proxy variables growth rates of households with electricity and the national road. A political indicator variable is incorporated in the model using the percentage of mayors and governors in the provinces who are affiliated with the same political party as the incumbent president of the country. The vector of economic and political variables (known as the Barro's Core) is included in the econometric since these usually affect the steady state growth of income.

The empirical results are shown in Tables 5 and 6 below. The results in Table 5 show the full model, while Table 6 shows the results of the reduced model after eliminating the statistically insignificant coefficients. In Tables 5 and 6, the coefficient of the neighborhood effect (or the spatial effect) is positive and significantly affecting the average income growth of the home province, controlling for other factors. In particular, if the average per capita income growth of the neighboring provinces increase by 1 percentage point, the average per capita income growth of the home province will increase by about 0.5 percentage points, all things being the same. This value shows a strong and positive spatial dependence between the home province and its neighbors. The positive sign of the spatial effect is also consistent with the values of the Moran's indices. The other variables that are significantly affecting the average growth of provincial per capita income are the initial income, percentage of young dependents and the political variable.

The magnitude of the coefficient of the natural logarithm of initial income (at -8.054 for the final model) implies that (conditional) convergence of provincial income occurs at the rate of about 8 percent per year.¹⁹ This result is congruent with the expectation of conditional convergence, that is, the economy grows faster the further it is from its own steady state level of income. Thus, on the average, provinces with higher per capita income at the start of the sample period experienced lower average growth rate relative to provinces with lower initial income per capita, all other things being the same. In other words, poorer provinces have the opportunity to catch up (in terms of income growth) with the richer provinces in the long run. Note, however, that this convergence is conditional in that it predicts a higher growth in response to a lower starting provincial income per person if the other explanatory variables are held constant.

The demographic variable, proportion of young dependents, has a negative and significant effect on the average per capita income growth of the provinces. The estimated coefficient of -7.342 (in the final model) implies that a one-percentage point reduction in the percentage of young dependents

¹⁹ This estimate of the rate of conditional convergence of the model is higher than that previously estimated by Balisacan (2005) at 4% per year and closer to the estimate of Balisacan and Fuwa (2002) which was 9% per year for the Philippines provincial data.

at the start of the period will results in an estimated 7.3 basis points increase on the average growth rate of income per person, all things being the same. The absolute figure of 7.3 basis points may seem small but it should be considered that the estimated increase in income growth, as provided by the model, is accumulated over the period 1988 to 2009 (over 21 years), which is substantially large at the end of the period 2009. This result supports the earlier studies, notably Mapa and Balisacan (2004) and Bloom and Williamson (1997), using cross-country data, that a country with a large proportion of young dependents will experience constricting effects on its economic growth during the first phase of the demographic transition.

The political variable, percentage of mayors and governors that are affiliated with the same party as the country's president, is also a significant determinant of per capita income growth. However, the sign of the estimated coefficient for this variable is negative, somewhat inconsistent with the expectation that being affiliated with the same party as the country's president may result higher, rather than lower income growth. Being affiliated with the seating president's political party may bring in more infrastructure projects that will likely increase income growth. A study on the impact of political variable on infrastructure development by Balisacan, Mapa, Fuwa, Abad-Santos and Piza (2011) showed that growth rate in the percentage of household with electricity (the authors' proxy for infrastructure) is targeted to provinces with higher proportion of mayors and governor affiliated with the President's party.

Variable	Estimated Coefficient	SE	p-value
Neighborhood Effect (Spatial Effect)	0.522	0.226	0.021
Initial Income (in natural logarithm)	-9.515	1.701	0.000
Percentage of young dependents	-5.808	4.495	0.196
Average Rainfall	0.000	0.000	0.239
Percentage of Mayor and Gov with the President's Party	-3.239	1.454	0.026
Gini Coefficient	22.688	36.155	0.530
Square of Gini	-31.165	51.383	0.544
Growth Rate - HHs with Electricity	-0.018	0.042	0.668
Growth Rate - National Road	-0.028	0.025	0.268
Dummy Variable for Geography (Landlock)	-0.428	0.323	0.186
Slope of the Land	0.297	0.221	0.180
Province with Mineral Resources	0.186	0.485	0.701
Year Indicator (for 1994-2000)	2.724	0.489	0.000
Year Indicator (for 2000-2009)	0.492	0.525	0.348
Constant Term	41.648	9.861	0.000

Table 5. Results of the Spatial Lag Econometric Model for Averag	e
Provincial Growth Rate (Full Model)	

Note: Wald's Test for Spatial Coefficient (Rho) = 0 (Test Stat = 5.335; p-value = 0.021)

Table 6 Results of the Spat	ial Lag Econometric Model f	or Average Provincial Gro	wth Rate (Final Model)
Tuble 6. Robulto of the oput	iai Lag Loononiothotho moaor i	or / worugo r rovinioiar ore	

Variable	Estimated Coefficient	SE	p-value
Neighborhood Effect (Spatial Effect)	0.548	0.216	0.011
Initial Income (in natural logarithm)	-8.054	1.497	0.000
Percentage of young dependents	-7.342	4.218	0.082
Percentage of Mayor and Gov with the President's Party	-2.895	1.441	0.045
Year Indicator (for 1994-2000)	2.465	0.406	0.000
Constant Term	39.416	7.114	0.000

Note: Wald's Test for Spatial Coefficient (Rho) = 0 (Test Stat = 6.459; p-value = 0.011)

Conclusion

This study looks at the spatial relationship of the average per capita income growth using intracountry or provincial data from 1988 to 2009. The results from the study provide insights on the geographical dimensions of provincial income growth and showed evidence on the role of spatial effects in the formal econometric analysis of intra-country income growth models. Despite the data limitations, the study provides a strong empirical evidence of the presence of positive spatial dependence or degree of similarity in the average per capita income growth of the provinces, albeit the degree of positive spatial dependence weakens in the latter periods. This positive spatial correlation suggests the provinces may be converging in terms of their income growth and they do so in movements similar to their neighbors. The finding of the study that spatial dependence weakened in the latter periods (1994-2000 and 2000-2009) needs to be analyzed further. The weakening of spatial dependence may provide insights on the uneven provincial/regional income growth experienced in the country. One possible explanation of the weak spatial dependence is that two or more groups of neighboring provinces are growing at similar rates within the group, but at different rates across groups. This opens the possibility of having different convergence clubs (of provinces) within the country. We hope this paper can stimulate others to investigate further the spatial dimensions of income growth in the country.

References

- [1] Abreu, M., De Groot, H.L.F., Florax, R.J.G.M. (2004). Space and Growth: A Survey of Empirical Evidence and Methods. Working Paper, Department of Spatial Econometrics, Vrije Universiteit, The Netherlands.
- [2] Anselin, L. (1988). Spatial Econometrics: Methods and Models. Boston, Kluwer Academic.
- [3] Anselin, L., Bera, A. (1998). Spatial Dependence in Linear Regression Models with an Application to Spatial Econometrics, in A. Ullah and D.E.A. Giles (Eds.), *Handbook of Applied Economics Statistics*, Springer-Verlag, Berlin, 21-74.
- [4] Anselin, L. (2002). Under the Hood. Issues in the Specification and Interpretation of Spatial Regression Models, *Agricultural Economics*, 27: 247-267.
- [5] Anselin, L. (2009). Spatial Regression, in A. S. Fotherringham and P.A. Rogerson (eds.), *The Sage Handbook of Spatial Analysis*, Sage Publications.
- [6] Balisacan, A.M. (2007). Local Growth and Poverty Reduction, in A. Balisacan and H. Hill (eds), *The Dynamics of Regional Development: The Philippines in East Asia,* Ateneo De Manila University Press, 398-421.
- [7] Balisacan, A.M. (2005). Sub-National Growth and Poverty Reduction. Paper prepared for the Finalization Workshop on Policies and Institutions for Advancing Regional Development: The Philippines in Comparative East Asian Context. Asian Development Bank, Manila, February 2005.
- [8] Balisacan, A.M., Hill, H. (2007). The Dynamics of Regional Development: The Philippines in East Asia. Ateneo De Manila University Press.
- [9] Balisacan, A.M., Hill, H. (2003). An Introduction to the Key Issues, in A. Balisacan and H. Hill (eds), *The Philippine Economy: Development, Policies and Challenges, New York: Oxford* University Press.
- [10] Balisacan, A.M., Fuwa, N. (2002). Going Beyond Cross-Country Averages: Revisiting Growth, Inequality and Poverty in the Philippines. Discussion Paper Series on International Development Strategies No. 2001-005, Foundation for Advanced Studies on International Development, Tokyo.

- [11] Balisacan, A.M., Mapa, D., Fuwa, N., Abad-Santos, C., Piza, S. (2011). Addressing Infrastructure Constraints to the Development of the Rural Non-farm Sector: Implications for Poverty and Rural Development, World Bank Report.
- [12] Barro, R.J., Sala-i-Martin, X. (2004), Economic growth, 2nd Ed. Cambridge, Mass, MIT Press.
- [13] Baumont C., Ertur, C., Le Gallo, J. (2001). A spatial econometric analysis of geographic spillovers and growth for European regions, 1980-1995. Working Paper No. 2001-04, LATEC, University of Burgundy, Dijon - France.
- [14] Bloom, D.E., Williamson, J.G. (1997). Demographic Transitions, Human Resource Development, and Economic Miracles in Emerging Asia, in J. Sachs and D. Bloom (eds) *Emerging Asia*. Manila: Asian Development Bank.
- [15] Cliff, A.D., Ord, J.K. (1975). The choice of a test for spatial autocorrelation, in J.C. Davies and M.J. McCullagh (eds) *Display and Analysis of Spatial Data*, John Wiley and Sons, London, 54-77
- [16] Cliff, A. D., Ord, J.K. (1981). Spatial processes models and applications. London: Pion.
- [17] Durlauf, S.N., Quah D. (1999). The New Empirics of Economic Growth, in Taylor J. and Woodford M. (Eds.), *Handbook of Macroeconomics*, North-Holland Elsevier Science.
- [18] Fortin, M., Dale, M.R. (2009). Spatial Autocorrelation, in A. S. Fotherringham and P.A. Rogerson (eds.), *The Sage Handbook of Spatial Analysis*, Sage Publications.
- [19] Gallup, Sachs, and Millenger (1999). Geography and economic development, *International Regional Science Review*, 22 (2): 179-232.
- [20] Geary, R.C. (1954). The Contiguity Ratio and Statistical Mapping, *The Incorporated Statistician*, 5(3): 115-127.
- [21] Lim, U. (2003). A Spatial Analysis of Regional Income Convergence, *Planning Forum*, 9: 66-80.
- [22] Mapa, D.S., Balisacan, A. (2004). Quantifying the Impact of Population on Economic Growth and Poverty: The Philippines in an East Asian Context, E. Sevilla (ed), *Population and Development in the Philippines: The Ties That Bind*.AIM Policy Center, Makati City.
- [23] Mapa, D.S., Balisacan, A.M., Briones, K.S. (2006). Robust Determinants of Income Growth in the Philippines. *Philippine Journal of Development*, Volume XXXIII, (1-2): 1-32.
- [24] Moran, P.A.P. (1950). Notes on continuous stochastic phenomena, Biometrika, 37(1): 17-23.
- *** National Statistical Coordination Board, 2012. Website, www.nscb.gov.ph



DOI: http://dx.doi.org/10.14505/tpref.v6.1(11).02

Bubbles, Bluffs and Greed

Taiji HARASHIMA Department of Economics, Kanazawa Seiryo University, Japan <u>harashim@seiryo-u.ac.jp</u>

Suggested Citation:

Harashima, T. (2015). Bubbles, bluffs, and greed. *Theoretical and Practical Research in Economic Field*, (Volume VI, Summer), 1(11): 21-49. DOI:10.14505/tpref.v6.1(11).02. Available from: http://www.asers.eu/journals/tpref/curent-issue.

Article's History:

Received April, 2015; Revised June, 2015; Accepted August, 2015. 2015. ASERS Publishing. All rights reserved.

Abstract:

A rational bubble cannot theoretically exist if people have infinite horizons. This paper shows that a bubble-like phenomenon can be generated by a "bluff" even if people are rational and have infinite horizons. A bluff is defined as the behavior of an agent who pretends to possess private information to gain profits, particularly (false or misleading) information that the representative household's rate of time preference (RTP RH) has changed. An alternative definition of the representative household indicates that households must ex ante generate an expected RTP RH to behave optimally, but the expected RTP RH has to be generated based on beliefs about the RTP RH. Bluffers exploit the opportunities derived from the fragile nature of the expected RTP RH. The driving force behind bluffs is greed because bluffers do not work hard to gain profits by producing and selling better goods and services more cheaply, but by disseminating contaminated information, or acting in such a way to mislead people into believing the expected RTP RH has changed.

Keywords: bubble, bluff, greed, time preference, representative household, financial supervision.

JEL Classification: E32, E44, G14

1. Introduction

The Great Recession that occurred in the latter half of the 2000s forced us to once again realize the importance of the economic phenomenon known as a "bubble". In this case, it was not only a single commodity's bubble that burst but a global bubble-led economic boom. Theoretically, a rational bubble cannot exist if agents have infinite horizons (Blanchard and Watson, 1982; Santos and Woodford, 1997). Hence, an economic boom led by a rational bubble is also impossible if agents have infinite horizons. An assumption of some kind of irrationality may therefore be necessary to explain the existence of bubble-like phenomena. However, merely making *ad hoc* assumptions of irrational agents does not appear to be a compelling argument. It is too easy to *ad hoc* assume irrationality because, if we assume irrationality in human behaviors, we can explain any otherwise unexplainable human phenomena. However, there is another possible source of this bubble-like phenomenon. If some factor

or factors obstruct agents' rational decision making, a bubble may be generated. In this paper, I show that a mechanism which I call a "bluff" is such a factor, and it generates a bubble-like phenomenon even if people are rational and have infinite horizons. In this context, I consider a bluff to be a strategy or trick in which an agent pretends to possess private information that is actually untrue to gain profit. To the best of my knowledge, this type of strategy has not been studied as a source of bubble-like phenomena, but it can potentially obstruct agents' decision making and force them to make non-optimal decisions *ex post*; in other words, people can potentially be fooled or cheated by an agent who bluffs.

I show that, by utilizing private information about the rate of time preference of the representative household (RTP RH), a bluffer can generate a bubble-like phenomenon. Becker (1980) and Harashima (2014a, b) indicate that it is not possible to assume the representative household as the average household in dynamic models. An alternative definition of the representative household is shown in Harashima (2014a, b); it entails the collective behavior of households under sustainable heterogeneity. This alternative definition of the representative household must generate an expected RTPRH*exante* for it to behave optimally. However, although a household knows its own rate of time preference (RTP), it cannot directly observe the RTP RH. Therefore, the expected RTP RH has to be generated based on each household's beliefs about the RTP RH. The expected RTP RH is therefore fragile. If bluffers can exploit opportunities provided by this fragility and manipulate the expected RTP RH, the use of bluffs can cause a bubble-like phenomenon.

Greed is the driving force behind bluffs. Financial crises and the ensuing economic crises (e.g., the Great Depression or the Great Recession) are often seen as the consequences of greed. However, some people have argued that greed is an indispensable driving force of capitalism and therefore should not be blamed. Bluffers' greed, however, should clearly be blamed for negative economic outcomes because bluffers do not work harder to obtain profits by producing and selling better goods and services more cheaply; rather, they gain profit by acting in such a way to mislead people into believing the expected RTP RH has changed.

2. The expected rate of time preference of representative household (RTP RH)

2.1. An alternative definition of the representative household

2.1.1. The definition

As Becker (1980) and Harashima (2014a, b) indicate, it is not possible to assume the representative household as the average household in dynamic models. Harashima (2014a, b) shows an alternative definition of the representative household such that the behavior of the representative household is defined as the collective behavior of all households under sustainable heterogeneity. The reason why this alternative definition is needed and the nature of sustainable heterogeneity are described in detail in the Appendix. Unlike the case in which the representative household is assumed to be the average household, this alternatively defined representative household reaches a steady state in which all households satisfy all of their optimality conditions in dynamic models, even if households are heterogeneous. In addition, the alternatively defined representative household has an RTP that is equal to the average RTP as shown in Eqs. (A7) and (A8) in the Appendix.²⁰ Hence, we can assume not only a representative household but also that its RTP is the average rate of all households.

2.1.2 Necessity of expecting RTP RH

This alternatively defined representative household requires that each household must generate an expected RTP RH *ex ante* for it to behave optimally as shown in the Appendix (see also Harashima, 2014a, b). However, a problem remains. Although a household knows its own RTP, it cannot directly observe the RTP RH, and therefore, the expected RTP RH has to be generated based

²⁰ If sustainable heterogeneity is achieved with the help of government intervention, the time preference rate of the representative household will not be exactly equal to the average rate of time preference.

on its beliefs about RTP RH. Beliefs will be formed by using heuristics (a detailed explanation of the expected RTP RH and the heuristics used to estimate it are shown in the Appendix).

2.2 The path when the expected and intrinsic RTP RHs are different

The alternative definition of the representative household requires sustainable heterogeneity. Achieving sustainable heterogeneity affects the behavior of individual households because sustainable heterogeneity requires that each household must consider other households' optimality (as well as the behavior of the government, if necessary). This feature does not, however, mean that households behave cooperatively. Each household behaves autonomously based on its own RTP, but at the same time, its behavior is influenced by whether or not the other households' optimality conditions are achieved. This consideration affects the actions a household takes in that it affects the choice of a household's initial consumption.

Sustainable heterogeneity indicates that a household's future path of consumption has to be consistent with the future path of sustainable heterogeneity. Therefore, each heterogeneous household sets its initial consumption such that it proceeds on the path that is consistent with the path of sustainable heterogeneity and eventually reaches a steady state. That is, each heterogeneous household sets its initial consumption counting backward from its expected consumption in the steady state, which is calculated based on the expected RTP RH, to its present consumption supposing that all the other households also set their initial consumptions in a similar manner. All households thereby behave autonomously in the same manner. Even if the expected RTP RH is not equal to the intrinsic RTP RH, all households set their initial consumption, each heterogeneous household proceeds according to its own optimality conditions and eventually reaches a steady state. In the steady state, households' capital accumulation stops, and production, consumption, and capital no longer change. Therefore, even if the expected RTP RH, the expected RTP RH, the steady state is stable and can be kept for a long period as long as households believe that the expected RTP RH is the correct one.

This nature is important because it indicates that the economy is subject not to the intrinsic RTP RH but to the expected RTP RH. Even if the expected RTP RH is actually very different from the intrinsic RTP RH, the economy will appear to proceed quite "normally" for an indefinite period of time without any inconsistencies among observed economic indicators. Therefore, the observed economic indicators alone cannot tell us whether the expected RTP RH is truly identical to the intrinsic RTP RH or whether or not the current economy is in a bubble-like state.

3. Bluff

The important role of the expected RTP RH as discussed in Section 2 indicates that, if the expected RTP RH can be manipulated, the economy can also be manipulated. The fragility of the expected RTP RH (i.e., it is formed based on beliefs) indicates that there is room for an agent (probably a malicious agent) to manipulate the expected RTP RH, for example, by intentionally disseminating misleading information.

3.1 Bubble

In bubble theory, a bubble is defined as an excess of asset prices over their fundamental values or intrinsically useless assets that are traded at positive prices. The theory concludes that if all agents are rational and has infinite horizons, bubbles cannot be generated. Therefore, no bubble-led economic boom can exist theoretically. Hereafter, I use the term "bubble" not only to indicate an excess of asset prices but more broadly to include a bubble-led economic boom. An excess of asset prices may not always accompany an economic boom, but I focus on the cases where an excess of asset prices does accompany the boom.

Despite their theoretical impossibility, many bubble-like phenomena have been observed across economies and time periods. According to bubble theory, if intrinsically useless assets are really being traded at positive prices, the condition of rational decision makes must be violated to some extent somewhere along the line; for example, many agents behaving as if they have finite horizons. However, the *ad hoc* assumption that agents are intrinsically irrational is not compelling. There is, however, another possible source of interference with rational decision making. If some factor or factors disturb the decision-making process, the decisions made will not be optimal *ex post*. If expectations of the future economic path are skewed or somehow flawed, for example, because information is incomplete and asymmetric among agents, intrinsically useless assets may be traded at positive prices and then a bubble-like phenomenon may be generated. The existence of bluffs may therefore allow the generation of bubble-like phenomenon if the bluffers can successfully manipulate information.

Even if information manipulation can cause small-scale episodes of trading in intrinsically useless assets, can it generate economy-wide large-scale bubble-like phenomena? Although excesses in some asset prices may be caused through information manipulation, it does not necessarily follow that these excess prices will lead to an economy-wide boom. To generate an economy-wide boom, the manipulated information not only needs to have a very large impact on the economy as a whole, but it must also not be easily detected by people. If the manipulation is easily detected, the effects of the manipulation will soon vanish.

3.2 Bluff

Although it may seem that there is no type of manipulation that would satisfy the conditions for generating the bubble-like phenomenon discussed in Section 3.1, manipulating information with regard to the expected RTP RH could satisfy the conditions and cause bubble-like phenomena. The expected RTP RH has a huge impact on the economy because it is the discount factor of future utilities in the expected future economy, and it is not easy for households to detect information manipulation merely by observing economic indicators as discussed in Section 2. Bluffs can be used to intentionally manipulate the expected RTP RH. In a poker game, a bluff is generally a bet made by a player with an inferior hand. By analogy, a bluff herein is defined as the behavior of an agent who pretends to possess the (false) information that the intrinsic RTP RH has changed. The bluffer hopes that his actions will lead people believe the false information and take actions that will ultimately benefit him. The misleading actions of RTP RH. Households may become confused by the information the bluffer has disseminated, and in some cases, they may believe that this false information is actually true.

One way of bluffing is to inject huge amounts of money into financial markets to make loans or buy assets that are usually regarded to be too risky to own. Episodes when such large amounts of money have been injected for loans or risky stocks are not rare. Examples include the subprime mortgage loans in the United States in the first decade of the 21st century, the dot-com bubble in the United States in the late 1990s, or the real estate loans by banks in Japan in the later 1980s. Although it remains uncertain whether these episodes were generated intentionally by bluffers, we can agree that during these periods financial institutions injected huge amounts of money into projects that have historically been viewed as quite risky.

As households observe such huge injections of money into highly risky projects, they may ask themselves why some agents are taking these actions. There are several possibilities - the agents may be irrational or foolish, they may be bluffing, or the expected RTP RH of many households has changed and therefore projects that previously were deemed to be too risky are no longer viewed in that way. House holds judge which of these possibilities is true by using heuristics. However, there is no guarantee that households will always reach the correct conclusion. There is the possibility that at least some households will come to believe that the expected RTP RH of other households has really

changed even though, in actuality, the information has been contaminated by the bluffers, whose aim is to exploit these misconceptions.

As stated previously, if the expected RTP RH is successfully manipulated in this manner, its impact on the economy is huge because the RTP RH is the discount factor in expectations of future utilities. Even a small downward shift of the discount factor will largely increase the expected steadystate production and generate a bubble-like economic boom. The more the bluff changes people's expected RTP RH, the larger the bubble and the expected payoffs to the bluffers who initiated this movement will be. No household can know the bluffer's initial state of mind, and as shown in Section 2, it is not easy for households to detect contaminated information by observing economic indicators. As a result, the economy may appear to be "normal" and steady as shown in the Appendix. Hence, the bubble-like economic boom can continue for a long period.

If bluffers can successfully change households' expected RTP RH, their "investments" (i.e., money used for the bluff) can yield a huge amount of profits by successfully influencing the sale of stocks or other financial instruments before households detect the bluff.

3.3 Return on assets or RTP RH?

Information can also be distorted by altering information about the expected future dividend. The present value of an asset (the non-bubble part of the price) is most simply expressed as:

 $E\int_0^\infty \exp\left(-\theta t\right) D_t dt$

where: θ is RTP, D_t is the return on asset in period t and E is the expectation operator. Hence, in addition to the expected RTP RH, contaminated information can also be used to skew future dividends to generate a bubble-like phenomenon.

However, on average, the returns D_t are proportionate to the marginal returns on capital $\frac{df(k_t)}{dk_t}$

and at the steady state,
$$\frac{df(k_t)}{dk_t} = \theta$$
.

That is, the expected future stream of average returns is determined depending on the expected RTP RH. Bluffers may disseminate false information about expected future dividends by, for example pretending to possess knowledge on future technologies, but at the steady state, the equation $\underline{df(k_i)} = \theta$ still holds.

 dk_t

Therefore, given the expected RTP RH, there will be little room for bluffers to exploit opportunities for payoffs by contaminating information about the average returns on assets in the economy. Hence, the bluffer's target manipulations will be limited to the expected RTP RH.

3.4 Bluff conditions

3.4.1 Conditions

Suppose for simplicity that all bluffers are identical. Therefore, bluffers' actions (bluffs) are represented by a representative bluffer (hereafter, "the bluffer"). Let $\overline{\pi}$ be the subjective payoffs of a bluff to the bluffer if the expected RTP RH is successfully manipulated. Let p ($0 \le p \le 1$) be the probability that, after observing the information the bluffer disseminated, households decide that the expected RTP RH has changed. If households do not believe the information and do not change the expected RTP RH, the bluff fails and the bluffer suffers the loss $-\underline{\pi}$ where $0 < \underline{\pi} < \overline{\pi}$. It is assumed for simplicity that $\overline{\pi}$, $-\underline{\pi}$, and p are identical for any bluff.

The expected payoffs to the bluffer (Π) for a bluff is therefore

 $(-p^2) > 0$

$$\Pi = p\overline{\pi} + (1-p)(-\underline{\pi}) = p(\overline{\pi} + \underline{\pi}) - \underline{\pi}$$
⁽¹⁾

Eq. (1) indicates that, even if p is small, $\Pi > 0$ if $\overline{\pi}$ is sufficiently large. The variance of Π , σ^2 , is

$$\sigma^2 = \left(p - p^2\right)\left(\overline{\pi} + \underline{\pi}\right)^2 \,.$$

Because $0 \le p \le 1$, then $p - p^2 \ge 0$, $\sigma^2 \ge 0$,

$$\frac{d\sigma^2}{d\overline{\pi}} = (p - p^2)2(\overline{\pi} + \underline{\pi}) > 0 \qquad \text{and} \qquad \frac{d^2\sigma^2}{d\overline{\pi}^2} = 2(p - p^2)2(\overline{\pi} + \underline{\pi}) > 0$$

In addition, because $0 \le p \le 1$,

$$\frac{d\sigma^2}{d\Pi} = 2(1-p)(\overline{\pi} + \underline{\pi}) > 0 \qquad \text{and} \qquad \frac{d^2\sigma^2}{d\Pi^2} = 2(p^{-1}-1) > 0$$

Hence, a payoff curve can be drawn as the bold line on the Π - σ^2 plane in Figure 1. The thin solid line in Figure 1 indicates the indifference curve of the bluffer that has a point of contact at A with the payoff curve. At point A, the bluffer undertakes a bluff.

For simplicity, additional assumptions are introduced. In every period, the bluffer can undertake one bluff because of financial constraints, and the chances of a bluff being made at any point on the curve in the Π - σ^2 plane occur randomly. That is, in a given period, a bluff succeeds with the probability p if a bluff that corresponds to point A on the curve in the Π - σ^2 plane occurs. If the chances of a bluff corresponding to point A do not occur frequently and p is small, then a bubble-like phenomenon caused by a bluff will not necessarily be frequently observed.

In the case of a more risk adverse bluffer, the shape of indifference curve will be something like the thin dotted line shown in Figure 1. A bluff will not be undertaken in this case because the bluffer's indifference curve has a point of contact at B with the payoff curve, but at point B, payoffs are negative and thus the bluff is not profitable. Nevertheless, bluffers are likely to be less risk averse than ordinary people. Hence, the shape of the bluffer's indifference curve will usually resemble the thin solid line in Figure 1 where the payoff (Π) at point A is positive.



3.4.2 Initial costs

The occurrence probability of a bubble-like phenomenon caused by a bluff will be different depending on the value of $\underline{\pi}$ because point A varies depending on $\underline{\pi}$. Suppose that p is identical for any π .

 $\underline{\pi} \cdot Because \quad \frac{d^2\sigma^2}{d\overline{\pi}^2} = 2(p-p^2), \quad \text{as:} \quad \underline{\pi} \text{ increases}, \quad \Pi \quad \text{decreases} \quad \text{and} \ \sigma^2 = (p-p^2)(\overline{\pi} + \underline{\pi})^2,$ $\frac{d\sigma^2}{d\overline{\pi}} = (p-p^2)2(\overline{\pi} + \underline{\pi}), \text{ and} \quad \frac{d^2\sigma^2}{d\overline{\pi}^2} = 2(p-p^2)\text{ increase}.$

There fore, when $\underline{\pi}$ increases, the bold solid line shifts to the bold dashed line in Figure 2, and point A moves to the point A'. Π for A' is smaller than Π for A. As $\underline{\pi}$ increases, Π decreases and eventually becomes negative. Hence, if $\underline{\pi}$ is sufficiently large, even if the bluffer's degree of risk averseness is very low, the bluffer will not undertake a bluff because the expected payoffs for a bluff are negative. Conversely, if $\underline{\pi}$ is sufficiently small, the bluffer will undertake a bluff even if the bluffer's degree of risk averseness is relatively high.



3.5 The end of a bluff

As shown in Section 2, once a bluff succeeds the economy proceeds "normally" and steadily on the path calculated based on the manipulated expected RTP RH until households detect that a bluff have occurred. If the bluffer can successfully continue to hide their intentions, households do not doubt that they are behaving optimally. Nevertheless if households detect the bluff, the bubble-like phenomenon will end. However, how do households detect a bluff? As Section 2 demonstrated, the economy looks to be proceeding "normally" on a path that was expected to be optimal. In addition, most

observed economic data are consistent with this expected optimal path and can be interpreted as indicating that the economy is quite normal.

There are two ways of detecting a bluff. First, authorized information suggesting the existence of the bluffmay be provided by the government. Here, the supervising financial authority plays an important role. Unlike households, the financial authority has special powers of investigation. If its supervision is sufficiently prudent and extensive, it is likely that the authority will eventually uncover the bluff. When the financial authority perceives the existence of a bluff, it has the power to force the bluffer to stop bluffing and provide this information to households. In this sense, prudential supervision is crucially important to help households avoid the damages inflicted on them by bluffers. The importance of the financial authority's supervision conversely indicates that, if households begin to doubt whether the authority's supervision is caused by a bluff because the supervision has failed. The bluff and bubble-like phenomenon will end when households begin to doubt either the supervising authority.

A second possible way of detecting a bluff is if the bluffer pushes too far because of greed. The bluffer may have to take on a high level of risk of being detected to obtain greater payoffs. The probability that the hidden intention of the bluffer will be detected by households will increase as the bluffer pursues greater amounts money. A bluffer who has become complacent or conceited because of the success of the bluff may indeed engage in such high-risk behavior. This possibility appears to be likely because bluffers are quite likely to be greedy by nature. The bluffer does not work hard to gain profits by producing and selling better goods and services more cheaply, but by disseminating contaminated information. This behavior clearly indicates that they are intrinsically greedy.

When a bluff is detected, households will not only return the expected RTP RH to the former value, but they may go further. That is, they may generate a much higher expected RTP RH than the original one (before the bluff) because the credibility of the financial authority has been undermined and therefore future uncertainty about the economy increases. The structural model of RTP indicates that an increase in future uncertainty increases RTP (see the Appendix and Harashima, 2014a, b), so the expected RTP RH will be changed to a higher value than that before the bluff. The subsequent economic stagnation will worsen as compared with the case where the expected RTP RH simply returns to the pre-bluff value.

3.6 The bluffer is not a noise trader or criminal

Although a bubble-like phenomenon generated by a bluff is not a rational bubble as defined in bubble theory, it can be observed even if all agents behave rationally and have infinite horizons. A bubble-like phenomenon is generated not because of irrationality, but rather because of fragility in the belief and expectation of the RTP RH as shown in Section 2. This fragility provides the opportunity for bluffers. Hence, the bluffer is not a noise trader because noise traders are irrational and bluffers behave completely rationally. In fact, both households and bluffers behave quite rationally.

Furthermore, a bluff is not a crime in most, if not all, countries. "Pump and dump" stock sales are a crime in most countries because false news is intentionally disseminated to gain profits. Although the bluffer contaminates information similar to a pump and dump strategy, false news is not actually disseminated. Instead, the bluffer engages in equivocal behavior and confuses households, but the bluffer's behavior itself is not false news. The interpretation of the bluffer's actions is completely up to households. Even if households wrongly interpret them, the bluffer is not responsible for their wrong interpretation. In theory, confusing households could be ruled out by law, but there are many kinds of economic activities that have the potential to confuse households to a greater or lesser extent. It would be very difficult to enact a law that strictly defines a bluff, while also distinguishing it from other confusing activities. Bluffs therefore should not be controlled by law enforcement agencies, but they should be strictly and prudently supervised by the financial authority.

Conclusion

Rational bubbles and economic booms led by them are not theoretically possible if agents have infinite horizons. However, assuming irrationality *ad hoc* does not appear to be a compelling argument. Even if people are rational and have infinite horizons, a bubble-like phenomenon can be generated if the expectation of the future economic path is hindered by some factors. In this paper, I show that bluffs are such a factor. The bluffer manipulates the household's expected RTP RH by acting in such a way as to lead people to believe incorrect information. A bluffer can thereby generate a bubble-like phenomenon even if people are rational and have infinite horizons because the expected RTP RH is the discount factor that is used to determine the expected future economic path.

An example of bluff is the injection of a huge amount of money into financial markets to give loans or buy assets that would usually be regarded too risky. Bluffs are effective because each household must generate an expected RTP RH *ex ante* for it to behave optimally and the expected RTP RH has to be formed based on beliefs about RTP RH. The bluffer exploits the opportunity provided by the fragility of the expectation.

The driving force behind bluffs is greed because bluffers do not work hard to get profits by producing and selling better goods and services more cheaply, but by disseminating contaminated information or acting in such a way to mislead people.

References:

- [1] Antonelli, Giovanni Battista (1886). *Sulla Teoria Matematica Della Economia Politica*, Pisa, Tipografia del Falchetto.
- [2] Barsky, R.B., Sims, E.R. (2012). Information, Animal Spirits, and the Meaning of Innovations in Consumer Confidence, *American Economic Review*, 102(4): 1343-77.
- [3] Becker, Gary S., Mulligan, C. (1997). The Endogenous Determination of Time Preference, *Quarterly Journal of Economics*, 112(3): 729–758.
- [4] Becker, R.A. (1980). On the Long-run Steady State in a Simple Dynamic Model of Equilibrium with Heterogeneous Households, *The Quarterly Journal of Economics*, 95(2): 375–382.
- [5] Blanchard, O.J., Watson M.W. (1982). Bubbles, Rational Expectations, and Financial Markets in Crises in the Economic and Financial Structure (Paul Wachtel ed.), Lexington Press, Lexington, Mass.
- [6] Böhm-Bawerk, Eugen von (1889). *Capital and Interest*, Reprinted by Libertarian Press, South Holland, IL, 1970.
- [7] Das, Mausumi. (2003). Optimal Growth with Decreasing Marginal Impatience, *Journal of Economic Dynamics & Control*, 27(10): 1881–1898.
- [8] Epstein, Larry G. and J. Allan Hynes (1983). The Rate of Time Preference and Dynamic Economic Analysis, *Journal of Political Economy*, 91(4): 611-635.
- [9] Fisher, I. (1930). *The Theory of Interest*, Macmillan, New York.
- [10] Frederick, Shane, George Loewenstein and Ted O'Donoghue (2002). Time Discounting and Time Preference: A Critical Review, *Journal of Economic Literature*, 40(2): 351-401.
- [11] Harashima, Taiji (2004). A More Realistic Endogenous Time Preference Model and the Slump in Japan, *EconWPA Working Papers*, ewp-mac0402015.
- [12] Harashima, Taiji (2010). Sustainable Heterogeneity: Inequality, Growth, and Social Welfare in a Heterogeneous Population, *MPRA* (*The Munich Personal RePEc Archive*) *Paper*, No. 24233.

- [13] Harashima, Taiji (2014a). The Representative Household Assumption RequiresSustainable Heterogeneity in Dynamic Models, *MPRA* (*The Munich Personal RePEc Archive*) Paper, No. 57520.
- [14] Harashima, Taiji (2014b). Time Preference Shocks, MPRA (The Munich Personal RePEc Archive) Paper, No. 60205.
- [15] Lawrance, Emily C. (1991). Poverty and the Rate of Time Preference: Evidence from Panel Data, Journal of Political Economy, 99(1): 54–77.
- [16] Lucas, Robert E., Jr., Stokey, N.L (1984). Optimal Growth with Many Consumers, Journal of Economic Theory, 32: 139-171.
- [17] Obstfeld, M. (1990). Intertemporal Dependence, Impatience, and Dynamics, *Journal of Monetary Economics*, 26(1): 45-75.
- [18] Samuelson, P. (1937). A Note on Measurement of Utility, *Review of Economic Studies*, 4(2): 155– 161.
- [19] Samwick, A.A. (1998). Discount Rate Heterogeneity and Social Security Reform, Journal of Development Economics, 57(1): 117–146.
- [20] Santos, M.S., Woodford, M. (1997). Rational Asset Pricing Bubbles, *Econometrica*, 65(1): 19–57.
- [21] Uzawa, H. (1968). Time Preference, the Consumption Function, and Optimal Asset Holdings, in Value, Capital, and Growth: Papers in Honour of Sir John Hicks (J. N. Wolfe ed.), University of Edinburgh Press, Edinburgh, Scotland.
- [22] Ventura, L. (2003). Direct Measure of Time-preference, *Economic and Social Review*,34(3): 293– 310.

APPENDIX

A1. The representative household in dynamic models

A1.1 The assumption of the representative household

The concept of the representative household is a necessity in macroeconomic studies. It is used as a matter of course, but its theoretical foundation is fragile. The representative household has been used given the assumption that all households are identical or that there exists one specific individual household, the actions of which are always average among households (I call such a household "the average household" in this paper). The assumption that all households are identical seems to be too strict; therefore, it is usually assumed explicitly or implicitly that the representative household is the average household. However, the average household can exist only under very strict conditions. Antonelli (1886) showed that the existence of an average household requires that all households have homothetic and homogeneous utility functions. This type of utility function is not usually assumed in macroeconomic studies because it is very restrictive and unrealistic. If more general utility functions are assumed, however, the assumption of the representative household as the average household is inconsistent with the assumptions underlying the utility functions.

Nevertheless, the assumption of the representative household has been widely used, probably because it has been believed that the representative household can be interpreted as an approximation of the average household. Particularly in static models, the representative household can be seen to approximate the average household. However, in dynamic models, it is hard to accept the representative household as an approximation of the average household because, if RTPs of households are heterogeneous, there is no steady state where all of the optimality conditions of the heterogeneous households are satisfied (Becker, 1980). Therefore, macroeconomic studies using dynamic models are fallacious if the representative household is assumed to approximate the average household.

A1.2 The representative household in static models

Static models are usually used to analyze comparative statics. If the average household is represented by one specific unique household for any static state, there will be no problem in assuming the representative household as an approximation of the average household. Even though the average household is not always represented by one specific unique household in some states, if the average household is always represented by a household in a set of households that are very similar in preferences and other features, then the representative household assumption can be used to approximate the average household.

Suppose, for simplicity, that households are heterogeneous such that they are identical except for a particular preference. Because of the heterogeneous preference, household consumption varies. However, levels of consumption will not be distributed randomly because the distribution of consumption will correspond to the distribution of the preference. The consumption of a household that has a very different preference from the average will be very different from the average household consumption. Conversely, it is likely that the consumption of a household that has the average preference will nearly have the average consumption. In addition, the order of the degree of consumption will be almost unchanged for any static state because the order of the degree of the preference does not change for the given state.

If the order of consumption is unchanged for any given static state, it is likely that the household with consumption that is closest to the average consumption will also always be a household belonging to a group of households that have very similar preferences. Hence, it is possible to argue that, approximately, one specific unique household's consumption is always average for any static state. Of course, it is possible to show evidence that is counter to this argument, particularly in some special situations, but it is likely that this conjecture is usually true in normal situations, and the

assumption that the representative household approximates the average household is acceptable in static models.

A1.3 The representative household in dynamic models

In dynamic models, however, the story is more complicated. In particular, heterogeneous RTPs pose a serious problem. This problem is easily understood in a dynamic model with exogenous technology (i.e., a Ramsey growth model). Suppose that households are heterogeneous in RTP, degree of risk aversion (ε), and productivity of the labor they provide. Suppose also for simplicity that there are many "economies" in a country, and an economy consists of a household and a firm. The household provides labor to the firm in the particular economy, and the firm's level of technology (A) varies depending on the productivity of labor that the household in its economy provides. Economies trade with each other: that is, the entire economy of a country consists of many individual small economies that trade with each other.

A household maximizes its expected utility, $E \int_{0}^{\infty} u(c_t) \exp(-\theta t) dt$, subject to $\dot{k}_t = f(k_t) - c_t$, where $u(\bullet)$ is the utility function; $f(\bullet)$ is the production function; θ is RTP; E is the expectation operator; $y_t = \frac{Y_t}{L_t}$, $k_t = \frac{K_t}{L_t}$, and $c_t = \frac{C_t}{L_t}$; $Y_t (\ge 0)$ is output, $K_t (\ge 0)$ is capital input, $L_t (\ge 0)$ is labor input, and C_t (≥ 0) is consumption in period t. The optimal consumption path of this Ramsey-type growth model is

$$\frac{\dot{c}_t}{c_t} = \varepsilon^{-1} \left(\frac{\partial y_t}{\partial k_t} - \theta \right),$$

and at steady state,

$$\frac{\partial y_t}{\partial k_t} = \theta \quad . \tag{A1}$$

Therefore, at steady state, the heterogeneity in the degree of risk aversion (ε) is irrelevant, and the heterogeneity in productivity does not result in permanent trade imbalances among economies because $\frac{\partial y_t}{\partial k_t}$ in all economies is kept equal by market arbitrage. Hence, heterogeneity in the degree of

risk aversion and productivity does not matter at steady state. Therefore, the same logic as that used for static models can be applied. Approximately, one specific unique household's consumption is always average for any time in dynamic models, even if the degree of risk aversion and the productivity are heterogeneous. Thus, the assumption of the representative household is also acceptable in dynamic models even if the degree of risk aversion and the productivity are heterogeneous.

However, Eq. (A1) clearly indicates that heterogeneity in RTP is problematic. As Becker (1980) shows, if RTP is heterogeneous, the household that has the lowest RTP will eventually possess all capital. With heterogeneous RTPs, there is no steady state where all households achieve all of their optimality conditions. In addition, the household with consumption that is average at present has a very different RTP from the household with consumption that is average in the distant future. The consumption of a household that has the average RTP will initially be almost average, but in the future the household with the lowest RTP will be the one with consumption that is almost average. That is, the consumption path of the household that presently has average consumption is notably different from that of the household with average consumption in the future. Therefore, any individual household cannot be almost average in any period and thus cannot even approximate the average household. As a result, even if the representative household is assumed in a dynamic model, its discounted expected utility

$E\int_{0}^{\infty} u(c_t) \exp(-\theta t) dt$ is meaningless, and analyses based on it are fallacious.

If we assume that RTP is identical for all households, the above problem is solved. However, this solution is still problematic because that assumption is not merely expedient for the sake of simplicity; rather, it is a critical requirement to allow for an assumed representative household. Therefore, the rationale for identical RTPs should be validated; that is, it should be demonstrated that identical RTPs are actually and universally observed. RTP is, however, unquestionably not identical among households. Hence, it is difficult to accept the representative household assumption in dynamic models based on the assumption of identical RTP.

The conclusion that the representative household assumption in dynamic models is meaningless and leads to fallacious results is very important, because a huge number of studies have used the representative household assumption in dynamic models. To solve this severe problem, an alternative interpretation or definition of the representative household is needed.

Note that in an endogenous growth model the situation is even more complicated. Because a heterogeneous degree of risk aversion also matters, the assumption of the representative household is more difficult to accept, so an alternative interpretation or definition is even more important when endogenous growth models are used.

A2 Sustainable heterogeneity

A2.1 The model

Suppose that two heterogeneous economies—economy 1 and economy 2—are identical except for their RTPs. Households within each economy are assumed to be identical for simplicity. The population growth rate is zero. The economies are fully open to each other, and goods, services, and capital are freely transacted between them, but labor is immobilized in each economy.

Each economy can be interpreted as representing either a country (the international interpretation) or a group of identical households in a country (the national interpretation). Because the economies are fully open, they are integrated through trade and form a combined economy. The combined economy is the world economy in the international interpretation and the national economy in the national interpretation. In the following discussion, a model based on the international interpretation is called an international model and that based on the national interpretation is called a national model. Usually, the concept of the balance of payments is used only for the international transactions. However, because both national and international interpretations are possible, this concept and terminology are also used for the national models in this paper.

RTP of household in economy 1 is θ_1 and that in economy 2 is θ_2 , and $\theta_1 < \theta_2$. The production function in economy 1 is $y_{1,t} = A^{\alpha} f(k_{1,t})$ and that in economy 2 is $y_{2,t} = A^{\alpha} f(k_{2,t})$, where $y_{i,t}$ and $k_{i,t}$ are, respectively, output and capital per capita in economy *i* in period *t* for *i* = 1, 2; *A* is technology; and $\alpha(0 < \alpha < 1)$ is a constant. The population of each economy is $\frac{L}{2}$; thus, the total for both is *L*, which is

sufficiently large. Firms operate in both economies. The current account balance in economy 1 is τ_t and that in economy 2 is $-\tau_t$. The production functions are specified as:

$$y_{i,t} = A^{\alpha} k_{i,t}^{1-\alpha} ;$$

thus, $Y_{i,t} = K_{i,t}^{1-\alpha} (AL)^{\alpha} (i = 1,2)$. Because *A* is given exogenously, this model is an exogenous technology model (Ramsey growth model). The examination of sustainable heterogeneity based on an endogenous growth model is shown in Harashima (2014a).

Because both economies are fully open, returns on investments in each economy are kept equal through arbitration, such that

Theoretical and Practical Research in Economic Fields

Volume VI, Issue 1(11) Summer 2015

$$\frac{\partial y_{1,t}}{\partial k_{1,t}} = \frac{\partial y_{2,t}}{\partial k_{2,t}} . \tag{A2}$$

Because Eq. (A2) always holds through arbitration, equations $k_{1,t} = k_{2,t}$, $\dot{k}_{1,t} = \dot{k}_{2,t}$, $y_{1,t} = y_{2,t}$, and $\dot{y}_{1,t} = \dot{y}_{2,t}$ also hold.

The accumulated current account balance $\int_{0}^{t} \tau_{s} ds$ mirrors capital flows between the two economies. The

economy with current account surpluses invests them in the other economy. Because $\frac{\partial y_{1,t}}{\partial k_{1,t}} \left(= \frac{\partial y_{2,t}}{\partial k_{2,t}} \right)$

are returns on investments, $\frac{\partial y_{1,t}}{\partial k_{1,t}} \int_0^t \tau_s ds$ and $\frac{\partial y_{2,t}}{\partial k_{2,t}} \int_0^t \tau_s ds$ represent income receipts or payments on

the assets that an economy owns in the other economy. Hence, $\tau_t - \frac{\partial y_{2,t}}{\partial k_{2,t}} \int_0^t \tau_s ds$

is the balance on goods and services of economy 1, and $\frac{\partial y_{1,t}}{\partial k_{1,t}} \int_0^t \tau_s ds - \tau_t$

is that of economy 2. Because the current account balance mirrors capital flows between the economies, the balance is a function of capital in both economies, such that: (I - I - I)

 $\tau_t = \kappa \big(k_{1,t}, k_{2,t} \big) \; .$

The government (or an international supranational organization) intervenes in the activities of economies 1 and 2 by transferring money from economy 1 to economy 2. The amount of transfer in period *t* is g_t , and it is assumed that g_t depends on capital inputs, such that:

 $g_t = \overline{g}k_{1,t}$, where \overline{g} is a constant.

Because $k_{1,t} = k_{2,t}$ and $\dot{k}_{1,t} = \dot{k}_{2,t}$, $g_t = \overline{g}k_{1,t} = \overline{g}k_{2,t}$. Each household in economy 1 therefore maximizes its expected utility:

$$E\!\!\int_0^\infty\!\!u_1\!\!\left(\!c_{1,t}
ight)\!\!\exp\!\left(\!- heta_1\,t
ight)\!dt\,$$
 ,subject to

$$\dot{k}_{1,t} = A^{\alpha} k_{1,t}^{1-\alpha} - c_{1,t} + (1-\alpha) A^{\alpha} k_{1,t}^{-\alpha} \int_{0}^{t} \tau_{s} ds - \tau_{t} - \overline{g} k_{1,t} , \qquad (A3)$$

and each household in economy 2 maximizes its expected utility:

$$E\int_0^\infty u_2(c_{2,t})\exp(-\theta_2 t)dt$$
,

subject to:

$$\dot{k}_{2,t} = A^{\alpha} k_{2,t}^{1-\alpha} - c_{2,t} - (1-\alpha) A^{\alpha} k_{2,t}^{-\alpha} \int_{0}^{t} \tau_{s} ds + \tau_{t} + \overline{g} k_{2,t} \quad ,$$
(A4)

*whereu*_{*i*,*t*} and *c*_{*i*,*t*}, respectively, are the utility function and per capita consumption in economy *i* in period *t* for *i* = 1, 2; and *E* is the expectation operator. Eqs.(A3) and (A4) implicitly assume that each economy does not have foreign assets or debt in period t = 0.

A2.2 Sustainable heterogeneitywithout government intervention

Heterogeneity is defined as being sustainable if all of the optimality conditions of all heterogeneous households are satisfied indefinitely. First, the natures of the model when the government does not intervene (i.e., $\overline{g} = 0$) are examined. The growth rate of consumption in economy 1 is:

$$\frac{\dot{c}_{1,t}}{c_{1,t}} = \varepsilon^{-1} \left\{ (1-\alpha)A^{\alpha}k_{1,t}^{-\alpha} + (1-\alpha)A^{\alpha}k_{1,t}^{-\alpha} \frac{\partial \int_{0}^{t} \tau_{s} ds}{\partial k_{1,t}} - \alpha(1-\alpha)A^{\alpha}k_{1,t}^{-\alpha-1} \int_{0}^{t} \tau_{s} ds - \frac{\partial \tau_{t}}{\partial k_{1,t}} - \theta_{1} \right\} \cdot$$

Hence,

$$\lim_{t\to\infty}\frac{\dot{c}_{1,t}}{c_{1,t}} = \varepsilon^{-1}\lim_{t\to\infty}\left\{ (1-\alpha)A^{\alpha}k_{1,t}^{-\alpha} + (1-\alpha)A^{\alpha}k_{1,t}^{-\alpha}\frac{\partial\int_{0}^{t}\tau_{s}ds}{\partial k_{1,t}} - \alpha(1-\alpha)A^{\alpha}k_{1,t}^{-\alpha-1}\int_{0}^{t}\tau_{s}ds - \frac{\partial\tau_{t}}{\partial k_{1,t}} - \theta_{1} \right\} = 0$$

and thereby

$$\lim_{t\to\infty} (1-\alpha) A^{\alpha} k_{1,t}^{-\alpha} [1+(1-\alpha)\Psi] - \Xi - \theta_1 = 0 ,$$

where
$$\Xi = \lim_{t \to \infty} \frac{\tau_t}{k_{1,t}} = \lim_{t \to \infty} \frac{\tau_t}{k_{2,t}}$$
 and $\Psi = \lim_{t \to \infty} \frac{\int_0^t \tau_s ds}{k_{1,t}} = \lim_{t \to \infty} \frac{\int_0^t \tau_s ds}{k_{2,t}}$. $\lim_{t \to \infty} \frac{\dot{y}_{1,t}}{y_{1,t}} = \lim_{t \to \infty} \frac{\dot{c}_{1,t}}{c_{1,t}} = \lim_{t \to \infty} \frac{\dot{y}_{1,t}}{c_{1,t}} = \lim_{t \to \infty} \frac{\dot{y}_{1,t}}{c_{1,t}}} = \lim_{t \to \infty} \frac{\dot{y}_{1,t}}{c_{1,t}} = \lim_{t$

 $\lim_{t \to \infty} \frac{\dot{k}_{1,t}}{k_{1,t}} = \lim_{t \to \infty} \frac{\dot{\tau}_t}{\tau_t} = 0, \text{ and } \Psi \text{ is constant at steady state because } k_{1,t} \text{ and } \tau_t \text{ are constant; thus,}$ $\Xi = \lim_{t \to \infty} \frac{\tau_t}{k_{1,t}} \text{ is constant at steady state. For } \Psi \text{ to be constant at steady state, it is necessary that}$ $\lim_{t \to \infty} \tau_t = 0 \text{ and thus } \Xi = 0. \text{ Therefore,}$

$$\lim_{t \to \infty} (1-\alpha) A^{\alpha} k_{1,t}^{-\alpha} [1+(1-\alpha)\Psi] - \theta_1 = 0 \quad , \tag{A5}$$

and

$$\lim_{t\to\infty} (1-\alpha) A^{\alpha} k_{2,t}^{-\alpha} [1-(1-\alpha)\Psi] - \theta_2 = 0$$

because

$$\lim_{t \to \infty} \frac{\dot{c}_{2,t}}{c_{2,t}} = \varepsilon^{-1} \lim_{t \to \infty} \left\{ (1-\alpha) A^{\alpha} k_{2,t}^{-\alpha} - (1-\alpha) A^{\alpha} k_{2,t}^{-\alpha} \frac{\partial \int_{0}^{t} \tau_{s} ds}{\partial k_{2,t}} + \alpha (1-\alpha) A^{\alpha} k_{2,t}^{-\alpha-1} \int_{0}^{t} \tau_{s} ds + \frac{\partial \tau_{t}}{\partial k_{2,t}} - \theta_{2} \right\} = 0 \cdot t^{-1} \left\{ \int_{0}^{t} \tau_{s} ds + \frac{\partial \tau_{t}}{\partial k_{2,t}} - \theta_{2} \right\}$$

Because
$$\lim_{t \to \infty} (1 - \alpha) A^{\alpha} k_{1,t}^{-\alpha} [1 + (1 - \alpha) \Psi] = \theta_1,$$

$$\lim_{t \to \infty} (1 - \alpha) A^{\alpha} k_{2,t}^{-\alpha} [1 - (1 - \alpha) \Psi] = \theta_2, \text{ and } \frac{\partial y_{1,t}}{\partial k_{1,t}} = \frac{\partial y_{2,t}}{\partial k_{2,t}} = A^{\alpha} k_{1,t}^{-\alpha} = A^{\alpha} k_{2,t}^{-\alpha}, \text{ then } h_{1,t}^{-\alpha} = A^{\alpha} k_{2,t}^{-\alpha}, \text{ then } h_{1,t}^{-\alpha} = A^{\alpha} k_{2,t}^{-\alpha}$$

$$\Psi = \frac{\theta_1 - \theta_2}{2(1 - \alpha) \lim_{t \to \infty} \frac{\partial y_{1,t}}{\partial k_{1,t}}}$$
 (A6)

By Eqs. (A5) and (A6),

$$\lim_{t\to\infty}\frac{\partial y_{1,t}}{\partial k_{1,t}} + \lim_{t\to\infty}\frac{\partial y_{1,t}}{\partial k_{1,t}} (1-\alpha)\Psi = \theta_1 ;$$

thus,

$$\lim_{t \to \infty} \frac{\partial y_{1,t}}{\partial k_{1,t}} = \frac{\theta_1 + \theta_2}{2} = \lim_{t \to \infty} \frac{\partial y_{2,t}}{\partial k_{2,t}}$$
(A7)

If Eq. (A7) holds, all of the optimality conditions of both economies are indefinitely satisfied. The state indicated by Eq. (A7) is called the "multilateral steady state" or "multilateral state" in the following discussion. By procedures similar to those used for the endogenous growth model in Harashima (2014a), the condition of the multilateral steady state for *H* economies that are identical except for their RTPs is shown as

$$\lim_{t \to \infty} \frac{\partial y_{i,t}}{\partial k_{i,t}} = \frac{\sum_{q=1}^{H} \theta_q}{H}$$
(A8)

for any *i*, where *i* = 1, 2, ..., *H*.Because $\Psi = \frac{\theta_1 - \theta_2}{2(1 - \alpha) \lim_{t \to \infty} \frac{\partial y_{1,t}}{\partial k_{1,t}}} = \frac{\theta_1 - \theta_2}{(1 - \alpha)(\theta_1 + \theta_2)} < 0$ by Eq. (A7), then

 $by \lim_{t \to \infty} \frac{\int_0^t \tau_s ds}{k_{1,t}} = \Psi < 0, \lim_{t \to \infty} \int_0^t \tau_s ds < 0 \text{ that is, economy 1 possesses accumulated debts owed to}$

economy 2 at steady state, and economy 1 has to export goods and services to economy 2 by: $|(1-\alpha)A^{\alpha}k_{1,t}^{-\alpha}\int_{0}^{t}\tau_{s}ds|$ in every period to pay the debts. Nevertheless, because $\lim_{t\to\infty}\tau_{t} = 0$ and $\Xi = 0$, the debts do notexplode but stabilize at steady state. Because of the debts, the consumption of economy 1 is smaller than that of economy 2 at steady state under the condition of sustainable heterogeneity.

Note that many empirical studies conclude that RTP is negatively correlated with income (e.g., Lawrance, 1991; Samwick, 1998; Ventura, 2003). Suppose that, in addition to the heterogeneity in RTP $(\theta_1 < \theta_2)$, the productivity of economy 1 is higher than that of economy 2. At steady state, the consumption of economy 1 would be larger than that of economy 2 as a result of the heterogeneity in productivity. However, as a result of the heterogeneity in RTP, the consumption of economy 1 is smaller than that of economy 2 at steady state under sustainable heterogeneity. Which effect prevails will depend on differences in the degrees of heterogeneity. For example, if the difference in productivity is relatively large whereas that in RTP is relatively small, the effect of the productivity difference will prevail and the consumption of economy 1 will be larger than that of economy 2 at steady state under sustainable heterogeneity.

A3. An alternative definition of the representative household

A3.1 The definition

Section A2 indicates that, when sustainable heterogeneity is achieved, all heterogeneous households are connected (in the sense that all households behave by considering other households' optimality) and appear to be behaving collectively as a combined supra-household that unites all households, as Eqs. (A7) and (A8) indicate. The supra-household is unique and its behavior is time-consistent. Its actions always and consistently represent those of all households. Considering these
natures of households under sustainable heterogeneity, I present the following alternative definition of the representative household: "the behavior of the representative household is defined as the collective behavior of all households under sustainable heterogeneity."

Even if households are heterogeneous, they can be represented by a representative household as defined above. Unlike the representative household defined as the average household, the collective representative household reaches a steady state where all households satisfy all of their optimality conditions in dynamic models. In addition, this representative household has a RTP that is equal to the average RTP as shown in Eqs. (A7) and (A8).²¹ Hence, we can assume not only a representative household but also that its RTP is the average rate of all households.

A3.2 Universality of sustainable heterogeneity

An important point, however, is that this alternatively defined representative household can be used in dynamic models only if sustainable heterogeneity is achieved, but this condition is not necessarily always naturally satisfied. Sustainable heterogeneity is achieved only if households with lower RTPs behave multilaterally or the government appropriately intervenes. Therefore, the representative household assumption is not necessarily naturally acceptable in dynamic models unless it is confirmed that sustainable heterogeneity is usually achieved in an economy.

Notwithstanding this flaw, the representative household assumption has been widely used in many macroeconomic studies that use dynamic models. Furthermore, these studies have been little criticized for using the inappropriate representative household assumption. In addition, in most economies, the dire state that Becker (1980) predicts has not been observed even though RTPs of households are unquestionably heterogeneous. These facts conversely indicate that sustainable heterogeneity—probably with government interventions—has been usually and universally achieved across economies and time periods. In a sense, these facts are indirect evidence that sustainable heterogeneity usually prevails in economies.

Note that because the representative household's behavior in dynamic models is represented by the collective behavior of all households under sustainable heterogeneity, RH's RTP is not intrinsically known to households, but they do need to have an expected rate. Each household intrinsically knows its own preferences, but it does not intrinsically know the collective preference of all households. Therefore, in dynamic models, it must be assumed that all households do not *ex ante* know RH's RTP, but households estimate it from information on the behaviors of other households and the government.

A4 Need for an expected RTP RH

A4.1 The behavior of household

Achieving sustainable heterogeneity affects the behavior of the individual household because sustainable heterogeneity indicates that each household must consider the other households' optimality (as well as the behavior of the government, if necessary). This feature does not mean that households behave cooperatively with other households. Each household behaves non-cooperatively based on its own RTP, but at the same time, it behaves considering whether the other households' optimality conditions are achieved or not. This consideration affects the actions a household takes in that it affects the choice of a household's initial consumption.

Sustainable heterogeneity indicates that a household's future path of consumption has to be consistent with the future path of sustainable heterogeneity. Thereby, a household sets its initial consumption such that it will proceed on the path that is consistent with the path of sustainable heterogeneity and eventually reach a steady state.

²¹ If sustainable heterogeneity is achieved with the help of the government's intervention, the time preference rate of the representative household will not be exactly equal to the average rate of time preference.

A4.2 Deviation from sustainable heterogeneity

A4.2.1 Political elements

What happens if a household deviates from sustainable heterogeneity? A deviation means that a household sets its initial consumption at a level that is not consistent with sustainable heterogeneity. For less advantaged households (i.e., households with higher RTPs), the only way to satisfy all of their optimality conditions is to set their initial consumption consistent with sustainable heterogeneity.

Therefore, they will not take the initiative to deviate. In contrast, the most advantaged households (i.e., those with the lowest RTP) can satisfy all of their optimality conditions even if they set initial consumption independent of sustainable heterogeneity. The incentive for the most advantaged household to select a multilateral path will be weak because the growth rate of the most advantaged household on the multilateral path is lower than that on the unilateral path.

When economy 1 selects the unilateral path, does economy 2 quietly accept the unfavorable consequences shown in Becker (1980)? From an economic perspective, the optimal response of economy 2 is the one shown in Harashima (2010): economy 2 should behave as a follower and accept the unfavorable consequences. However, if other factors—particularly political ones—are taken into account, the response of economy 2 will be different. Faced with a situation in which all the optimality conditions cannot be satisfied, it is highly likely that economy 2 would politically protest and resist economy 1. It should be emphasized economy 2 is not responsible for its own non-optimality, which is a result of economy 1's unilateral behavior in a heterogeneous population. Economy 2 may overlook the non-optimality if it is temporary, but it will not if it is permanent. As shown in Harashima (2010), the non-optimality is permanent, it is quite likely that economy 2 will seriously resist economy 1 politically.

If economy 1 could achieve its optimality only on the unilateral path, economy 1 would counter the resistance of economy 2, but this is not the case. Because of this, economy 2's demand does not necessarily appear to be unreasonable or selfish. Faced with the protest and resistance by economy 2, economy 1 may compromise or cooperate with economy 2 and select the multilateral path.

A4.2.2 Resistance

The main objective of economy 2 is to force economy 1 to select the multilateral path and to establish sustainable heterogeneity. This objective may be achieved through cooperative measures, non-violent civil disobedience (e.g., trade restrictions), or other more violent means.

Restricting or abolishing trade between the two economies will cost economy 1 because it necessitates a restructuring of the division of labor, and the restructuring will not be confined to a small scale. Large-scale adjustments will develop that involve all levels of divided labor, because they are all correlated with each other. For example, if an important industry had previously existed only in one economy, owing to a division of labor, and trade between the two economies was no longer permitted, the other economy would have to establish this industry while also maintaining other industries. As a result, economy 1 would incur non-negligible costs. More developed economies have more complicated and sophisticated divisions of labor, and restructuring costs from the disruption of trade will be much higher in developed economies. In addition, more resources will need to be allocated to the generation of technology because technology will also no longer be traded. Finally, all of the conventional benefits of trade will be lost. Trade is beneficial because of the heterogeneous endowment of resources, as the Heckscher-Ohlin theorem shows. Because goods and services are assumed to be uniform in the models presented in this paper, the benefits of trade are implicit in the models. However, in the real word, resources such as oil and other raw materials are unevenly distributed, so a disruption or restriction of trade will substantially damage economic activities on both national and international levels.

The damage done by trade restrictions has an upper limit, however, because the restructuring of the division of labor, additional resource allocation to innovation, and loss of trade benefits are all

finite. Therefore, in some cases, particularly if economies are not sufficiently developed and division of labor is not complex, the damage caused will be relatively small. Hence, a disruption of trade (non-violent civil disobedience in the national models) may not be sufficiently effective as a means of resistance under some these conditions.

In some cases, harassment, sabotage, intimidation, and violence may be used, whether legal or illegal. In extreme cases, war or revolution could ensue. In such cases, economy 1 will be substantially damaged in many ways and be unable to achieve optimality. The resistance and resulting damages will continue until sustainability is established.

In any case, the objective of economy 2's resistance conversely implies that establishing sustainability eliminates the risk and cost of political and social instability. The resistance of economy 2 will lower the desire of economy 1 to select the unilateral path.

A4.2.3 United economies

An important countermeasure to the fragility of sustainable heterogeneity for less advantaged economies is the formation of a union of economies. If economies other than economy 1 are united by commonly selecting the multilateral path within them, their power to resist economy 1 will be substantially enhanced. Consider the multi-economy model shown in Harashima (2010). If the economies do not form a union, the power to resist the unilateral actions of economy 1 is divided and limited to the power of each individual economy. However, if the economies are united, the power to resist economy 1 increases. If a sufficient number of economies unite, the multilateral path will almost certainly be selected by economy 1.

To maintain the union, any economy in the union should have the explicit and resolved intention of selecting the multilateral path within the union, even if it is relatively more advantaged within the union. To demand that relatively more advantaged economies select the multilateral path, less advantaged economies themselves must also select the multilateral path in any case. Otherwise, less advantaged economies will be divided and ruled by more advantaged economies. For all heterogeneous people to happily coexist, all of them should behave multilaterally. At the same time, Harashima (2010) indicates that the more advantaged an economy is, the more modestly it should behave, i.e., the more it should restrain itself from accumulating extra capitals.

In general, therefore, the most advantaged (the lowest RTP) household will be forced to set its initial consumption consistent with sustainable heterogeneity.

A4.3 Need for an expected RTP RH

Because all households need to set their initial consumption consistent with sustainable heterogeneity to achieve it, households must calculate the path of sustainable heterogeneity before setting their initial consumption levels. To calculate this level, each household first must know the value of RTP RH. However, although a household naturally knows the value of its own RTP, it does not intrinsically know the value of RTP RH. To know this, a household would have to know the values of all of the other households' RTPs. Hence, the expected value of RTP RHmust somehow be generated utilizing all other relevant available information. The necessity of an expected RTP RHis critically important because RTP plays a crucial role as the discount factor in dynamic models.

Note that, if we assume that RTP is identical for all households, an expected RTP RHis no longer needed because any household's own RTP is equal to the RTP RH. This solution is still problematic, however, because the assumption is not merely expedient for the sake of simplicity; rather, it is a critical requirement to eliminate the need for an expected RTP RH. Therefore, any rationale for assuming identical RTPs should be validated; that is, it should be demonstrated that identical RTPs do exist and are universally observed. However, RTP is unquestionably not identical among households. Therefore, households must use expected values of RTP RH.

A5. The RTP model

A5.1 Need to know the structural model

If RTP RHis a constant parameter, as has been long and widely assumed, the need for an expected RTP RH would not be a serious problem. The historical mean of an unchanging RTP RH could be estimated relatively precisely based on long-term data of various economic indicators even if the structural model remained unknown. The RTP RH could be specified as the RTP that is most consistent with long-term trends of the indicators.

Although RTP has been treated as a constant parameter in many studies, this feature has not been demonstrated either empirically or theoretically. Rather, the assumption is merely expedient for the sake of simplicity. There is another practical reason for this treatment: models with a permanently constant RTP exhibit excellent tractability (see Samuelson, 1937). However, some have argued that it is natural to view RTP as temporally variable, and the concept of a temporally varying RTP has a long history (e.g., Böhm-Bawerk, 1889; Fisher, 1930). More recently, Lawrance (1991) and Becker and Mulligan (1997) showed that people do not inherit permanently constant RTPs by nature and that economic and social factors affect the formation of RTPs. Their arguments indicate that many incidents can affect and change RTP. Models of endogenous RTP have been presented, the most familiar of which is Uzawa's (1968) model.

If the RTP RH is temporally variable, its future stream must be expected by households, and a rational expectation is a model-consistent expectation. To generate rational expectations of RTP RH, therefore, the structural model of the RTP RH (i.e., equations that fundamentally describe how it is endogenously formed) needs to be known.

A5.2 Endogenous RTP models

A5.2.1 Uzawa's (1968) model

The most well-known endogenous RTP model is that of Uzawa (1968). It has been applied in many analyses (e.g., Epstein and Hynes, 1983; Lucas and Stokey, 1984; Epstein, 1987; Obstfeld, 1990). However, Uzawa's model has not necessarily been regarded as a realistic expression of the endogeneity of RTP because it has a serious drawback in that impatience increases as income, consumption, and utility increase.

The basic structure of Uzawa's model is $\theta_t = \theta^*[u(c_t)]$, $0 < \frac{d\theta_t}{du(c_t)}$, in which RTP in period $t(\theta_t)$

is temporally variable and an increasing function of present utility $u(c_t)$ where c_t is consumption in period t. The condition $o_{t,t} = \frac{d\theta_t}{dt}$ is necessary for the model to be stable.

t. The condition $0 < \frac{d\theta_t}{du(c_t)}$ is necessary for the model to be stable.

This property is quite controversial and difficult to accept *a priori* because many empirical studies have indicated that RTP is negatively correlated with permanent income (e.g., Lawrance, 1991); thus, many economists are critical of Uzawa's model. Epstein (1987), however, discussed the plausibility of increasing impatience and offered some counter-arguments. However, his view is in the

minority, and most economists support arguments in favor of a decreasing RTP, such that $\frac{d\theta_{t}}{du(c_{t})} < 0$.

Hence, although Uzawa's model attracted some attention, the analysis of the endogeneity of RTP has progressed very little. Although Uzawa's model may be flawed, it does not mean that the conjecture that RTP is influenced by future income, consumption, and utility is fallacious. Rather, it means that an appropriate model in which RTP is negatively correlated with income, consumption, and utility has not been presented.

A5.2.2 Size effect on impatience

The problem of $0 < \frac{d\theta_t}{du(c_t)}$ in Uzawa's model arises because distant future levels of

consumption have little influence on factors that form RTP; that is, RTP is formed only with the information on present consumption, and it must be revised every period in accordance with consumption growth. However, there is no *a priori* reason why information on distant future activities should be far less important than the information on the present and near future activities. Fisher (1930) argued that our first step, then, is to show how a person's impatience depends on the *size* of his income, assuming the other three conditions to remain constant; for, evidently, it is possible that two incomes may have the same time shape, composition and risk, and yet differ in size, one being, say, twice the other in every period of time.

In general, it may be said that, other things being equal, the smaller the income, the higher the preference for the present over the future income. It is true of course that a permanently small income implies a keen appreciation of wants as well as of immediate wants. But it increases the want for immediate income *even more* than it increases the want for future income. (p. 72)

According to Fisher's (1930) view, a force that influences RTP is a psychological response derived from the perception of the "size of the entire income or utility stream." This view indicates that it is necessary to probe how people perceive the size of the entire income or utility stream.

Little effort has been directed toward probing the nature of the size of the utility or income stream on RTP, although numerous psychological experiments have been performed with regard to the anomalies of the expected utility model with a constant RTP (e.g., Frederick *et al.*, 2002). Analyses using endogenous RTP models so far have merely introduced the *a priori* assumption of endogeneity of RTP without explaining the reasoning for doing so in detail. Hence, even now, Fisher's (1930) insights are very useful for the examination of the size effect. An important point in Fisher's quote is that the size of the infinite utility stream is perceived as "permanently" high or low. The size difference among the utility streams may be perceived as a permanently continuing difference of utilities among different utility streams. Anticipation of a permanently higher utility may enhance an emotional sense of well-being because people feel they are in a long-lasting secure situation, which will generate a positive psychological response and make people more patient. If that is true, distant future utilities should be taken into account equally with present utility. Otherwise, it is impossible to distinguish whether the difference of utilities will continue permanently.

From this point of view, the specification that only the present utility influences the formation of RTP, as is the case of Uzawa's model, is inadequate. Instead, a simple measure of the size where present and future utilities are summed with equal weight will be a more appropriate measure of the size of a utility stream.²²

A5.3 Model of RTP23

A5.3.1 The model

The representative household solves the maximization problem as shown in Section A1.3. Taking the arguments in Section A5.2 into account, the "size" of the infinite utility stream can be defined as follows.

Definition 1: The size of the utility stream *W* for a given technology *A* is:

$$W = \lim_{T \to \infty} E \int_0^T \rho(t) u(c_t) dt$$
 ,

²² Das (2003) showed another stable endogenous time preference model with decreasing impatience. Her model is stable, although the rate of time preference is decreasing because endogenous impatience is almost constant. In this sense, the situation her model describes is very special.

²³ The idea of this type of endogenous time preference model was originally presented in Harashima (2004).

whereE is the expectation operator, and

$$\rho(t) = \frac{1}{T} \text{ if } 0 \le t \le T$$

 $\rho(t) = 0$ otherwise $\rho(t)$ indicates weights and has the same value in any period. Thus, the weights for the evaluation of future utilities are distributed evenly over time, as discussed in Section A5.2.

To this point, technology *A* has been assumed to be constant. If *A* is temporally variable (A_t) and grows at a constant rate and the economy is on a balanced growth path such that A_t , y_t , k_t , and c_t grow at the same rate, then the definition of *W* needs to be modified because any stream of c_t and $u(c_t)$ grows to infinity. It is then impossible to distinguish the sizes of the utility stream by simply summing up c_t as $T \rightarrow \infty$ as shown in Definition 1. Because balanced growth is possible only when technological progress is Harrod neutral, I assume a Harrod neutral production function such that:

$$y_t = \omega A_t^{\varpi} k_t^{1-\varpi}$$
 ,

where $\varpi(0 < \varpi < 1)$ and $\omega(0 < \omega)$ are constants. To distinguish the sizes of utility stream, the following value is set as the standard stream of utility,

$$u(\widetilde{c}\;e^{arphi t})$$
 ,

where $\tilde{c}(0 < \tilde{c})$ is a constant and $\psi(0 < \psi)$ is a constant rate of growth. Streams of utility can be compared with this standard stream. If a constant relative risk aversion utility function is assumed, a stream of utility can be compared with the standard stream of utility as follows:

$$\frac{u(c_t)}{u(\widetilde{c}\,e^{\psi t})} = \frac{c_t^{1-\gamma}}{(\widetilde{c}\,e^{\psi t})^{1-\gamma}} = \frac{1-\gamma}{\widetilde{c}^{1-\gamma}}\,u\left(\frac{c_t}{e^{\psi t}}\right) \,.$$

By using this ratio, a given stream of utility can be distinguished from the standard stream of utility. That is, the size of a utility stream *W* for a given stream of technology A_t that grows at the same rate ψ as y_t , k_t , and c_t can be alternatively defined as:

$$W = \lim_{T \to \infty} E \int_0^T \rho(t) u\left(\frac{c_t}{e^{\psi t}}\right) dt$$

Clearly, if $\psi = 0$, then the size (*W*) degenerates into the one shown in Definition 1.If there is a steady state such that:

$$\lim_{t \to \infty} E[u(c_t)] = E[u(c^*)] \text{ ,or for the case of expected balanced growth } \lim_{t \to \infty} E\left[u\left(\frac{c_t}{e^{\psi t}}\right)\right] = E[u(c^*)] \text{ ,}$$

where c^* is a constant and indicates steady-state consumption, then $W = E[u(c^*)]$ for the following reason.

Because
$$\lim_{t \to \infty} E[u(c_t)] = E[u(c^*)]$$
 or $\lim_{t \to \infty} E\left[u\left(\frac{c_t}{e^{\psi t}}\right)\right] = E[u(c^*)]),$

then

$$\begin{split} &\lim_{T\to\infty} \int_0^T \rho(t) \{ E[u(c^*)] - E[u(c_t)] \} dt = E[u(c^*)] - W \quad \text{or} \\ &\lim_{T\to\infty} \int_0^T \rho(t) \{ E[u(c^*)] - E[u(\frac{c_t}{e^{\psi t}})] \} dt = E[u(c^*)] - W) . \\ &\ln addition, \\ &\lim_{T\to\infty} \int_0^T \rho(t) \{ E[u(c^*)] - E[u(c_t)] \} dt = 0 \quad \text{or} \end{split}$$

$$\lim_{T\to\infty}\int_0^T \rho(t)\left\{E\left[u(c^*)\right]-E\left[u\left(\frac{c_t}{e^{\psi t}}\right)\right]\right\}dt=0).$$

Hence, $W = E[u(c^*)]$; that is, RTP is determined by steady-state consumption (c*). The RTP model presented in this paper is constructed on the basis of this measure of W. An essential property that must be incorporated into the model is that RTP is sensitive to, and a function of, W such that

$$\theta = \theta^{**}(W)$$

where $\theta^{**}(W)$ is monotonically continuous and continuously differentiable. Because W is a sum of utilities, this property simply reflects the core idea of an endogenous RTP. However, this property is new in the sense that RTP is sensitive not only to the present utility but also to the entire stream of utility, that is, the size of the utility stream represented by the utility of steady-state consumption. This property is intuitively acceptable because it is likely that people set their principles or parameters for their behaviors considering the final consequences of their behavior (i.e., the steady state; see, e.g., Barsky and Sims, 2012).

Another essential property that must be incorporated into the model is:

$$\frac{d\theta}{dW} < 0$$

Because $W = E[u(c^*)]$ and $0 < \frac{du(c_i)}{dc_i}$, RTP is inversely proportionate to c^* . This property is consistent

with the findings in many empirical studies, which have shown that RTP is negatively correlated with permanent income (e.g., Lawrance, 1991).

In summary, the basic structure of the model is:

$$\theta = \theta^{**}(W) = \theta^{**} \left\{ E[u(c^*)] \right\},$$

$$\frac{d\theta}{dW} = \frac{d\theta}{dE[u(c^*)]} < 0$$
(A9)

This model is deceptively similar to Uzawa's endogenous RTP model and simply replaces c_t with c^* and $0 < \frac{d\theta_t}{du(c_t)}$ with $\frac{d\theta}{dE[u(c^*)]} < 0$. However, the two models are completely different because of

the opposite characteristics of:

$$0 < \frac{d\theta_t}{du(c_t)}$$
 and $\frac{d\theta}{dE[u(c^*)]} < 0$.

A5.3.2 Nature of the model

The model can be regarded as successful only if it exhibits stability. In Uzawa's model, the economy becomes unstable if:

$$0 < \frac{d\theta_t}{du(c_t)}$$
 is replaced with $\frac{d\theta_t}{du(c_t)} < 0$.

In this section, I examine the stability of the model.

A5.3.2.1 Equilibrium RTP

In Ramsey-type models, such as shown in Section A1.3, if a constant RTP is given, the value

of the marginal product of capital (i.e., the value of the real interest rate) converges to that of the given RTP as the economy approaches the steady state. Hence, when a RTP is specified at a certain value, the corresponding expected steady-state consumption is uniquely determined. Given fixed values of other exogenous parameters, any predetermined RTP has unique values of expected consumption and utility at steady state. There is a one-to-one correspondence between the expected utilities at steady state and the RTPs; therefore, the expected utility at steady state can be expressed as a function of RTP. Let c_x^* be a set of steady-state consumption levels, given a set of RTPs (θ_X) and other fixed exogenous parameters. The concept of $\theta \rightarrow W$ discussed above can be described as

$$g(\theta) = E[u(c^*)](=W), \tag{A10}$$

where $c^* \in c_x^*$ and $\theta \in \theta_x$. On the other hand, RTP is a continuous function of steady-state consumption as shown in Eq. (A9) such that $\theta = \theta^{**}(W) = \theta^{**} \{ E[u(c^*)] \}$. The reverse function is

$$h(\theta) = E[u(c^*)](=W)$$
(A11)

The equilibrium RTP is determined by the point of intersection of the two functions, $g(\theta)$ and $h(\theta)$, as shown in Figure A1. Figure A2 shows the special but conventionally assumed case for $h(\theta)$ in which θ is not sensitive to W, and RTP is constant. There exists a point of intersection because both $g(\theta)$ and $h(\theta)$ are monotonically continuous for $\theta > 0$. $h(\theta)$ is monotonically continuous because $\theta^{**}(W)$ is monotonically continuous. $g(\theta)$ is monotonically continuous because, as a result of utility maximization, $c^* = f(k^*)$ and $\theta = \frac{df(k^*)}{dk^*}$, where k^* is capital input per capita at steady state such that $k^* = \lim_{t \to \infty} (k_t)$. Because $f(k^*)$ and $\frac{df(k^*)}{dk^*}$ are monotonically continuous for $k^* > 0$, c^* is a monotonically continuous function of θ for $\theta > 0$. Here, because u is monotonically continuous, then $E[u(c^*)] = g(\theta)$ is also monotonically continuous for $\theta > 0$.



θ

Figure A1. Endogenous time preference



Figure A2. Permanently constant time preference

The function $g(\theta) = E[u(c^*)] = W$ is a decreasing function of θ because higher RTP results in lower steady state consumption. The function $h(\theta) = E[u(c^*)] = W$ is also a decreasing function of θ because $\frac{d\theta}{dW} < 0$. Thus, both $g(\theta)$ and $h(\theta)$ are decreasing, but the slope of $h(\theta)$ is steeper than that of $g(\theta)$ as shown in Figure A1. This is true because $g(\theta) = W$ is the consequence of a Ramsey-type model as shown in Section A1.3; thus, if $\theta \to \infty$, then $g(\theta) = W \to 0$ because $\theta = i_t \to \infty$ and $k_t \to 0$, and if $\theta \to 0$, then $g(\theta) = W \to \infty$ because $\theta = i_t \to 0$ and $k_t \to \infty$. The function $h(\theta) = W$ indicates the endogeneity of RTP, and because RTP is usually neither zero nor infinity, then even if $h(\theta) = W \to 0$, $\theta < \infty$, and $h(\theta) = W \to \infty$, $0 < \theta$. Hence, the locus $h(\theta) = W$ is more vertical than $g(\theta) = W$, and thereby a permanently constant RTP, as shown in Figure 2, has probably been used as an approximation of the locus $h(\theta) = W$ for simplicity.

A5.3.2.2 Stability of the model

RTP is constant unless a shock that changes the expected c^* occurs because W does not depend on t but on the expected c^* . Thus, the same RTP and steady state continue until such a shock hits the economy. Therefore, the endogeneity of RTP only matters when a shock occurs. This constancy is the key for the stability of the model. Once the RTP corresponding to the intersection (Figure 1) is determined, it is constant and the economy converges at a unique steady state unless a shock that changes the expected c^* occurs. The shock is exogenous to the model, and the economy does not explode endogenously but stabilizes at the steady state. Hence, the property $\frac{d\theta}{dW} < 0$ in the model, which is consistent with empirical findings, does not causes instability. The model is therefore

model, which is consistent with empirical findings, does not causes instability. The model is therefore acceptable as a model of endogenous RTP. Furthermore, because RTP is endogenously determined, the assumption of irrationality is not necessary for the determination of RTP. Nevertheless, a shock on RTP can be initiated by a shock on the expected c^* ; thus, even if the so-called animal spirits are directly

irrelevant to determination of RTP, they may be relevant in the generation of shocks on the expected c^* .

A6 Frequent RTP shocks

A6.1 Difficulty in knowing RTP RH

To estimate the parameter values of Eq. (A11) in the structural model of RTP RH, it is necessary to obtain a sufficiently large amount of data on the value of RTP RH. To obtain these data, a household must know the RTPs of all the other households. Although a household knows its own RTP, it has almost no information about the RTPs of all the other households much less time-series data on each household's RTP. Because of the lack of available data, a household cannot estimate the parameter values in Eq. (A11) in the structural model of RTP RHeven if it knows the functional forms of equations in the structural model. We can easily generate data on aggregate consumption, investment, production, inflation, trade, and other factors at a relatively low cost, but we cannot directly observe the value of RTP RH. Nonetheless, many estimates of RTP have been reported, but they are not based on a structural model of RTP. Most are the results of experimental studies or indirect estimates based on other models (e.g., Ramsey growth models) on the assumption that RTP is constant. Experiments can give us some information on the RTPs of test subjects, but we should not naively use these estimates as the RTP RHin the calculation of the future path of economy because they vary widely according to the experimental environments. Furthermore, most of the indirect estimates were calculated on the assumption that RTP is constant, which as discussed previously, is most likely not the case. The basic problem is that no credible estimation method of RTP RHhas been established.

A6.2 Expectations based on beliefs

The lack of observable data on RTP RH will significantly hinder households from generating rational expectations of the future path of economy. How do households rationally expect their future streams of consumption and production and calculate their optimal paths without information on RTP RH, which is indispensable as the discount factor? The historical mean of RTP RHestimated by long-term data is not consistent with a rational expectation of the future stream because RTP is not constant. Without a reliable method for estimating the parameters of the structural model, it is impossible for households to generate rational expectations of the future path of the economy.

An alternative way of estimating expected values of RTP RH is needed, but even if an alternative method is utilized, households still have to behave as rationally as possible even in an environment of significantly incomplete information. In this situation, household may have to use the concept of bounded rationality to make decisions. It is possible that the only alternative for a household is to use its "belief" about the RTP RH. The use of a belief does not mean that households deviate from rationality; rather, it is the most rational behavior they can use in an environment where insufficient information is available.

Such a belief is defined in this paper as the range of values of RTP RH within which a household believes that the true RTP RH exists. Households utilize the belief in place of Eq. (A11). More specifically, suppose that household *i* ($i \in N$) believes that the RTP RH in the future is situated in the range λ_i , where the subjective probability density at any point on λ_i is identical (i.e., its distribution shape is uniform). Because households have no information about the shape of the distribution, they assume that it is uniform. This supposition means that household *i* believes that λ_i stationary. Let $\overline{\lambda_i}$ be the mean of λ_i . Suppose that household *i* calculate its optimal future path on the belief that the mean of future values of RTP RH is $\overline{\lambda_i}$. By Eq. (A10), *W* can be calculated based on $\overline{\lambda_i}$, and the expected future path of economy can be calculated.

Households can equally access all relevant information. Therefore, if the belief of a household is very different from those of the majority, the household will soon perceive that its belief is different, through observing the behavior of majority. The household will change its belief to the almost same as

those of the majority because otherwise it cannot achieve optimality as expected on the assumption that sustainable heterogeneity is achieved. Hence, it is likely that households' beliefs become similar, and thereby, it is assumed for simplicity that households' beliefs are identical.

Note that households do not cooperatively and collectively expect the future path of economy (i.e., the representative household's future path), but each household independently and individually generates its own expectations based on its belief in RTP RH. The household thereby creates its own expected future path considering the expected representative household's future path. The aggregates are the sum of all household's independent and individual activities, but if sustainable heterogeneity is achieved, the aggregates appear to be the same as the results of the representative household's activities.

A6.3 Refining beliefs

A household knows that its expectation is based on its beliefs and not the structural model. Therefore, it will always want to refine the belief, that is, raise the probability that the belief is the correct value, by exploiting all currently available relevant information. Let a set of currently available economic indicators be I_t (e.g., the observed data on consumption, production, inventory, etc.). These data may provide some useful information on the past RTP RH, and a household may refine its belief based on this information. These data and Eq. (A10) can be used to generate estimates of past values of RTP RH. However, I_t includes noise, and data in I_t will usually be somewhat inconsistent between the elements of I_t . In addition, because Eq. (A10) indicates the steady state values that are achieved after a long-period transition, the short-term past data included in I_t are basically insufficient to obtain a credible estimate. Therefore, the estimate of the past values of RTP RH based on I_t and Eq. (A10) will usually have a large confidence interval. Let $\overline{\mu}_I$ be the estimated past RTP RH and μ_I be its confidence interval of, for example, 95%. Because households can equally access all relevant information, assume for simplicity that μ_I and $\overline{\mu}_I$ are identical for all households.

Although a household knows that $\overline{\mu}_i$ is not a credible estimate, has a large confidence interval, and is merely an estimate (usually a point estimate) of a past value, it will strive to utilize the information derived from $\overline{\mu}_i$ to refine its beliefs in the future value of RTP RH. Usually $\overline{\mu}_i$ will not be equal to $\overline{\lambda}_i$, but the ranges of λ_i and μ_i may partly overlap. Household *i*may utilize the information from this partial overlap to refine its belief (i.e., information of how λ_i is different from μ_i). $\overline{\mu}_I \neq \overline{\lambda}_i$ indicates that the belief $\overline{\lambda}_i$ is wrong, $\overline{\mu}_i$ is wrong, both are wrong, or both are right if the true past RTP RH is $\overline{\mu}_i$ but the true future RTP RH is $\overline{\lambda}_i$. The belief $\overline{\lambda}_i$ may be wrong because the RTP RH will change in the near future, and $\overline{\mu}_i$ may be wrong because the RTP RH changed during the period in which the data were obtained. In addition, a household knows that μ_i is the result of all households' activities based on their beliefs, not on the true value of RTP RH. These uncertainties arise because households cannot know the parameters of the structural model. Without using the structural model, household *i* cannot judge whether $\overline{\lambda}_i$ is wrong, $\overline{\mu}_i$ is wrong, both are wrong, or both are right. As a result, household *i* will not easily adjust its belief from $\overline{\lambda}_i$ to $\overline{\mu}_i$.

However, it is still likely that information about the difference between λ_i and μ_l can be used to refine the belief. To extract the useful information, the following rules may be used:

Rule 1: if $\overline{\mu}_i$ is included in λ_i , the belief is not adjusted; otherwise, the belief is adjusted from $\overline{\lambda}_i$ to $\overline{\mu}_i$.

Rule 2: if $\overline{\lambda_i}$ is included in μ_l , the belief is not adjusted; otherwise, the belief is adjusted from $\overline{\lambda_i}$ to $\overline{\mu_i}$.

Rule 3: if λ_i and μ_i overlap at or above a specified ratio, the belief is not adjusted; otherwise, the belief

is adjusted from $\overline{\lambda}_i$ to $\overline{\mu}_i$.

The above rules may be seen as a type of adaptive expectation because μ_l indicates the past RTP RH. However, in the situation where the parameters of the structural model of the RTP RH are unknown, it may be seen as rational to utilize the information contained in μ_l by adopting one of these rules.

A6.4 Changing beliefs

However, it does not seem likely that a household will refine its belief following one of the rules shown above because the rules are basically backward looking and will not be adopted as a tool for refining the belief if a household is convinced that the RTP RH is temporally variable. The belief will only be changed if forward-looking information is available, that is, when a household becomes aware of information about the future RTP RH in μ_i . For example, the difference between λ_i and μ_i may reflect an unexpected and large positive technology shock that occurred after the formation of belief λ_i . Because the effects of the technology shock will persist for long periods in the future, household *i* will most likely change its belief. In this case, a household will not simply refine its belief from $\overline{\lambda_i}$ to $\overline{\mu_i}$; it will change to another value that is formed as an entirely new belief. Whether a household changes its belief or not, therefore, will depend not simply on μ_i but on the information the household can extract from μ_i about the future path of the economy. Hence, in some cases, a household will change its belief when new values of μ_i are obtained, but in other cases, it will not, depending on how the household interprets the information contained in μ_i .

A6.5 Heuristics

When a household interprets μ_l , it may also use heuristic methods, for example, a simplified linear reduced form model of RTP RH. Studies of the use of heuristics and bounded rationality in this context would be useful for better understanding the interpretation mechanism of μ_l . There are many possible simplified linear reduced form models of RH's RTP that could be used as heuristic methods although most of them may be ad hoc. Even though such reduced form models are far less credible than a structural model, they may be utilized as a heuristic method of interpreting μ_l by households. Although these types of models may often result in misleading conclusions, they may sometimes provide useful information. For example, if a linear correlation between RTP RH and a financial indicator exists, even if it is weak or temporary, changes in the financial indicator may contain useful information about changes in the RTP RH. Therefore, if a household believes that this correlation exists, it will use this information to interpret μ_l .

A6.6 Frequent RTP shocks

Households must have expected values of RTP RH for sustainable heterogeneity, but as previously discussed, the expectations are not based on the structural model but rather on a belief that is not guaranteed to generate the correct value. In addition, the belief can be influenced by heuristic considerations. These features indicate that the expected values of RTP RH will fluctuate more frequently than the intrinsic RTP RH.

Households' expectations of RTP RH will change when the intrinsic RTP RH shifts, for example, when new information about shocks on the factors that determine Eq. (A10) becomes available. For a given θ , $E[u(c^*)]$ changes if the expectation of future productivity changes. Productivity at the macro level will be influenced by scientific technology, financial technology, social infrastructure, and other factors. If expectations about these factors in the future changes, the expected future productivity and $E[u(c^*)]$ will also change. In addition, even if intrinsic RTP RH does not change, the expected RTP RH will change if a household's belief is altered because of new information

contained in μ_l . Hence, the expected RTP RH can change independently of intrinsic changes in RTP RH. Therefore, even if intrinsic changes in RTP RH occur infrequently, changes in the expected RTP RH may occur more frequently.

A household's expected RTP RH can potentially changeevery time new information on μ_l becomes available if it contains the information that makes beliefs change. Information concerning factors that affect the expected RTP RH will become available frequently, and at least some of the information may be both very important and unexpected. In addition, there will be many disturbances in the fundamental factors that affect Eq. (A10), and many of these disturbances will also cause μ_l to change. As discussed previously, a household may interpret these changes in μ_l as a change in the true RTP RH. Therefore, it is likely that households' expected RTP RH change more frequently than the intrinsic RTP RH, and thereby, that time preference shocks also occur more frequently than previously thought.

Even a small piece of additional information about the belief can significantly change the path of the economy. For example, if many households believe a rumor (whether it is true or not) related to information about the interpretation of μ_l and respond similarly to it, their expectations will be changed in the same direction by the rumor. If all households respond similarly to an untrue rumor and change their expectations equally to an untrue value, the economy will proceed based on the incorrect expectation of RTP RH. The μ_l that is observed a few periods later will follow these wrongly expected values of RTP RH. Upon obtaining new data of μ_l that are consistent with these wrongly expected values, households will judge that their (incorrect) changes were in fact correct. As a result, the incorrect expectations become self-fulfilling. This spurious situation may reach an impasse at some point in the future because the expectations are based not on a structural model but on the (incorrect) beliefs. Households will not anticipate the impasse until the economy reaches it because they believe that the wrongly expected RTP RH (i.e., the currently held belief) is true.



DOI: http://dx.doi.org/10.14505/tpref.v6.1(11).03

The Whole Economy Approach of the Input-Output Model

Fengquan LIU fengquanliu212@gmail.com

Suggested Citation:

Liu, F (2015). The Whole Economy Approach of the Input-Output Model. *Theoretical and Practical Research in Economic Field*, (Volume VI, Summer), 1(11):50-64. DOI:10.14505/tpref.v6.1(11).03. Available from: http://www.asers.eu/journals/tpref/curent-issue.

Article's History:

Received May 2015; Revised July 2015; Accepted August 2015

2015. ASERS Publishing. All rights reserved.

Abstract:

In macroeconomy research that uses technological table, most studies have only considered a special aspect. This paper develops an input-output model that extends to the whole economy by adding finance into transaction table. This forms an integrated capital flow cycling system for whole economy. Economic growth is used to reflect economic dynamic characteristics. Whole economy is divided into five subsystems. On the basis of subsystem's balance sheet, we set up the simultaneous equation of whole economy. As part of the disposal income, the new loan of the subsystems is affected by money supply badly and the income ratio is not stable. In this paper, expenditure ratio of subsystems is used to solve the simultaneous equation. The subsystem's balance sheets and the simultaneous equation can be applied to study fundamental economic issues effectively.

Keywords: macroeconomy, input-output model, balance sheet, economic mechanism, fundamental economic issues, economic simulation experiment.

JEL Classification: C00,C30.

1. Introduction

In mathematical logic, Leontief built up the legitimate relationship among macroeconomic variables. It is a system of simultaneous equation of economic variables expressed with technological matrix (Leontief 1970) Input-output model (I-O in short) is very well known among economists and policy analysts, and it is widely used to study economic impacts of one sector on an economy in the national and the regional level.

Leontief set up the relationship of environment pollution with consumption and production (Leontief 1986). M. Peneder illustrated the effect of industrial structure on aggregate income and growth (Peneder 2002). Swenson and Moore presented a method to determine full economic impacts of tax law (Swenson, Moore, 1987). Santos identified regional perturbations pursuant to disaster scenarios (Fisher 1916)

The application of technological table is too long to list. They have a common feature that every application is specialized in a particular aspect due to short of finance system. With finance in transaction table, it forms an integrated capital flow cycling system for the whole economy. Together

with production, consumption and investment, taking finance into consideration is essential to build technological matrix, with which we can depict macroeconomy phenomena and explain its mechanism reasonably as a whole.

This paper attempts to fill the gap. On the closed economy, we divided whole economy into five subsystems: household, government, enterprise, investment and finance. As a result, the technological matrix is simplified to each subsystem's receipts and expenditure table. For every sector on the technological table, input equals output and for every subsystem's on transaction table, disposal income equals payment. Both of them are in accordance with Walras general equilibrium theory (Walras 1957)

To illustrate economic dynamic characteristic, economic growth is introduced into transaction table. Then we get simultaneous equation of whole economy on the basis of transaction table and balance sheets. To solve the equation, we use output (expenditure) ratio instead of input (income) ratio. As part of disposal income, new loan amount is affected by money supply very badly, and as part of expenditure, the amount of loan to be repaid is decided by total loan of previous period. This suggests that expenditure ratio is more stable than income ratio. With the help of expenditure ratio, independent variables replace the dependent ones. After plugging replaced variables into these equations, they become solvable. They can be used to illustrate economic mechanism, to perform economic simulating experiments, and to explain economic phenomena and fundamental issues such as money supply, calculation of inflation rate and economy growth, reasonable interest rate and exchange rate, and the cause of stagnation.

2. Transaction table

To analyse macroeconomic situation as a whole, we divide closed economy system into 5 subsystems: household, government, enterprise, investment and finance. By tracing capital flows among subsystems, we get closed macroeconomy's technological table similar to Leontief's I-O table. Because subsystem's receipt is its input and payment is the output, the macroeconomy's technological table is also transaction table. We use x_{ij} to represent the capital flows among subsystems. x_{ij} indicates subsystem *i* paid xamount of money to subsystem *j*. The row sum $\sum_{j=1}^{n} x_{ij}$ represents subsystem's total payments and column sum $\sum_{i=1}^{n} x_{ij}$ total receipts.

	Н	G	E	l I	F
Н	\	<i>x</i> ₁₂	<i>x</i> ₁₃	<i>x</i> ₁₄	<i>x</i> ₁₅
G	<i>x</i> ₂₁	١	<i>x</i> ₂₃	<i>x</i> ₂₄	<i>x</i> ₂₅
E	<i>x</i> ₃₁	<i>x</i> ₃₂	١	<i>x</i> ₃₄	<i>x</i> ₃₅
I	<i>x</i> ₄₁	<i>x</i> ₄₂	<i>x</i> ₄₃	\	<i>x</i> ₄₅
F	<i>x</i> ₅₁	<i>x</i> ₅₂	<i>x</i> ₅₃	<i>x</i> ₅₄	N

Table 1. Transaction table

Note: *H: Household; G: Government; E: Enterprise; I: Investment; F: Finance.

For Leontief's I-O table, it is evident that every sector's total input equals total output and for the transaction table above, receipts equal payments. Both of them are in accordance with the general equilibrium presented by Walras. The equilibrium of subsystems capital flows are exactly the balance sheets used in accounting. For the purpose of easy to understand and convenient to discuss in depth, the components of income and payment of every subsystem will be listed and their relationship will be studied.

3. Balance sheets

The balance sheet discussed here is in a broad sense. The income is all the money a subsystem can get (including new loan), and payment consists of all the money a subsystem can use. To complete balance sheets, the variables and their meanings are given in Table2.

VARIABLE	MEANING
A _i	Asset bought from government for investment (mainly land).
AR	Available money to be loaned and deposit reserve.
В	Bond - B_e , B_{en} , B_g and B_{gn} are the bonds becoming due and newly issued by enterprise and government respectively.
С	Consumption: C_g - Government consumption; C_h – Household consumption; C_i - Investment consumption; C_{ig} – Capital Construction consumption; C_{ih} – Housed investing consumption
C_r	Unit material cost rate:
D	Deposit: D_e , D_g , D_h and D_i – Enterprise, government, household and investment deposits respectively
D _i	Depositinterest: D_{ie} , D_{ig} , D_{ih} and D_{ii} – Interest income from enterprise, government household and investment deposits respectively
Ι	Investment: I_{gi} is government investment to state owned companies.
L	Loan: L_e, L_g, L_h, L_i - Enterprise, government, household and investment loans respectively
L _a	The amount of the <i>loan to be repaid:</i> L_{ae} , L_{ag} , L_{ah} and L_{ai} – the loan amount needed to be paid to finance by enterprise, government, household and investment respectively
L _c	Laborcost: L_{ce}, L_{cg}, L_{ci} – Enterprise, government capital construction, and investment labor costs
L _i	Loaninterest: L_{ie}, L_{ig}, L_{ih} and L_{ii} - the loan interest paid to finance by enterprise, government, household and investment respectively
М	Money. Newly issued money.
Р	Price
P _r	Profit: P_{rg} , P_{rh} – profits paid to government and household by enterprise. P_{ri} – profit used for new investment by enterprise.
Q	Quantity. The amount of goods and service.
S	Stock.
SL	Salary. Government employee salary.
Т	Tax.

Table 2. List of variables and their meanings.

Take householding as an example to demonstrate the establishment of subsystem's balance sheets. Household incomes and payments are listed in Table 3.

INCOME	COMES FROM	PAYMENT	GOES TO
Labour cost <i>L_{ce}</i>	E	Consumption C _h	E
Matured Bond B_e	E	Investment consumption C _{ih}	E
Profit P _{rh}	E		
Salary <i>SL</i>	G	New bond B_{gn}	G
Matured bond B_g	G		
Labour cost L_{cg}	G		
Labour cost <i>L_{ci}</i>	I	New bond B_{en} , new stock S	I
New loan L _h	F	Due Ioan L _{ah}	F
Denosit interest D.	F	Loan interestL _{ih}	F
Deposit interest D_i		Deposit D _h	F

Table 3. Household incomes and payments.

Note: *Pensions are included in salary.

Writing in symbols, household balance sheet becomes:

$$L_{ce} + B_e + P_{rh} + S_L + B_g + L_{cg} + L_{ci} + L_h + D_{ih}$$

= $C_h + C_i h + S + B_{en} + B_{an} + L_{ah} + D_h + L_{ih}$

Inspecting the balance sheet, we find that the loan is still part of disposal income. Actually in the whole inspecting period, the loan was gradually used and used up at the end of period. The new loan added to entire economy caused the price goes up and quantity rises. In turn, comparing with last period, it enhanced every sub-system's actual income. In short, every sub-system's loan melts down in the entire economy by the work of economical mechanism and increased sub-systems' actual incomes by feedback. The problem appeared above is caused by analyzing dynamic economic issues with static analysis method. Therefore the balance sheet has to be reformatted to approach actual economic condition.

For simplicity, we assume constant increment to scale. That is if nominal growth increase by 5 percent, all the transaction variables in inspecting period will also increase by 5 percent, except for the variables related to finance x_{5j} and x_{i5} (i, j = 1, ..., 5). The reason that finance related variables do not move together with growth is that new loan is decided by new issued money and repaid loan is affected by total loan.

Take actual income from last period plus new loan and interest of deposit from this period as this period's disposal income, and take the sum of actual payment of last period and its nominal increment from this period, plus payback to finance from this period as this period's payments, the reformatted household balance sheet is shown below:

$$\begin{aligned} L_{ce}^{o} + B_{e}^{o} + P_{rh}^{o} + S_{L}^{o} + B_{g}^{o} + L_{cg}^{o} + L_{ci}^{o} + L_{h}^{'} + D_{ih}^{'} \\ &= (1 + Q') \Big(C_{h}^{o} + C_{ih}^{o} + S^{o} + B_{en}^{o} + B_{gn}^{o} \Big) + L_{ah}^{'} + D_{h}^{'} + L_{ih}^{'} \end{aligned}$$

where superscripts .^o indicates Last period (year), and .' indicates the inspecting period (this year). Likewise Government balance sheet:

$$T^{o} + P_{rg}^{o} + B_{gn}^{o} + A_{i}^{o} + L'_{g} + D'_{ig} = (1 + Q') (C_{g}^{o} + C_{ig}^{o} + I_{gi}^{o} + SL^{o} + B_{g}^{o} + L_{cg}^{o}) + L'_{ag} + D'_{g} + L'_{ig}$$

andEnterprise balance sheet:

$P^{o}Q^{o} + L'_{e} + D'_{ie} = (1+Q)(C^{o}_{r}Q^{o} + L^{o}_{ce} + B^{o}_{e} + P^{o}_{rh} + P^{o}_{ri} + P^{o}_{rg} + T^{o}) + L'_{ae} + D'_{e} + L'_{ie}$ Note that total sales *PQ*, intermediate goods *C_rQ* and final goods satisfy:

$$PQ - C_r Q = C_h + C_{ih} + C_g + C_{ig} + C_i$$

If the ratio of intermediate goods to total sales λ is a constant. The intermediate goods could be replaced by λ and the final goods:

$$PQ = \frac{1}{1-\lambda} (C_h + C_{ih} + C_g + C_{ig} + C_i), \quad PQ = \frac{C_r Q}{\lambda}$$

This gives:

. .

$$C_r Q = \frac{\lambda}{1 - \lambda} (C_h + C_{ih} + C_g + C_{ig} + C_i), \quad Q' C_r Q = \frac{\lambda Q'}{1 - \lambda} (C_h + C_{ih} + C_{ig} + C_i)$$

Hence the transformed enterprise balance sheet will be:

$$\left(1 - \frac{\lambda Q'}{1 - \lambda}\right) \left(C_h^o + C_{ih}^o + C_g^o + C_{ig}^o + C_i^o\right) + L'_e + D'_{ie} = (1 + Q') \left(L_{ce}^o + B_e^o + P_{rh}^o + P_{ri}^o + P_{rgo} + T_{o} + Lae' + De' + Lie'\right)$$

Investment balance sheet and the Finance balance sheet will respectively be:

$$\begin{split} P_{ri}^{o} + S^{o} + B_{en}^{o} + I_{gi}^{o} + L_{i}' + D_{ii}' &= (1 + Q')(C_{i}^{o} + A_{i}^{o} + L_{ci}^{o}) + L_{ai}' + D_{i}' + L_{ii}';\\ M' + L_{ah}' + D_{h}' + L_{ih}' + L_{ag}' + D_{g}' + L_{ig}' + L_{ae}' + D_{e}' + L_{ie}' + L_{ai}' + D_{i}' + L_{ii}'\\ &= \Delta A R' + L_{h}' + D_{ih}' + L_{g}' + D_{ig}' + L_{e}' + D_{ie}' + L_{i}' + D_{ii}'. \end{split}$$

4. Macro economy mechanism

Summing up all the money one subsystem can get from others forms capital flow variables x_{ij} used in transaction table. See Table 4.

<i>x</i> ₂₁	<i>x</i> ₁₂	<i>x</i> ₃₁	<i>x</i> ₁₃	<i>x</i> ₄₁
$S_L + B_g + L_{cg}$	B_{gn}	$L_{ce} + P_{rh} + B_e$	$C_h + C_{ih}$	L _{ci}
<i>x</i> ₁₄	<i>x</i> ₂₄	<i>x</i> ₄₂	<i>x</i> ₂₃	<i>x</i> ₃₂
$B_{en} + S$	I_{gi}	A_i	$C_g + C_{ig}$	$P_{rg} + T$
<i>x</i> ₃₄	<i>x</i> ₄₃	<i>x</i> ₅₁	<i>x</i> ₁₅	<i>x</i> ₅₂
P _{ri}	C _i	$L_h + D_{ih}$	$L_{ah} + L_{ih} + D_h$	$L_g + D_{ig}$
<i>x</i> ₂₅	<i>x</i> ₅₄	<i>x</i> ₄₅	<i>x</i> ₅₃	<i>x</i> ₃₅
$L_{ag} + L_{ig} + D_g$	$L_i + D_{ii}$	$L_{ai} + L_{ii} + D_i$	$L_e + D_{ie}$	$L_{ae} + L_{ie} + D_e$
$M_1 = M - \Delta AR$				

On the basis of sub-system's balance sheets and summed up capital flow variables, the working mechanism chart of entire-dynamic macro economy can be drawn as Figure 1.

Figure 1 depicts the working mechanism of entire dynamic macro economy and how it is running orderly and effectively. On the market, profits drive economic units, resource provides basic needs and monetary policy ensures the market moving smoothly. The capital flows formed in market transaction makes all the economic activities a whole macro economy. The changes of amount and direction of capital flows determine the state of macro economy.



Figure 1.Illustration of the entire dynamic macro economy. NR: Natural Resource; M: Money Supply; AR:Available money to be loaned and deposit reserve.

5. Dynamic transaction table

Among sub-systems of macro economy, there are capital inter-flows. It is the capital inter-flows of sub-systems, which connect one sub-system to another inseparably, forming a whole organic macro economy. In the following discussion, the simultaneous equations will be set-up according to the quantities of capital flow among sub-systems, which contains the logical relationship of all the macro-economic variables.

Putting summed up variable x_{ij} into balance sheets; we get simultaneous equations composed of 5 subsystem balance sheets:

$$\begin{aligned} x_{21}^{o} + x_{31}^{o} + x_{41}^{o} + x_{51}^{'} &= (1+Q')(x_{12}^{o} + x_{13}^{o} + x_{14}^{o}) + x_{15}^{'}; \\ x_{12}^{o} + x_{42}^{o} + x_{32}^{o} + x_{52}^{'} &= (1+Q')(x_{21}^{o} + x_{24}^{o} + x_{23}^{o}) + x_{25}^{'}; \\ (1 - \frac{\lambda Q'}{1-\lambda})(x_{13}^{o} + x_{23}^{o} + x_{43}^{o}) + x_{53}^{'} &= (1+Q')(x_{31}^{o} + x_{32}^{o} + x_{34}^{o}) + x_{35}^{'}; \\ x_{14}^{o} + x_{24}^{o} + x_{34}^{o} + x_{54}^{'} &= (1+Q')(x_{41}^{o} + x_{42}^{o} + x_{43}^{o}) + x_{45}^{'}; \\ x_{15}^{'} + x_{25}^{'} + x_{45}^{'} + x_{35}^{'} + M_{1}^{'} &= x_{51}^{'} + x_{52}^{'} + x_{54}^{'} + x_{53}^{'}. \end{aligned}$$

Table 1 is a static and settled accounts transaction matrix. We can establish a dynamic one according to the equations above. See Table 5.

	Н	G	E		F
		$(1 + \theta') x_{12}^{o}$	$(1 + \theta') x_{13}^o$	$(1 + \theta') x_{14}^o$	
Η		<i>x</i> ^{<i>o</i>} ₁₂	$\left(1-rac{\lambda heta '}{1-\lambda} ight) x_{13}^{o}$	<i>x</i> ^o ₁₄	<i>x</i> [′] ₁₅
	$(1 + \theta') x_{21}^o$		$(1 + \theta') x_{23}^o$	$(1 + \theta') x_{24}^o$	
G	<i>x</i> ^o ₂₁		$\left(1-\frac{\lambda\theta'}{1-\lambda}\right)x_{23}^o$	x ^o ₂₄	<i>x</i> ′ ₂₅
Е	$(1 + \theta') x_{31}^o$	$(1 + \theta') x_{32}^o$		$(1 + \theta') x_{34}^o$	x'_{35}

 Table 5. Dynamic transaction table.

	x ^o ₃₁	x ^o ₃₂		x_{34}^{o}	
	$(1 + \theta') x_{41}^o$	$(1 + \theta') x_{42}^o$	$(1 + \theta') x_{43}^o$		
I	<i>x</i> ^{<i>o</i>} ₄₁	x ⁰ ₄₂	$\left(1-rac{\lambda heta '}{1-\lambda} ight) x_{43}^o$		<i>x</i> [′] ₄₅
F	x' ₅₁	x' ₅₂	x' ₅₃	x' ₅₄	M'_1

The variables on the top of each cell are used to calculate payments and those on the bottom are used to calculate receipts.

6. Simultaneous equation

Even taking M'_1 as known, the simultaneous equation has 20 unknowns and only 5 linear equations. To solve it, we need the second assumption. Similar to single process production function assumption used in I-O model (Mouhammed 2000), we assume that there is a fixed expenditure pattern for every subsystem. The constant payment to scale is that subsystems single payment will increase with its total income by the same percent. The payment proportion to total income of subsystems is subject to economic structure. They are relatively stable, so we take it as constant. For example, during the 11 years from 2001 to 2012 in United Kingdom, the living cost and food expenditure remains steady, 489 pound per week (inflation adjusted). It is almost a level line (Office for National Statistics, 2013). The statistical analysis of Americans household expenditure proportion got similar conclusion (The Wall Street Journal, 2012).

Emphasizing the stability of expenditure ratio does not deny its little change. The payment ratio is only an instrument used to solve the equation here. In the process of applying them, we need to adjust them according to actual statistic results. The payment ratio can be used to reform the simultaneous equations. Payment ratio of sub-systems is in Table 6.

Household	Patio	<i>x</i> ₁₂	<i>x</i> ₁₃	<i>x</i> ₁₄	<i>x</i> ₁₅	
	Natio	α_1	α_2	α3	$lpha_4$	$\sum \alpha_i = 1$
Covernment	Variable	<i>x</i> ₂₁	<i>x</i> ₂₄	<i>x</i> ₂₃	<i>x</i> ₂₅	
Government	Ratio	β_1	β_2	β_3	β_4	$\sum \beta_i = 1$
E. t	Variable	<i>x</i> ₃₁	<i>x</i> ₃₂	<i>x</i> ₃₄	<i>x</i> ₃₅	
Enterprise	Variable	γ_1	γ_2	γ_3	γ_4	$\sum \gamma_i = 1$
Invootmont	Variable	<i>x</i> ₄₁	<i>x</i> ₄₂	<i>x</i> ₄₃	<i>x</i> ₄₅	
investment	Ratio	ζ_1	ζ_2	ζ_3	ζ_4	$\sum \zeta_i = 1$
Finance	Variable	<i>x</i> ₅₁	<i>x</i> ₅₂	<i>x</i> ₅₄	<i>x</i> ₅₃	
	Ratio	ϵ_1	ϵ_2	ϵ_3	ϵ_4	$\sum \epsilon_i = 1$

 Table 6 .Payment ratio of subsystems.

Note: The calculation of payment ratio is shown in Table 7.

Table 7. Payment ratio calculation.

α1	$\frac{(1+Q')x_{12}^o}{x_{21}^o+x_{31}^o+x_{41}^o+x_{51}'}$	β_1	$\frac{(1+Q')x_{21}^o}{x_{12}^o+x_{42}^o+x_{32}^o+x_{52}'}$
α2	$\frac{(1+Q')x_{13}^o}{x_{21}^o+x_{31}^o+x_{41}^o+x_{51}'}$	β_2	$\frac{(1+Q')x_{24}^o}{x_{12}^o+x_{42}^o+x_{32}^o+x_{52}'}$
α3	$\frac{(1+Q')x_{14}^o}{x_{21}^o+x_{31}^o+x_{41}^o+x_{51}'}$	eta_3	$\frac{(1+Q')x_{23}^o}{x_{12}^o+x_{42}^o+x_{32}^o+x_{52}'}$
α4	$\frac{x_{15}'}{x_{21}^o + x_{31}^o + x_{41}^o + x_{51}'}$	eta_4	$\frac{x_{25}'}{x_{12}^o + x_{42}^o + x_{32}^o + x_{52}'}$

ζ_1	$\frac{(1+Q')x_{41}^o}{x_{14}^o+x_{24}^o+x_{34}^o+x_{54}'}$	γ_1	$\frac{(1+Q')x_{31}^o}{\left(1-\frac{\lambda Q'}{1-\lambda}\right)(x_{13}^o+x_{23}^o+x_{43}^o)+x_{53}'}$
ζ_2	$\frac{(1+Q')x_{42}^o}{x_{14}^o+x_{24}^o+x_{34}^o+x_{54}^\prime}$	γ_2	$\frac{(1+Q')x_{32}^o}{\left(1-\frac{\lambda Q'}{1-\lambda}\right)(x_{13}^o+x_{23}^o+x_{43}^o)+x_{53}'}$
ζ_3	$\frac{(1+Q')x_{43}^o}{x_{14}^o+x_{24}^o+x_{34}^o+x_{54}'}$	γ_3	$\frac{(1+Q')x_{34}^o}{\left(1-\frac{\lambda Q'}{1-\lambda}\right)(x_{13}^o+x_{23}^o+x_{43}^o)+x_{53}'}$
ζ_4	$\frac{x_{45}'}{x_{14}^o + x_{24}^o + x_{34}^o + x_{54}'}$	γ_3	$\frac{(1+Q')x_{35}^o}{\left(1-\frac{\lambda Q'}{1-\lambda}\right)(x_{13}^o+x_{23}^o+x_{43}^o)+x_{53}'}$

ϵ_1	$\frac{x_{51}'}{x_{15}'+x_{25}'+x_{45}'+x_{35}'+M_1'}$
ϵ_2	$\frac{x_{52}'}{x_{15}' + x_{25}' + x_{45}' + x_{35}' + M_1'}$
ϵ_3	$\frac{x_{54}'}{x_{15}' + x_{25}' + x_{45}' + x_{35}' + M_1'}$
ϵ_4	$\frac{x_{53}'}{x_{15}' + x_{25}' + x_{45}' + x_{35}' + M_1'}$

On the basis of the second assumption, every subsystem's expenditure variables are all interrelated and dependent variables. There is only one independent expenditure variable in each subsystem. They can be replaced one by the other. Therefore, the simultaneous equation has 5 independent variables with 5 linear equations. It can be solved uniquely. Choosing one payment variable such as x_{21}^o and 4 income variables x_{12}^o , x_{42}^o , x_{32}^o , x'_{52} in one subsystem's balance sheet as basic variables, we build new simultaneous equations, shown below.

$$\begin{aligned} x_{21}^{o} &= \frac{\beta_{1}}{1+Q'} (x_{12}^{o} + x_{42}^{o} + x_{32}^{o} + x_{52}^{o}); \\ x_{12}^{o} &= \frac{\alpha_{1}}{1+Q'} (x_{21}^{o} + x_{31}^{o} + x_{41}^{o} + x_{51}^{'}); \\ x_{42}^{o} &= \frac{\zeta_{2}}{1+Q'} (x_{14}^{o} + x_{24}^{o} + x_{34}^{o} + x_{54}^{'}); \\ x_{32}^{o} &= \frac{\gamma_{2}}{1+Q'} \Big[\Big(1 - \frac{\lambda Q'}{1-\lambda} \Big) (x_{13}^{o} + x_{23}^{o} + x_{43}^{o}) + x_{53}^{'} \Big]; \\ x_{52}^{'} &= \epsilon_{2} (x_{15}^{'} + x_{25}^{'} + x_{45}^{'} + x_{35}^{'} + M_{1}^{'}). \end{aligned}$$

To get solvable equations, we need to replace all other variables by basic ones with the aid of variable replacement table. See Table 8.

	Variable	<i>x</i> ⁰ ₂₁	<i>x</i> ^{<i>o</i>} ₂₄	<i>x</i> ^o ₂₃	x'25
Government	Replacement	$\frac{\beta_1}{\beta_1} x_{21}^o$	$\frac{\beta_2}{\beta_1} x_{21}^o$	$\frac{\beta_3}{\beta_1} x_{21}^o$	$\frac{\beta_4(1+Q^{'})}{\beta_1}x_{21}^o$
Household	Variable	x_{12}^{o}	x_{13}^{o}	x_{14}^{o}	$x_{15}^{'}$
	Replacement	$\frac{\alpha_1}{\alpha_1} x_{12}^o$	$\frac{\alpha_2}{\alpha_1} x_{12}^o$	$\frac{\alpha_3}{\alpha_1} x_{12}^o$	$\frac{\alpha_4(1+Q^{'})}{\alpha_1}x_{12}^o$

	Variable	x_{42}^{o}	x_{41}^{o}	x^{o}_{43}	x' ₄₅
Investment	Replacement	$\frac{\zeta_2}{\zeta_2} x_{42}^o$	$\frac{\zeta_1}{\zeta_2} x_{42}^o$	$\frac{\zeta_3}{\zeta_2} x_{42}^o$	$\frac{\zeta_4(1+Q^{'})}{\zeta_1} x_{42}^o$
	Variable	<i>x</i> ^{<i>o</i>} ₃₂	x_{31}^{o}	x_{34}^{o}	x'_{35}
Enterprise	Replacement	$\frac{\gamma_2}{\gamma_2} x_{32}^o$	$\frac{\gamma_1}{\gamma_2} x_{32}^o$	$rac{\gamma_3}{\gamma_2} x^o_{32}$	$\frac{\gamma_4(1+Q')}{\gamma_2} x_{32}^o$
	Variable	$x_{52}^{'}$	$x_{51}^{'}$	x'_{54}	x'_{53}
Finance	Replacement	$\frac{\epsilon_2}{\epsilon_2} x'_{52}$	$\frac{\epsilon_1}{\epsilon_2} x'_{52}$	$\frac{\epsilon_3}{\epsilon_2} x_{52}'$	$\frac{\epsilon_4}{\epsilon_2} x'_{52}$

Putting replaced variables into new simultaneous equations, we have re-organized 5 elements linear function simultaneous equations. See below.

$$\begin{aligned} x_{21}^{o} &= \frac{\beta_{1}}{1+Q'} (x_{12}^{o} + x_{42}^{o} + x_{32}^{o} + x_{52}^{o}); \\ x_{12}^{o} &= \frac{\alpha_{1}}{1+Q'} \left(x_{21}^{o} + \frac{\zeta_{1}}{\zeta_{2}} x_{42}^{o} + \frac{\gamma_{1}}{\gamma_{2}} x_{32}^{o} + \frac{\epsilon_{1}}{\epsilon_{2}} x_{52}^{\prime} \right); \\ x_{42}^{o} &= \frac{\zeta_{1}}{1+Q'} \left(\frac{\beta_{2}}{\beta_{1}} x_{21}^{o} + \frac{\alpha_{3}}{\alpha_{1}} x_{12}^{o} + \frac{\gamma_{3}}{\gamma_{2}} x_{32}^{o} + \frac{\epsilon_{3}}{\epsilon_{2}} x_{52}^{\prime} \right); \\ x_{32}^{o} &= \frac{\gamma_{2}}{1+Q'} \left[\left(1 - \frac{\lambda Q'}{1-\lambda} \right) \left(\frac{\beta_{3}}{\beta_{1}} x_{21}^{o} + \frac{\alpha_{2}}{\alpha_{1}} x_{12}^{0} + \frac{\zeta_{3}}{\zeta_{2}} x_{42}^{o} \right) + \frac{\epsilon_{4}}{\epsilon_{2}} x_{52}^{\prime} \right]; \\ x_{52}^{\prime} &= \epsilon_{2} (1+Q') \left(\frac{\beta_{4}}{\beta_{1}} x_{21}^{o} + \frac{\alpha_{4}}{\alpha_{1}} x_{12}^{o} + \frac{\zeta_{4}}{\zeta_{2}} x_{42}^{o} + \frac{\gamma_{4}}{\gamma_{2}} x_{32}^{o} \right) + \epsilon_{2} M_{1}^{\prime}. \end{aligned}$$

Taking economic structure ratio $\alpha_i, \beta_i, \zeta_i, \epsilon_i$ and nominal growth rate Q' as known. x_{ij} can be solved, which are expressed by M'_1 . The determinant solution of the simultaneous equation is given below:

$$\begin{pmatrix} 1 & -\frac{\beta_1}{1+Q'} & -\frac{\beta_1}{1+Q'} & -\frac{\beta_1}{1+Q'} & -\frac{\beta_1}{1+Q'} & -\frac{\beta_1}{1+Q'} \\ -\frac{\alpha_1}{1+Q'} & 1 & -\frac{\zeta_1\alpha_1}{\zeta_2(1+Q')} & -\frac{\gamma_1\alpha_1}{\gamma_2(1+Q')} & -\frac{\epsilon_1\alpha_1}{\epsilon_2(1+Q')} \\ -\frac{\beta_2\zeta_2}{\beta_1(1+Q')} & -\frac{\alpha_3\zeta_2}{\alpha_1(1+Q')} & 1 & -\frac{\gamma_3\zeta_2}{\gamma_2(1+Q')} & -\frac{\epsilon_3\zeta_2}{\epsilon_2(1+Q')} \\ -\frac{\beta_3\gamma_2\left(1-\frac{\lambda Q'}{1-\lambda}\right)}{\beta_1(1+Q')} & -\frac{\alpha_2\gamma_2\left(1-\frac{\lambda Q'}{1-\lambda}\right)}{\alpha_1(1+Q')} & -\frac{\zeta_3\gamma_2\left(1-\frac{\lambda Q'}{1-\lambda}\right)}{\zeta_2(1+Q')} & 1 & -\frac{\epsilon_4\gamma_2}{\epsilon_2(1+Q')} \\ -\frac{\beta_4\epsilon_2(1+Q')}{\beta_1} & -\frac{\alpha_4\epsilon_2(1+Q')}{\alpha_1} & -\frac{\zeta_4\epsilon_2(1+Q')}{\zeta_2} & -\frac{\gamma_4\epsilon_2(1+Q')}{\gamma_2} & 1 \end{pmatrix} = \begin{pmatrix} 0 \\ 0 \\ 0 \\ \epsilon_2M'_1 \end{pmatrix}$$

7. The usage of simultaneous equation

To demonstrate the usage of simultaneous equation, a hypothetical payment ratio table is given in Table 9.

Table 9. Hypothetical payment ratio.

Series	1	2	3	4
Ratio				
α	.05	.4	.1	.45
β	.15	.1	.55	.2

γ	.35	.3	.2	.15
ζ	.2	.1	.55	.15
ε	.12	.3	.4	.18

If the hypothetical nominal growth θ' and intermediate goods ratio λ are .25 and .4 respectively. The calculated determinant solution of the simultaneous equation is:

	/ 1	12	12	12	12 \		$\begin{pmatrix} 0 \end{pmatrix}$
1	04	1	08	0466	016		0
	0533	16	1	0533	1067	=	0
	7333	-1.6	-1.1	1	144		0
١	\5	-3.375	5625	1875	1 /		$\langle .3M_1' \rangle$

After solving the equation, we get x_{ii}^{o} and x_{ii}' in Table 10.

x_{12}^{o}		<i>x</i> ^o ₁₃	x^{o}_{14}	x' ₁₅
	.04077	.32616	.08154	.45866
x_{21}^{0}		x ^o ₂₃	x_{24}^{o}	x'_{25}
.13687		.50816	.09125	.22812
<i>x</i> ^o ₃₁	x_{32}^{o}		x_{34}^{o}	x'_{35}
.42864	.36741		.24494	.22963
<i>x</i> ^o ₄₁	x_{42}^{o}	x_{43}^{o}		x'_{45}
.20158	.10079	.55435		.18898
x'_{51}	$x_{52}^{'}$	$x_{53}^{'}$	$x_{54}^{'}$	
.25264	.63160	.37896	.84213	

Table 10.*x*_{*ij*} solutions.

Note: We omit M'_1 factor in all entries.

Now the dynamic transaction table becomes:

Table 11. Hypothetical dynamic transaction table.

	Н	G	E	l I	F
ш		.05096	.40770	.10193	15866
п		.04077	.27180	.08154	.40000
G	.17109		.62733	.11406	00010
G	.13687		.41821	.09125	.22012
F	.53580	.45926		.30618	22062
Ē	.42864	.36741		.24494	.22903
I	.25198	.12600	.69294		10000
	.20158	.10079	.46196		.10090
F	.25264	62160	.37896	.84213	1
		.03100			

8. Applications

The result discussed above can be used widely in macroeconomy. We give a few examples demonstrating applications of balance sheets and the simultaneous equation.

8.1. Applications of Balance Sheets

Monetary equation

The money supply can be captured by summing up sub-system's 5 balance sheets. After added up and reorganized, we have the money supply:

$$M' = \Delta AR$$

Available loan and reserve increment:

$$+\Delta C_h^o + \Delta C_{ih}^o + \Delta C_q^o + \Delta C_{iq}^o + \Delta C_i^o + \Delta C_r^o \cdot Q^o$$

Total sales increment $\Delta(PQ)$:

 $+\Delta L_{ce}^{o} + \Delta L_{ci}^{o} + \Delta L_{cg}^{o} + \Delta SL^{o}$

Labor cost increment ΔL_c

 $+\Delta P_{rh}^{o} + \Delta P_{rg}^{o} + \Delta P_{ri}^{o}$

Profit increment ΔP_r

 $+\Delta B_{en}^{o} + \Delta B_{an}^{o} + \Delta B_{e}^{o} + \Delta B_{a}^{o} + \Delta S^{o}$ security market volume increment ΔB

 $+\Delta T^{o}$ Tax increment

 $+\Delta A_i^o$ Asset increment

 $+\Delta I_{an}^{o}$ State owned co investment increment by government

Let $Y = LC + P_r + B + T + A_i + I_{gi}$ denote the yield. The money supply is equal to total sales increment and total yield increment plus available loan and reserve increment. Equation can be written in short:

$$M' = \Delta(PQ)^o + \Delta Y^o + \Delta AR'$$
 or $M'_1 = \Delta(PQ)^o + \Delta Y^o$

Compared with monetary exchange equation MV = PQ by Irving Fisher in 1911(Office for National Statistics, 2013) it is evident that its foundation is not enough. Except for total sales increment and total yield incrementof the year, last year's total sales and total yield is supported by existing money in the system. Furthermore, the velocity of money has no direct connection with money supply, because money owner can only use their money once.

Inflation rate and economy growth calculation

By changing monetary equation into another form, the relationship among money supply, all sales, inflation rate and economic growth will surface. Dividing both sides of monetary equation with total sales $P^{o}Q^{o}$ and reorganizing it we get inflation rate and the economy growth formula:

$$\frac{\Delta P}{P} + \frac{\Delta Q}{Q} = \frac{M' - \Delta A R' - \Delta Y^o}{P^o Q^o} \qquad \text{or} \qquad \frac{\Delta P}{P} + \frac{\Delta Q}{Q} = \frac{M'_1 - \Delta Y^o}{P^o Q^o}$$

The formula above tells that inflation rate plus growth equals new issued money minus available loan and reserve increment and yield increment divided by total sales of last year.

Actually the article *Relationship between the Money Supply and Inflation*(F.M.I. Strategy, 2012) already got close conclusion with the formula above: The perception relationship between money supply and inflation should take economic growth into consideration.

Interest rate calculation in closed economy

By summing up the balance sheets of four sub-systems (H, G, E, I), the relationship among economy growth, total loan and deposit of the term, accumulative total loan and deposit of last term, difference value between loan and deposit interest rate, time limit of loan and loan interest rate is

uncovered. The reasonable loan interest rate in a closed enonomy setting can be calculated with the formula below:

$$i_{l} = \frac{TL' - TD'}{ATL^{o} - ATD^{o}} - \frac{TL^{o}/K + wTD^{o}}{ATL^{o} - ATD^{o}} - \frac{Q'((PQ)^{o} + Y^{o})}{ATL^{o} - ATD^{o}}$$

Here TLand TD are the total loan and deposit of this term respectively, ATL and ATD are accumulate total loan and deposit of last term resp., K is the time limit of the loan and wrepresents the difference value of loan and deposit interest rate.

In Taylor's monetary and policy rule, interest rate responses to changes in inflation and output (Taylor, 1993). The formula above provides a more close approach by taking new issued money into consideration.

Exchange rate calculation in open economy

Adding merchandise trading amount of export and import and capital flows in and out of the country to subsystem's balance sheet of closed economy, we get subsystems' balance sheets of open economy. By summing up five subsystems' balance sheets in an open economy setting, the relationship among supply, economy growth, export of last term, import of this term, surplus of capital flow and exchange rate will appear. We can calculate the exchange rate from the formula below.

Re =
$$\frac{M'_1 - Q'(P^o Q^o + Y^o)}{(1 + Q')IM^o - EX^o - S'_e}$$

where: Re means exchange rate, IM represents import, EX denotes export and S_e means capital flow surplus. OANDA only listed top 5 factors affecting exchange rate: interest rate, employment outlook, economic growth expectation, trade balance and central bank action (OANDA). The formula also expounded the mathematical relationship among these factors.

8.2. Applications of simultaneous equation

The impacts of one variable to others

As the I-O model, simultaneous equation can be used to study the economic impacts of one variable change on others (Mouhammed, 2002). Actually, if we take M'_1 as a variable, the equation is a homogeneous system. Its solutions can be expressed with any one of six variables. So we can study the economic impacts on the other five variables caused by one variable change.

Economic simulating experiment

The simultaneous equation can be used to do economic simulating experiment. Input different set of macroeconomic control instruments like M'_1 , β_i and t(tax rate) with controlled level of inflation $\frac{\Delta p}{p}$ and expected target of growth $\frac{\Delta Q}{Q}$ together into simultaneous equation, we can get different set of x_{ij} . According to the desired economic state (x_{ij}) , economy reality and feasibility of the instruments, policy maker can opt one set of macroeconomic control instrument (M'_1, β_i, t) and put them into practice.

Reasons for stagnation

After the stagnation happened, the transaction variables x_{ij}^o in the period of economy crisis and x_{ij}^\prime in the period of stagnation are all known. Rewrite simultaneous equation, the inflation rate $\Delta P/P$, subsystems's actual growth $(\frac{\Delta Q_h}{Q_h}, \frac{\Delta Q_g}{Q_g}, \frac{\Delta Q_e}{Q_e}, \frac{\Delta Q_i}{Q_i})$ can be solved.

$$\begin{split} x_{21}^{o} + x_{31}^{o} + x_{41}^{o} + x_{51}^{\prime} &= \left(1 + \frac{\Delta P}{P} + \frac{\Delta Q_h}{Q_h}\right) (x_{12}^{o} + x_{13}^{o} + x_{14}^{o}) + x_{15}^{\prime} \\ x_{12}^{o} + x_{32}^{o} + x_{42}^{o} + x_{52}^{\prime} &= \left(1 + \frac{\Delta P}{P} + \frac{\Delta Q_g}{Q_g}\right) (x_{21}^{o} + x_{23}^{o} + x_{24}^{o}) + x_{25}^{\prime} \\ \left[1 - \frac{\lambda}{1 - \lambda} \left(\frac{\Delta P}{P} + \frac{\Delta Q_e}{Q_e}\right)\right] (x_{13}^{o} + x_{23}^{o} + x_{43}^{o}) + x_{53}^{\prime} \\ &= \left(1 + \frac{\Delta P}{P} + \frac{\Delta Q_e}{Q_e}\right) (x_{31}^{o} + x_{32}^{o} + x_{34}^{o}) + x_{35}^{\prime} \\ x_{14}^{o} + x_{24}^{o} + x_{34}^{o} + x_{54}^{\prime} &= \left(1 + \frac{\Delta P}{P} + \frac{\Delta Q_i}{Q_i}\right) (x_{41}^{o} + x_{42}^{o} + x_{43}^{o}) + x_{45}^{\prime} \end{split}$$

To show how to use the equation above, we use the hypothetical x_{ij}^o and x_{i5}^\prime , i = 1, ..., 4. Reallocate total loan with $\varepsilon_1(.1), \varepsilon_2(.4), \varepsilon_3(.35), \varepsilon_4(.15)$, and

 $x'_{51} = .21584$ $M'_{1}, x'_{52} = .86336$ $M'_{1}, x'_{53} = .75544$ $M'_{1}, x'_{54} = .32376M'_{1}$

Plug those known values (x_{ij}^o, x_{ij}') into the equation we get:

 $\frac{\Delta P}{P} + \frac{\Delta Q_h}{Q_h} \approx .17, \frac{\Delta P}{P} + \frac{\Delta Q_g}{Q_g} \approx .57, \frac{\Delta P}{P} + \frac{\Delta Q_e}{Q_e} \approx .22, \frac{\Delta P}{P} + \frac{\Delta Q_i}{Q_i} \approx .15.$

By the try and error method, we will obtain:

Inflation rate:
$$\frac{\Delta P}{P} = 28\%$$
Household real consumption growth: $\frac{\Delta Q_h}{Q_h} = -11\%$ Government consumption growth: $\frac{\Delta Q_g}{Q_g} = 29\%$ Enterprise real production growth: $\frac{\Delta Q_e}{Q_e} = -6\%$ Invest real growth: $\frac{\Delta Q_i}{Q_i} = -13\%$

Therefore on the basis of economic crisis, government's stimulating measures are the reason for stagnation. This conclusion provides a theoretical support to some existing deduction (The Christian Science Monitor, 2011).

Conclusion

We summarize below the main results and observations in this paper. Making use of balance sheets, we solve fundamental macro-economic issue.

Monetary equation:

$$M' = \Delta(PQ)^o + \Delta Y^o + \Delta AR'.$$

Inflation rate and economy growth formula:

$$\frac{\Delta P}{P} + \frac{\Delta Q}{Q} = \frac{M' - \Delta AR' - \Delta Y'}{P^{o}Q^{o}}$$

Interest rate formula:

$$i_{l} = \frac{TL' - TD'}{ATL^{o} - ATD^{o}} - \frac{\frac{TL^{o}}{\kappa} + wTD^{o}}{ATL^{o} - ATD^{o}} - \frac{Q'((PQ)^{o} + Y^{o})}{ATL^{o} - ATD^{o}}$$

• Exchange rate formula:

$$\operatorname{Re} = \frac{M'_1 - Q'(P^o Q^o + Y^o)}{(1 + Q')IM^o - EX^o - S'_{\rho}}$$

Applying simultaneous equation, we can do economic simulating experiment on computer, to provide scientific basic for macro economy management. We also see that government's stimulating measures in economic crisis are the reason for stagnation.

References

- [1] Fisher, I. (1916). Elementary Principles of Economics.Macmillan Co of Canada, p. 159.
- [2] Santos J.R. (2012). An input-output framework for assessing disaster impacts on Nashville metropolitan region, *submitted* in Economic Systems Research.
- [3] Leontief, W. (1970). Environmental repercussions and the economic structure:an input-output approach.TheReview of Economics and Statistics, 52(3): 262–271.
- [4] Leontief, W. (1986). Input-output economics. Oxford University Press, p. 20.
- [5] Mouhammed, A.H. (2000).Introductory Mathematical Economics. M.E. Sharpe, p.73.
- [6] Mouhammed, A.H. (2002).Introductory Mathematical Economics. M.E. Sharpe, pp. 85-87.
- [7] Peneder, M. (2002). Industrial structure and aggregate growth.WIFOWorking papers, No.182.
- [8] Swenson, C.W, Moore, M.L. (1987). Use of inputoutput analysis in tax research. Taxation, a Research Annual, 1: 49-83.
- [9] Taylor, J.B. (1993). Discretion versus policy rules in practice. In Carnegie-Rochester Conference series on public policy, 39: 195–214.
- [10] Walras, L.(1954). Elements of Pure Economics or the Theory of Social Wealth. American Economic Association, Richard D. Irwin, inc, page 315.
- *** Exchange rate dynamics & currency factors: Top 5 factors affecting exchange rates. OANDA, http:// fxtrade.oanda.ca/learn/top-5-factors-that-affect-exchange-rates
- *** The Christian Science Monitor.(2011).The real cause of economic stagnation. See http://www.csmonitor.com/Business/The-Daily-Reckoning/2011/0921/The-real-cause-of-economicstagnation
- *** The Wall Street Journal. (2012). How Americans spend their money, http://online.wsj.com/news/articles/SB10000872396390444358804578018823313863636
- *** F.M.I. Strategy (2012). Relationship between the money supply and inflation, http://www. macroinvesting-strategy.com/money-supply-and-inflation/
- *** Office for National Statistics (2013). The headlines: Household expenditure at a glance, http://www.ons.gov.uk/ons/rel/family-spending/family-spending/2013-edition/ sum-headlines.html



DOI: http://dx.doi.org/10.14505/tpref.v6.1(11).04

Networks in context: the concept of network in Manuel Castells' theory of the network society

Ari-Veikko ANTTIROIKO University of Tampere, Finland School of Management <u>ari-veikko.anttiroiko@uta.fi</u>

Suggested Citation:

Ari-Veikko A. (2015). Networks in context: the concept of network in Manuel Castells' theory of the network society. *Theoretical and Practical Research in Economic Field*, (Volume VI, Summer), 1(11): 65-80. DOI: 10.14505/tpref.v6.1(11).04.Available from: <u>http://www.asers.eu/journals/tpref/curent-issue</u>.

Article's History:

Received April, 2015; *Revised* June, 2015; *Accepted* August, 2015. 2015. ASERS Publishing. All rights reserved.

Abstract:

This article discusses the conceptualization of network in Manuel Castells' theory of network society. Networks appeared in Castells' works in the late 1980s, when he became interested in the configuration of the relationships between technology, economy, and society. The culmination of this phase was his opus magnum, The Information Age trilogy, which introduced network as a key concept of his macro theory, even though he remained laconic about the concept itself. This is paradoxical, for Castells became possibly the most prominent figure globally in adopting network terminology in macro sociological theory, but at the same time made hardly any empirical, theoretical or methodological contribution to social network analysis or network theory in general. This implies that 'network' in Castells' social theory is based on institutional network analysis. Moreover, it is not an analytical concept but rather a powerful metaphor that served to capture his idea of the new social morphology of informational capitalism.

Keywords: Manuel Castells, network, network society, informationalism, The Information Age, political economy

JEL Classification: A13, A14, B31, B52, F60, F68, O2, P16

1. Introduction

The Spanish sociologist Manuel Castells created one of the most ambitious macro theories of our time, which endeavors to explain and interpret power, economy, and social life in a world transformed by globalization and informatization. Castells already rose to fame as a radical urban sociologist in the 1970s, but his international reputation is largely due to *The Information Age* trilogy published in three volumes in the latter half of the 1990s. This is a political economy-oriented macro-analysis of the tensional relationship between the instrumental networks of informational economy and historically-rooted identities and the world-wide developments conditioned by it. The concept of network entered his thinking in the late 1980s, culminating in the first volume of the trilogy, *The Rise of the Network Society* (Castells 1996).

In this article I discuss the nature and role of the concept of network in Castells' theory of network society. The research question is: how does Castells operationalize and utilize the concept of network in his analysis of the network society? I first contextualize the discussion by briefly shedding light on the most significant events of Castells' academic career. The discussion then turns to introducing the concept of network and its role in his macro sociological theorization. Lastly, I position Castells in the field of social network analysis and assess his contribution to network theory.

2. Turning points of Castells' academic career

Manuel Castells Oliván was born in 1942 to a conservative family in the small town of Hellín in the autonomous region of Castilla-La Mancha, Spain. He spent most of his childhood in Barcelona, where he also started his university studies in the late 1950s. The life of Spanish people was shadowed at that time by Franco's military regime. The young Castells like many others participated in the opposition movement. When the grasp of the military government tightened, many of Castells' friends were captured and tortured; he himself was able to escape to France. (Castells and Ince 2003, 5-9; Fischer 1999; see also Castells 2009).

Castells settled in France and graduated from the University of Paris (la Sorbonne) in 1964. He was interested at that time in conducting sociological analysis of the working class, an interest shared by the famous French sociologist, Professor Alain Touraine, who eventually became Castells' academic mentor (Castells and Ince 2003, 10-12; see also Stalder 2006). In 1967 Castells took his doctoral degree and was soon nominated as an associate professor in sociology at the University of Paris (Castells and Ince 2003, 12-13). Due to his involvement in the 'May 68' movement he was deported from France. After being granted an amnesty by the French government, he was able to return in 1970, this time at the invitation of Touraine to hold an associate professorship at EHESS in Paris (*L'École des hautesétudes en sciences sociales*). This was the beginning of the gradual consolidation of his position in French and slowly also in international circles of urban sociology (Castells and Ince 2003, 15-16; Stalder 2006).

Academically, Castells' main objective became to combine Marxist theory, urban sociology and social movement research, the focus being on empirical research. The most important outcome of that period was *La question urbaine*, which was published in 1972 and translated into English five years later under the title *The Urban Question* (Castells 1977). The book was a great success both in France and internationally, providing considerable input to Marxist urban sociology (Castells and Ince 2003, 15-16; Castells 1977; 1978; see also Pickvance 1975; Rex 1977; Dunleavy 1977; 1980; Elliott 1980, 152; Whiteley 1980; Lowe 1986; Gurr and King 1987; Katznelson 1992; Walton 1993; Merrifield 2002, 114). Probably the most widely discussed issue around this work was the conceptualization of the city as the arena of collective consumption. The message of Castells' structural analysis was that political contradictions escalate unless they are eased by publicly organized and funded collective consumption that meets people's basic needs (Castells 1977; Lowe 1986, 2; see also Dunleavy 1980; Lojkine 1976; de Moura Costa 2007, 23; Cochrane 2003, 531).

The American universities had impressed Castells since the early years of his career. He even said that his own research orientation was more American than French, even though he clearly combined the empirical analysis typical of the American tradition with the theorization associated with the French tradition and the political activism that reflects his Spanish background (Castells and Ince 2003, 16-17; Institute of International Studies 2001). Castells' reputation as a radical reformer of urban political analysis became widely known among urban researchers. Yet his reliance on Marxism began to decline in the late 1970s and by the end of the decade had vanished (Rantanen 2005, 137). That was the time when Castells' academic career took a new turn as he was invited to take up a professorship of urban sociology at UC Berkeley, USA. He took up his post in 1979, at the age of 37. He started to pursue his intellectual passion, research on social movements, but this time in the socially diverse San Francisco (see for example, Castells and Murphy 1982). The culmination of this period was

The City and the Grassroots, published in 1983, which Castells himself regarded as his best work in urban studies (Castells 1983, 291-301; Castells and Ince 2003, 16-17; Fuchs 2008; Mayer 2006).

Around the year 1983 Castells decided to start investigating the connection between technology, economy, and society and how these were reflected in urban structure (e.g. Castells 1985; 1989; Porteset al. 1989; Mollenkopf and Castells 1991). He later stated that at that time he felt that in Europe the magnitude of the information technology revolution had not yet been understood as it was on the other side of the Atlantic (Institute of International Studies 2001; Stalder 2006; cf. Hydén 2001). The first major outcome of this new turn was The Informational City, a book that opened up a new contextual view of urban reality by paying attention to the geographical impacts of informational economy (Castells 1989). Castells had earlier emphasized in his urban analyses both productive relations (Castells and Godard 1974) and consumption processes (Castells 1977), but now broader exchange, interaction, and communication processes and spaces were taken into the spotlight (Castells 1989; 2002; Stalder 2006; 1998; cf. Hassan 2004, 59). Later the project expanded into an encyclopedic endeavor resulting in the trilogy entitled The Information Age, its three volumes being The Rise of the Network Society (1996), The Power of Identity (1997a), and End of Millennium (1998). The trilogy was essentially a crystallization of Castells' ideas developed throughout the previous decades, augmented by the ample evidence he needed to be able to present a grand narrative of the structural contradiction between the Net and the Self that characterizes the world we live in. The trilogy became a huge success and confirmed Castells' position as one of the most cited social theorists worldwide (Institute of International Studies 2001; Castells and Ince 2003, 18-20).

After retiring from his professorship in UC Berkeley in 2003, Castells accepted a professorship of communication in the prestigious Annenberg School for Communication and Journalism at the University of Southern California, Los Angeles. Castells approached communication in the context of societal changes and power structures, which reveals an unbroken connection with his earlier works (cf. Waterman 1999; see also Kling 2002; Howard 2011; Allan 2007). A cornerstone of this turn was *Communication Power*, which the media analyst Jan van Dijk (2010) has described as the extension of the discussion presented in the second volume of the trilogy, *The Power of Identity* (Castells 2009; on other contributions of Castells' late career, see Carnoy and Castells 2001; Castells 2001; Castells 2007; Arsenault and Castells 2008a; 2008b). We may add to this one of his more recent works, *Networks of Outrage and Hope: Social Movements in the Internet Age*, which illuminates how he fuses his earlier works (e.g. Castells 1983; 1998; 2001) into an insightful analysis of recent social conflicts from the Arab Spring to the Occupy Wall Street movement that are in the spotlight of global news (Castells 2012).

Next I will take a closer look at Castells' concept of network and after that assess its use in his analysis of the network society.

3. The concept of network in Castells' theorization

3.1. Introduction of networks in the informational city

In the light of the Marxist influence it is understandable that the concept of network did not notably figure in Castells' works of the 1970s. His approach changed in the 1980s, and more so in the following decade, however, as he started to renew his conceptual arsenal. The first significant attempt to clarify the network logic within a wider theoretical framework was *The Informational City*, which can actually be seen as the first sketch of *The Information Age* trilogy. Even if it included only a few direct references to networks, the concept itself aptly described something fundamental in the emerging social morphology and related changes in the techno-economic system. Castells (1989, 32) writes,

"These networks, which could not exist on such a large scale without the medium provided by new information technologies, are the emerging organizational form of our world, and have played a fundamental role in ensuring the restructuring process: ---. Networks, on the basis of

new information technologies, provide the organizational basis for the transformation of socially and spatially based relationships of production into flows of information and power that articulate the new flexible system of production and management."

In the analysis presented in *The Informational City* 'network' remained an auxiliary concept subject to the structural framework, having only a modest role in the account of informational capitalism, at least if measured by explicit references to the concept. On the other hand, the few references made by Castells showed the tremendous explanatory power of the concept of network: it depicted the organizational basis for the transformation of production system; it provided a means for corporate system to control its inputs and expand its outputs, thus creating a new global socio-spatial division of labor; its adaptive potential became conducive to the restructuring of the capitalist economy; and it empowered the libertarian spirit of capitalism through organizational networks and information flows which were able to dissolve locales and supersede societies (Castells 1989, 32, 125). The concept of network sketched in *The Informational City* started to resonate later in *The Information Age* as a depiction of the fundamental transformative organizational logic, which relates dialectically to the informational mode of development as an enabler and globalization of the economy as the indication of the transformation of social life (Castells 1996; 1997a; 1998).

3.2. Networks as sets of interconnected nodes

In *The Information Age* and many subsequent works Castells defines 'network' explicitly as a set of interconnected nodesof which he mentions such examples as stock exchange markets and their ancillary centers of advanced financial services in the network of global financial flows; political elites in political networks, such as national councils of ministers and EU commissioners in the governance network of the European Union; broadcasting systems, studios, computer-aided communications, social network service providers in the global network of media, and so forth (Castells 1996, 470). By definition, networks do not have one center but are characterized by binary logic (inclusion/exclusion) and decentralized structures and decision-making patterns. The existence of networks is determined by the utility of the nodes of the network. If some node ceases to serve the network, it will be phased out or replaced and the network rearranges itself in the manner of cells in biological processes. The importance of each node is determined by its ability to gain trust within the network by sharing information and to program and connect networks by mastering protocols, which enable the operation of the critical switches of the network (Castells 1996, 470-471; 2000a; 2000b; 2009; Stalder 2006, 135-136).

Such a formal view derives its rationality from the fact that the network concept was adjusted to a broader structural framework. The introduction of this concept was not motivated by its distinctiveness or content as such, nor did it include any particular methodological contribution, but rather served as a kind of historically grounded epochal "axiom" in a social theoretical deduction from general theoretical principles to the explanation of empirical phenomena, such as new business models, urban conflicts, or state restructuring (Castells 1996; 1997a; 1998). In this sense we may say that the concept of 'network' was subordinate to structural framework and its deductive logic. Castells' insightful idea was to use the global networks of instrumental exchanges as a point of reference to the notion of network. He does not seem to be puzzled by how networks come into being, how they evolve, the nature of given network ties, or even how the information networks really function, but provides only a formal definition and a few sketchy examples, which keeps the discussion at a high level of abstraction (Castells 2000a; 2004a).

Castells assumes that technological development is the most important individual precondition for the resurge of networks (Castells 1996; 2004b; Fuchs 2009). He supports his view with a claim that the emergence of networks as an efficient form of social organization is the result of three features that have proved their usefulness in the emerging techno-economic environment, namely flexibility, scalability and survivability (Castells 2004a, 5-6; cf. Castells 2009, 23),

"Flexibility: they can reconfigurate according to changing environments, keeping their goals while changing their components. They go around blocking points of communication channels to find new connections. Scalability: they can expand or shrink in size with little disruption. Survivability: because they have no center, and can operate in a wide range of configurations, they can resist attacks to their nodes and codes, because the codes of the network are contained in multiple nodes that can reproduce the instructions and find new ways to perform."

To summarize, Castells (2004a; 2009) defines networks as sets of interconnected nodes, which process financial and other value flows with the help of new technologies. The concept of network provided not only opposing model to hierarchy (Eriksson 2005) but also a view to the novel aspects of social organization that emanates from the crumbling foundations of modernity (Giddens 1992), which served perfectly the building of his grand narrative. In Castells' theorization networks are self-configurable, complex structures of communication and power, which cooperate and compete internally and externally according to interests expressed within the nodes using ultimately a binary logic of inclusion/exclusion. They have the capability of self-renewal in the sense that they may introduce new actors and content as conditions change. Their dynamic nature makes them flexible, scalable and survivable, which are indispensable for the survival of organizations in continuously changing techno-economic environment (Castells 1996; cf. Tampere 2011).

3.3. Binary logic of informational capitalism

Castells' theory of network society takes the concept of network to a high level of abstraction, utilizing it as a concept that depicts macro level tendencies associated with the social organization in informational capitalism. He expressed the role of networks in his social theory as follows (Castells 1996, 469):

"... dominant functions and processes in the information age are increasingly organized around networks. Networks constitute the new social morphology of our societies, and the diffusion of networking logic substantially modifies the operation and outcomes in processes of production, experience, power, and culture. While the networking form of social organization has existed in other times and spaces, the new information technology paradigm provides the material basis for its pervasive expansion throughout the entire social structure."

Understanding the societal context of such networks entails returning to the political economy of the social transformation of capitalist society. Initially, as an evocation of his Marxist roots, Castells repeatedly emphasized that the network society is a capitalist society. In fact, for the first time almost the entire globe can be said to function under the conditions of the capitalist system (Castells 2000a). The most central tension of such a societal formation is that between capital and labor, as depicted in Karl Marx's thesis of the in-built contradiction of capitalism. According to Castells, a key contradiction critical to understanding informational capitalism is the historical asymmetrization of the capital-labor relationship: while capital creates networks, labor becomes individualized (Castells 1996, 471; 1997b).

Informational capitalism is based on the informational mode of development, in which information becomes both key input and outcome of production processes and a source of productivity and growth. It makes informationalism in essential sense different from agrarian and industrial modes of development. Historically, such a new focus on information processing was accompanied by three key processes of the restructuring of capitalism that emerged in the 1980s and the following decades, those of increased capital intensity, changes in the pattern of state intervention, and accelerated globalization of all economic activities (Castells 1989; see also Schmiede 2006).

Informational economy works through global networks of instrumental exchanges. As they process flows within the fluid "space of flows," relationships surpass places in importance in the functioning of the system. The other side of the equation, waged labor, became historically united

against capital in the form of the labor movement from the late 19th century onwards. Disillusionment with socialism, the anomalies of Marxist doctrine and real-life developments in advanced democracies caused the inevitable evanescence of the subject of revolution. Castells, like many of his post-Marxist contemporaries, called for a more subtle and nuanced interpretation of the conditions and agenda of social change. The most serious attempts to restore the actorship behind radical social change have rested on social and ecological movements and more broadly on civil society (Kling and Posner 1991, 30; see e.g. Wright 1997; 2005; Katznelson 1981; Marcuse 1964). Castells sought individual and community-level awareness, adaptability and innovativeness in a pluralist society without committing to any ideologically fixed manifesto or political program (Institute of International Studies 2001; cf. Waterman 1999; Boyraz 2008; Rantanen 2005). Thus the analysis of the network society did not aim at providing a formula for the working class to solve the contradiction between capital and labor, but rather at opening up new horizons to understand our social conditions and to challenge global instrumentality through bottom-up strategies developed by people who ultimately rely on the power of their locally-rooted socio-historical identities.

Macro level network logic comes into this picture in Castells' dramatic claim that individuals, groups, communities, and even nations are included or excluded from the networks of economic power depending on their usefulness to such networks. The processes of human life are increasingly conditioned by global economic networks that position people according to their "use value" and create sophisticated means of controlling everyday life. This creates tension, which Castells (1996, 3) expressed in one of the most widely quoted crystallizations of his theory: "Our societies are increasingly structured around a bipolar opposition between the Net and the Self." Hence the emancipatory message built into Castells' theory and the related special nature of "power of identity" as counterforce to instrumental networks.

3.4. From neuropolitics to general network theory

The most recent phase of Castells' academic career has been devoted to the analysis of communication, power, and networks. Even if he adheres to topics elaborated in his earlier works, his interest in micro sociology gained ascendancy. According to van Dick (2009), Castells' main work in this area, *Communication Power*, illustrates the change of his thinking in two respects. First, tying the analysis to self and identity is not as superficial as it was in *The Information Age*, but developed a firmer psychological foundation for understanding social action. In this he relies on neuroscience, as exemplified the works of Professor Antonio Damasio (1994) of the University of Southern California, who has contributed to shifting the center of gravity in the analysis of human behavior from cognition and rationality to biology and emotions (cf. Kelly 1994). We may identify similar aspirations in Castells' attempt to apply Robert Entman's (1993) frame theory in analyzing power in communication. Another important departure from Castells' earlier approach was to take the internal power struggles of networks onto the agenda, which in his earlier works were only mentioned in passing (van Dijk 2001; 2010). Fascination with micro sociological explanatory schemes began to profoundly affect Castells' thinking (see Castells 2009).

In Castells' analysis network appears as a central structural element in the new forms of communication and in a mediatized society in general, which is manifest in the "mass self-communication" associated with social media, in the environmental movement's contribution to mitigate climate change, and in neuropolitics, which analyzes power through cognitive processes (Castells 2009; cf. Castells 2012). He became interested in biological analogies and neuroscience, which tries to explain how political affiliation and action relate to the human brain (e.g. Lakoff 2008). Castells ended up with a notion that as conscious actors we resemble organic networks with a connection to the external world of networks (Delfanti 2009). From that assumption it is only a short step to a general network theory, which makes sense of the material, psychological, and social aspects of our world,

serving as a unifying framework for the natural sciences, humanities and social sciences in the same way as general systems theory or complexity theory were assumed to do (Castells 2009; 2010).

Castells' micro-sociological excursions have been criticized for making an excessively simplistic use of the results of brain research to explain complex phenomena, such as power and politics. It is also worth asking how useful the network metaphor actually is when taken to its extreme, i.e. to explain practically everything, as Castells eventually seems to do (see Delfanti 2009; Stalder 2006; Thompson 2003, 192). This raises an interesting question of the relation between macro and micro in Castells' theorization. The concept of network has allowed Castells to maintain his macro sociological framework while deepening his analysis along micro sociological lines.

Castells placed great faith in the network theory, proposing that it may even offer a unifying language and framework to understand nature and human society through the networks of biological, neurological, digital, and human communication (Castells 2010). While for understandable reasons he maintained his belief in his theory of network society, his recent views of the role of networks in social research would require the refinement of the fundamentals of his theory if applied to his analysis. Such an endeavor is akin not only to the previously mentioned micro sociological endeavors (Damasio 1994; Kelly 1994; Entman 1993; Lakoff 2008) but also to complexity theory and the related new network conceptions, as represented by Capra (2003) and Barabási (2002). This is a methodological project which Castells touched upon in *Communication Power* and referred to in some of his speeches (e.g. Castells 2010) but which he obviously did not set himself to complete.

4. Castells in the field of network research

Castells is a renowned social theorist, whose influence is perceptible in various ways in various academic and social discourses of our time (McCarthy *et al.* 2004; May 2006; Barney 2004, 181). His trilogy in particular introduced a long awaited social theoretical perpective on the understanding of the trends and challenges of global transformation (Fuller 2004; Giddens 1996; Cabot 2003; Heiskala 2003). If we take the concept of network to be the core of his analysis, however, the picture changes in some respects. Namely, paradoxically Castells' influence is minimal in network research even though his network society theory is one of the most widely discussed social theoretical contributions of the 1990s.

4.1. Social network analysis

Castells' theorization is not imbued by references to social anthropologists who have studied social ties (e.g. Alfred Radcliffe-Brown and E.E. Evans-Pritchard), sociological classics that contributed to the network-oriented analysis of social structures and interaction (Georg Simmel, Leopold von Wiese, Norbert Elias, Herbert Blumer) or later generations of scholars who have developed social network analysis, including J. Clyde Mitchell, Harrison White, Michael Schwartz, Barry Wellman, and Mark Granovetter (on the sociology of social networks, see Freeman 2004; Pescosolido 2006; Borgatti*et al.* 2009). In its basic form social network analysis focuses on the interaction of small groups and dyadic relationships (Granovetter 1973; 1983), which are not addressed in Castells' analysis. He conceptualized "social ties" as imprecise impersonal exchange relations at a theoretically constructed whole-network level, relying on benefit-based antecedents or the utility maximization view of network formation as a reflection of instrumentality built into the logic of the capitalist system. While directing attention one-dimensionally to the formal features of network-based organization, he eventually totally excluded 'social interaction' from his network analysis.

Another noteworthy point is that in his analysis the economic sphere had a critical role as the primary locus of the network society, which is a fairly narrow slice of network theory as a whole. In this sense similarities with Castells' network conception could be sought from economic network analysis – the representatives of this approach being Christopher Freeman and Walter W. Powell – and within it especially from an approach in which the proliferation of networks is associated with technological and

economic development. More precisely, Castells' 'network' can be seen as institutionally defined concept embedded in the macro-theoretically oriented economic network analysis (Castells 1996; 2000a; 2000b; 2010). In short, for Castells its role in defining social morphology and related institutional effects (Owen-Smith and Powell 2008) are more important than the factual social relations within each network.

If Castells overhauled the traditions of the social network analysis lightly, it is not surprising that his network concept remained too sketchy for the network analysts. Castells' contribution to social research is not based on the sophistication of the network concept *per se*. Network society theory attracted attention because it afforded a rich perspective on societal change in which network was a social theoretical axiom or metaphor rather than an analytical concept. Be that as it may, social scientists who analyze network society at the macro-theoretical level have inevitably benefited from Castells' seminal work, which paves the way for a better understanding of the emerging forms of the social organization of society and their structural conditions (e.g. Barry 2001; Thompson 2003; Barney 2004; Lehmann *et al.* 2007).

4.2. Remarks on network society theorizations

If we discuss the concept of 'network' at a higher level of abstraction, within macro sociology or political economy, Castells' contribution gains currency. In the 1970s the center of gravity of network research in general shifted to sociology (Borgattiet al. 2009). Along with this shift network society theorizations emerged as a reflection of the increased complexity and interactivity occasioned by the digitalization, informatization, and mediatization of society (Craven and Wellman 1973; Wellman 1979; Hiltz and Turoff 1978; Martin 1977; see also van Dijk 1999). Other contributions to this field came from discussions about post-industrial society (Bell 1973; Kumar 2005), various strands of information and knowledge society discourses (see Webster 1995; Stehr 1994), theorizations of reflexive or late modernity in sociology (Giddens 1992; Beck *et al.* 1994; Lash and Urry 1994), and insightful views in futures research (Toffler 1970; 1980; 1990; Naisbitt 1984) to mention just a few prominent discourses that relate to network society theorizations. In this field Castells' work stands out as an unparalleled encyclopedic, critical social-theoretical endeavor that links structural analysis of global and societal transformations with various aspects of economy, power, and everyday life.

Castells did not strongly relate his discussion to any theoretically-oriented sociological tradition but set his own agenda (see lyer 1999; cf. Collins 1999; Smart 2000; Holton 2005). Most notably, he rejected the concepts of information and knowledge as primary explanatory categories and assigned such a role to networks. He made network a basic unit of analysis, which integrated social structures, social action, social organization, space of flows, and new technologies into a macro theoretical framework that utilizes quasi-Marxist ideas, such as the informational mode of development. With this solution he contributed to the wider acceptance of the concept of network as an interpretative norm in social sciences (Kasvio 2005; May 2006; Stalder 1998; on the network society analyses published after Castells' trilogy, see Schiller 2000; Thompson 2003; Barney 2004; Hassan 2004; McCarthy *et al.* 2004; Castells and Cardoso 2005; Lehmann *et al.* 2007; Cardoso 2007). Networks have undoubtedly been emphasized in economy and society long before Castells (Bridge and Watson 2003, 107), but none of the earlier works used network as the basic category of the analysis of capitalist system as Castells did in his theory.

5. Discussion

In numerous evaluations of his trilogy, Castells is usually unquestioningly applauded for identifying the profound social trends and their inter-relationships and for describing them with illustrative concepts that are incorporated into his social theoretical framework (Giddens 1996; Wilenius1998; Fischer 1999; Heiskala 2003). On the other hand, *The Information Age* does not critically analyze the themes it covers, why references to alternative approaches, theories and concepts remain
superficial or mostly non-existent (lyer 1999). The thematically broad agenda of the trilogy actually includes quite a lot of conclusions, which cannot be derived from the given premises, and generalizations that are too bold. Moreover, Castells' affinity with novel, emerging forms of social life increases the sketchiness and the imprecision of his generalizations (Barney 2004). These points are good to keep in mind when assessing his way of understanding and using the concept of network.

To begin with, Castells uses network as a theoretical concept that refers to the general form of social organization and development, which can be best referred to as a network metaphor as it lacks explanatory accuracy and analytical depth concerning the dynamics of networks. As an analytical concept network is abstract, and thus unable to frame the interpretation of real-life networks, whereas as a theoretical concept network is actually an excellent crystallization of the social morphology of informational capitalism. As an upshot of the latter, the concept of network society has a certain intellectual appeal, even if it looks almost as if the formal description of the concept of network was needed only to legitimate its use as a metaphor. Concerning the hard core of the metaphor, we come to the true message of Castellsian political economy: the network in its paradigmatic form is about the nodes and connections of multinational corporations and powerful economic institutions that process flows of values in pursuit of the accumulation of capital. Such networks of instrumental exchanges are challenged by social movements and people with their historical and locally-rooted identities, which are a source of genuine human meaning as opposed to global instrumentality. Hence the fundamental tension between the Net and the Self. Such a view echoes Castells' early adherence to Marxist analysis, even if purged of ideological fixities.

Castells built an insightful analysis, which was seemingly affected by a soft version of technological determinism. It was, however, surpassed by structurally oriented social determinism as evidenced by the "axiomatic" use of network. Concerning the interplay of these two forms of determinism, it is worth noting that in the 1990s, when new technologies penetrated developed countries and appeared to shape all aspects of capital accumulation and professional and everyday life, the use of informationalism and network logic to explain novel phenomena started to produce diminishing returns. The explanatory scheme that appeared to work well with the macro theoretical analysis in *The Informational City* and *The Information Age* trilogy at a later stage became disruptive when more nuanced analyses of social reality were clearly needed. As aptly pointed out by Harris (2010, 409) in his review of Castells' book *The Internet Galaxy*, the book in question offers a distinctive antidote to the teleology and hype on Internet and ICTs but,

"... stops short of explaining the consequences of these technologies in purely 'political' terms, relying instead on the network metaphor to unravel the complexities and contradictions of the digital age."

In the same vain, to explain what Castells wanted to explain in *Networks of Outrage and Hope* would have required a more nuanced understanding of the Internet and social media in everyday life, as pointed out by Barassi (2013) and Fuchs (2012). Castells perceives protest movements essentially as autonomous communication networks supported by the Internet and wireless communication irrespective of the actual penetration, use, and significance of ICTs in the given real-life setting. Likewise, if everything from Facebook to protests in the streets of Seattle or New York to an interconnected stock market system is explained in terms of network logic, the ability to accurately account for their emergence, forms, and operations becomes difficult. In such analyses the metaphorical use of the concept of network appears to be a social counterpart of technological determinism with less and less relevance as a formal conceptual tool.

Conclusion

The point of departure of Castells' intellectual conviction was Marxist urban sociology, which over the years developed into a social critique of the network society. Networks came into the picture as late as in the latter half of the 1980s, when Castells devoted himself to the macro-theoretical analysis of the relationship between technology, economy, and society, which culminated in the breakthrough trilogy, *The Information Age* (Castells 1996; 1997a; 1998). At this point the critical role of networks for social theory became apparent. Castells came to the conclusion that the concept of information society should be replaced by the notion of network society. This solution did not necessarily go from bad to any better, as pointed out by Kasvio (2005), but showed in any case how Castells distanced himself from conventional information society theories and the sociology of knowledge.

The reviewers of Castells' trilogy and of a series of books that followed it (Castells 2001; 2009; 2012) have repeatedly pointed out the deficiencies of his conceptualization of network and the obvious lack of related empirical evidence. Indeed, the institutional or metaphorical use of the network concept in theory building has a tendency to reify networks and limit our understanding of their inner logic. In spite of leaving social relations within networks only implicit, Castells was nevertheless able to demonstrate the power of institutional concept of network as a constitutive element of his grand theory. Retrospectively, Castells' theory of network society served as a long awaited structural framework for understanding contemporary societies and the human condition in a globalized world. It is a source of inspiration for those wishing to develop ideas for institutional renewal and contextualize the aspects of living in the global. It paints a rich picture of contemporary society, which in the spirit of C. Wright Mills' (1959) idea of sociological imagination helps us to conceive of ourselves as humans in a rapidly changing world capable of resisting the pervasive instrumentality exuded from the pores of global informational capitalism.

References

- [1] Allan, S. (2007). Citizen Journalism and the Rise of "Mass Self-Communication": Reporting the London Bombings. *Global Media Journal Australian* 1(1): 1-20, <u>http://www.hca.uws.edu.au/gmjau/archive/iss1_2007/stuart_allan.html</u>(accessed July 9, 2011).
- [2] Arsenault, A., Castells, M. (2008a). Switching Power: Rupert Murdoch and the Global Business of Media Politics. A Sociological Analysis. *International Sociology* 23(4): 488-513.
- [3] Arsenault, A., Castells, M. (2008b). The Structure and Dynamics of Global Multi-Media Business Networks. *International Journal of Communication*, 2: 707-748.
- [4] Barabási, A.-L. (2002). *Linked. The New Science of Networks*. Perseus Publishing.
- [5] Barassi, V. (2013). Review Networks of Outrage and Hope. *E-International Relations*, <u>http://www.e-ir.info/2013/02/27/review-networks-of-outrage-and-hope/(accessed 16 August 2014)</u>.
- [6] Barney, D. (2004). *The network society*. Polity Press.
- [7] Barry, A. (2001). Political machines: governing a technological society. Athlone Press.
- [8] Beck, U., Giddens, A., Lash, S. (1994). *Reflexive Modernization. Politics, Tradition and Aesthetics in the Modern Social Order.* Polity Press.
- [9] Bell, D. (1973). The Coming of Post-Industrial Society. A Venture in Social Forecasting. Basic Books.
- [10] Borgatti, S.P., Mehra, A., Brass, D., Labianca, G. (2009). Network Analysis in the Social Sciences. *Science*, 323(5916): 892-895.
- [11] Boyraz, C. (2008). The Information Age: Economy, Society and Culture, Volume I-II-III by Castells, Book review, International Journal of Social Inquiry 1(2), <u>http://www.socialinquiry.org/</u> <u>articles.asp?id=31</u>(accessed September 20, 2015).

- [12] Bridge, G., Watson, S. (2003). City Economies. In: Bridge, G. and Watson, S. (eds) A *Companion to the City*. Blackwell.
- [13] Cabot, J.E. (2003). The information age; Manuel Castells; the rise of the network society. *Research Policy* 32: 1141-1162.
- [14] Capra, F. (2003). *The Hidden Connections. A Science for Sustainable Living*. HarperCollins. Flamingo.
- [15] Cardoso, G. (2007). *The Media in the Network Society: Browsing, News, Filters and Citizenship*. CIES Centre for Research and Studies in Sociology.
- [16] Carnoy, M., Castells, M. (2001). Globalization, the knowledge society, and the Network State: Poulantzas at the millennium. *Global Networks*, 1(1): 1-18.
- [17] Castells, M. (1977). *The Urban Question. A Marxist Approach*. From original work La question urbaine translated by Alan Sheridan. Edward Arnold.
- [18] Castells, M. (1978). *City, Class and Power*. Macmillan.
- [19] Castells, M. (1983). The City and the Grassroots: A Cross-cultural Theory of Urban Social Movements. University of California Press.
- [20] Castells, M. (1985). High Technology, Economic Restructuring, and the Urban-Regional Process in the United States. In: Castells M (ed) *High Technology, Space, and Society*. Sage.
- [21] Castells, M. (1989). The Informational City. Information Technology, Economic Restructuring, and the Urban-Regional Process. Blackwell.
- [22] Castells, M. (1996). *The Rise of the Network Society*. The Information Age. Economy, Society and Culture, Volume I. Blackwell.
- [23] Castells, M. (1997a). *The Power of Identity*. The Information Age. Economy, Society and Culture, Vol. II. Blackwell.
- [24] Castells, M. (1997b). An Introduction to the Information Age. City 2(7): 6-16.
- [25] Castells, M. (1998). *End of Millennium*. The Information Age. Economy, Society and Culture, Vol. III. Blackwell.
- [26] Castells, M. (2000a). Materials for an exploratory theory of the network society. British Journal of Sociology, 51(1): 5–24.
- [27] Castells, M. (2000b). Toward a Sociology of the Network Society. *Contemporary Sociology*, 29(5): 693–699.
- [28] Castells, M. (2001). The Internet Galaxy: Reflections on the Internet, Business and Society. Oxford University Press.
- [29] Castells, M. (2002). Local and Global: Cities in the Network Society. *TijdschriftvoorEconomische en SocialeGeografie*, 93(5): 548–558.
- [30] Castells, M. (2004a). Informationalism, networks, and the network society: a theoretical blueprint. In: Castells, M. (ed) *The Network Society: a cross-cultural perspective*. Edward Elgar, pp. 3-7, 36-45.
- [31] Castells, M. (2004b). Afterword: Why networks matter? In: McCarthy, H., Miller, P. and Skidmore, P. (eds) Network Logic. Who governs in an interconnected world? Demos Collection, issue 20. Demos, <u>http://www.demos.co.uk/files/File/networklogic17castells.pdf</u>(accessed September 25, 2015).

- [32] Castells, M. (2007). Communication, Power and Counter-power in the Network Society. *International Journal of Communication* 1: 238-266.
- [33] Castells, M. (2009). Communication Power. Oxford University Press.
- [34] Castells, M. (2010). Introduction in International Seminar on Network Theory: Network Multidimensionality in the Digital Age. The Annenberg Networks Network (ANN), University of Southern California. YouTube video, <u>http://www.youtube.com/watch?v=vDh7eGJwhwg</u>(accessed March 22, 2011).
- [35] Castells, M. (2012). *Networks of Outrage and Hope: Social Movements in the Internet Age*. Polity Press.
- [36] Castells, M., Cardoso, G. (eds) (2005). The Network Society: From Knowledge to Policy. Washington, DC: Center forTransatlantic Relations, The Paul H. Nitze School of Advanced International Studies, The Johns Hopkins University, <u>http://www.umass.edu/digital</u> <u>center/research/pdfs/JF_NetworkSociety.pdf</u> (accessed July 10, 2011).
- [37] Castells, M., Fernandez-Ardevol, M., Qiu, J.L., Sey, A. (2004). The Mobile Communication Society. A cross-cultural analysis of available evidence on the social users of wireless communication technology. A research report prepared for the International Workshop on Wireless Communication Policies and Prospects: A Global Perspective, held at the Annenberg School for Communication, USC, Oct 8-9, 2004, <u>http://www.scribd.com/doc/6639987/Social-Uses-of-Wireless-Communications-the-Mobile-Information-Society-By-Manuel-Castells</u> accessed July 7, 2011.
- [38] Castells, M., Godard, F. (1974). *Monopolville. Analyse des rapports entre l'entreprise, l'État et l'urbain à partir d'une enquête sur la croissance industrielle et urbaine de la région de Dunkerque*. (Avec la collaboration de Vivian Balanowski). Mouton.
- [39] Castells, M., Ince, M. (2003). Conversations with Manuel Castells. Polity Press.
- [40] Castells, M., Murphy, K. (1982). Cultural Identity and Urban Structure: The Spatial Organization of San Francisco's Gay Community. In: Fainstein, N.I. and Fainstein, S.S. (eds) Urban Policy under Capitalism. Sage.
- [41] Cochrane, A.D. (2003). The Social Construction of Urban Policy. In: Bridge, G. and Watson, S. (eds) *A Companion to the City*. Blackwell.
- [42] Collins, S. (1999). Finding the Event Horizon along the Black Holes of Network Society. Review of Castells, Manuel, The Rise of Network Society, Castells, Manuel, The Power of Identity and Castells, Manuel, End of Millennium. *H-USA, H-Net Reviews in the Humanities & Social Sciences*, September, 1999, <u>http://www.h-net.org/reviews/showrev.php?id=3399</u>(accessed 10 July 2011).
- [43] Craven, P., Wellman, B. (1973). The Network City. Sociological Inquiry, 43: 57-88.
- [44] Damasio, A. (1994). Descartes' Error: Emotion, Reason, and the Human Brain. Avon Books.
- [45] Delfanti, A. (2009). Too much power to the networks. Review. Journal of Science Communication 8(4): 1-2, <u>http://jcom.sissa.it/archive/08/04/Jcom0804%282009%29R01/Jcom0804%282009%29R01/Jcom0804%282009%29R01.pdf</u> (accessed May 30, 2011).
- [46] De Moura Costa, H.S. (2007). Some theoretical formulations on the production of popular residential land developments in Belo Horizonte, Brazil. etc..., espaço, tempo e crítica, Revista Eletrônica de Ciências Sociais Aplicadas e outrascoisas, Volume I, 2(3): 1: 21-37, <u>http://www.uff.br/etc/UPLOADs/etc%202007_2_3.pdf</u>(accessed September 25, 2015).

- [47] Dunleavy, P. (1977). Protest and quiescence in urban politics. *International Journal of Urban and Regional Research* 1(1-4): 193–218.
- [48] Dunleavy, P. (1980). Urban Political Analysis: the Politics of Collective Consumption. Macmillan.
- [49] Elliott, B. (1980). Manuel Castells and the New Urban Sociology. *The British Journal of Sociology* 31(1): 151-158.
- [50] Entman, R.M. (1993). Framing: Toward Clarification of a Fractured Paradigm. *Journal of Communication*, 43(4): 51-58.
- [51] Eriksson, K. (2005). On the Ontology of Networks. *Communication and Critical/Cultural Studies* 2(4): 305-323.
- [52] Fischer, J. (1999). Manuel Castells' Brave New World. Reprinted from *West*, <u>http://www.acturban.org/biennial/DOC_planners/castells_article9904.pdf</u> (accessed_July_5, 2011).
- [53] Freeman, L.C. (2004). The development of social network analysis: a study in the sociology of science. Empirical Press.
- [54] Fuchs, C. (2008). International Sociological Association (ISA) World Forum: Day 2: Alain Touraine, Manuel Castells, Craig Calhoun. Information – Society – Technology & Media. Posted September 7, 2008, <u>http://fuchs.uti.at/195/</u> (accessed August 2, 2011)
- [55] Fuchs, C. (2009). Some Reflections on Manuel Castells' Book 'Communication Power'. *TripleC: Communication, Capitalism & Critique,* 7(1): 94-108.
- [56] Fuchs, C. (2012). Some Reflections on Manuel Castells' Book Networks of Outrage and Hope: Social Movements in the Internet Age. *Triple C: Communication, Capitalism & Critique,* 10(2): 775-797.
- [57] Fuller, S. (2004). Manuel Castells' Trilogy, the Information Age: Economy, Society and Culture. Book Review. A Journal of Cybernetic Revolution, Sustainable Socialism, and Radical Democracy, Issue 8, Winter, <u>http://www.net4dem.org/cyrev/archive/issue8/articles/ Castells Review/CastellsReview1.htm</u> (accessed May 30, 2011).
- [58] Giddens, A. (1992). The Consequences of Modernity. Polity Press.
- [59] Giddens, A. (1996). *Out of place*. A Review of Manuel Castells' Book 'The Rise of the Network. Society'. The Times Higher Education Supplement (London), 18.
- [60] Granovetter, M.S. (1973). The Strength of Weak Ties. The American Journal of Sociology, 78(6): 1360–1380.
- [61] Granovetter, M.S. (1983). The Strength of Weak Ties: A Network Theory Revisited. *Sociological Theory*, 1: 201-233.
- [62] Gurr, T.R., King, D. (1987). The State and the City. Macmillan.
- [63] Harding, R. (2006). Manuel Castells' Technocultural Epoch in the Information Age. *Science Fiction Studies*, 33(1): 18-29.
- [64] Harris, M. (2010). Complexities and contradictions of networked places. *City*, 6(3): 407-409.
- [65] Hassan, R. (2004). *Media, Politics and the Network Society*. Open University Press.
- [66] Heiskala, R. (2003). Informational revolution, the net and cultural identity. A conceptual critique of Manuel Castells' "The Information Age". *European Journal of Cultural Studies*, 6(2): 233–245.

- [67] Hiltz, S.R., Turoff, M. (1978). *The Network Nation: Human Communication via Computer*. Addison-Wesley.
- [68] Holton, R.J. (2005). Network discourses: proliferation, critique and synthesis. Review Essay. *Global Networks*, 5(2): 209-215.
- [69] Howard, P.N. (2011). Castells and the Media. Polity Press.
- [70] Hydén, H. (2001). Manuel Castells och förebådandet av en ny rättskultur. *Retfaerd* 24(94): 3-19, <u>http://retfaerd.org/retfaerd-nr-94-20013/haakan-hyden-manuel-castells-ochforebaadandet-av-en-ny-rattskultur/</u> (accessed September 25, 2015)
- [71] Iyer, S.D. (1999). Book Review: The Rise of the Network Society. Solstice: An Electronic Journal of Geography and Mathematics X(1), <u>http://hdl.handle.net/2027.42/60263</u> (accessed July 10, 2011)
- [72] Kasvio, A. (2005). What Information Society? In: Kasvio A and Anttiroiko A-V (eds) e-City: Analysing Efforts to Generate Local Dynamism in the City of Tampere. Tampere University Press, pp. 619-643.
- [73] Katznelson, I. (1981). *CityTrenches: Urban Politics and the Patterning of Class in the United States*. Pantheon Books.
- [74] Katznelson, I. (1992). *Marxism and the City*. The Clarendon Press.
- [75] Kelly, K. (1994). Out of Control: The Rise of Neo-Biological Civilization. Addison-Wesley.
- [76] Kling, J.M., Posner, P. (1991). Class and Community: Theories of Activism in an Era of Urban Transformation. In: Smith, M.P. (ed) *Breaking chains: social movements and collective action*. Comparative Urban and Community Research, Vol. 3. Transaction Publishers.
- [77] Kling, R. (2002). The Internet Galaxy: Reflections on the Internet, Business, and Society. Book Reviews. Academe Online, July-August 2002, <u>http://www.aaup.org/AAUP/pubsres/academe/</u>2002/JA/BR/Klin.htm (accessed July 20, 2011)
- [78] Kumar, K. 2005. From Post-Industrial to Post-Modern Society: New Theories of the Contemporary World. Second Edition. Blackwell.
- [79] Lakoff, G. 2008. The Political Mind: Why You Can't Understand 21st-Century American Politics with an 18th-Century Brain. Viking Penguin.
- [80] Lash, S., Urry, J. 1994. *Economies of Signs and Space.* Sage.
- [81] Lehmann, N., Qvortrup, L. and Walther, B.K. (eds) 2007. *The Concept of the Network Society: Post-Ontological Reflections*. Samfundslitteratur.
- [82] Lojkine, J. 1976. Contributions to a Marxist theory of urbanization. In: Pickvance, C. (ed) *Urban sociology: Critical essays*. Tavistock, pp. 119-146.
- [83] Lowe, S. 1986. Urban Social Movements. The City after Castells. Macmillan.
- [84] Marcuse, H. 1964. One-Dimensional Man:Studies in the Ideology of Advanced Industrial Society. Beacon.
- [85] Martin, J. 1977. The Wired Society: A Challenge for Tomorrow. Prentice-Hall.
- [86] May, C. 2006. The Network Society. Key Concepts Series. By Darin Barney. Reviews. Canadian Journal of Communication 31(3): 769-770, <u>http://www.cjc-online.ca/index.php/journal/</u> <u>article/view/1659/1801</u>(accessed June 6, 2011).

- [87] Mayer, M. (2006). Manuel Castells' the City and the Grassroots. International Journal of Urban and Regional Research 30: 202–206.
- [88] McCarthy. H., Miller, P., Skidmore, P. (2004). *Network Logic. Who governs in an interconnected world?* Demos.
- [89] Merrifield, A. (2002). *Metromarxism. A Marxist Tale of the City*. Routledge.
- [90] Mills, C.W. (1959). The Sociological Imagination. Oxford University Press.
- [91] Mollenkopf, J.H., Castells, M. (eds) (1991). *Dual City. Restructuring New York*. Russell Sage Foundation.
- [92] Naisbitt, J. (1984). *Megatrends. Ten New Directions Transforming Our Lives*. Futura.
- [93] Owen-Smith, J., Powell, W.W. (2008). Networks and Institutions. In: *The Sage Handbook of Organizational Institutionalism*. Sage, pp. 596-623.
- [94] Pescosolido, B.A. (2012). The Sociology of Social Networks. In: Bryant, C.D. and Peck, D.L. (eds) *The Handbook of 21st Century Sociology*. Sage, pp. 208-217.
- [95] Pickvance, C.G. (1975). Review Essay: Toward a Reconstruction of Urban Sociology. American Journal of Sociology 80(4): 1003-1008.
- [96] Portes, A., Castells, M., Benton, L.A. (eds) (1989). *The Informal Economy: Studies in Advanced and Less Developed Countries*. The Johns Hopkins University Press.
- [97] Rantanen, T. (2005). The message is the medium. An interview with Manuel Castells. *Global Media Communication*, 1(2): 135-147.
- [98] Rex, J. (1977). Sociological Theory and the City A Response to Some Recent Trends in Australasian Urban Sociology. *Journal of Sociology*, 13(3): 218-223.
- [99] Schiller, D. (2000). Digital Capitalism: Networking the Global Market System. MIT Press.
- [100] Schmiede, R. (2006). Knowledge, work and subject in informational capitalism. In: Berleur, J., Nurminen, M.I. and Impagliazzo, J. (eds) Social informatics – an information society for all? Proceedings of the 7th International Conference on Human Choice and Computers. Springer.
- [101] Smart, B. (2000). A Political Economy of New Times? Critical Reflections on the Network Society and the Ethos of Informational Capitalism. *European Journal of Social Theory*, 3(1): 51-65.
- [102] Stalder, F. (1998). The Network Paradigm: Social Formations in the Age of Information. Review Essay. *The Information Society*, 14(4), <u>http://www.indiana.edu/~tisj/readers/full-text/14-4%20Stalder.html</u> (accessed June 7, 2011).
- [103] Stalder, F. (2006). Manuel Castells and the Theory of the Network Society. Polity Press.
- [104] Stehr, N. (1994). Knowledge Societies. Sage.
- [105] Tampere, K. (2011). A Walk in the Public Relations Field: Theoretical discussions from a social media and network society perspective. *Central European Journal of Communication* 4(1): 49-61.
- [106] Thompson, G.F. (2003). Between hierarchies and markets: the logic of network forms of organization. Oxford University Press.
- [107] Toffler, A. (1970). *Future Shock*. Bantam Books.
- [108] Toffler, A. (1980). *The Third Wave*. Bantam Books.

- [109] Toffler, A. (1990). *Powershift: Knowledge, Wealth and Violence at the Edge of the* 21st Century. Bantam Books.
- [110] Walton, J. (1993). Urban Sociology: The Contribution and Limits of Political Economy. *Annual Review of Sociology* 19: 301-320.
- [111] Waterman, P. (1999). The Brave New World of Manuel Castells: What on Earth (or in the Ether) is Going On? Review Article. *Development and Change* 30(2): 357–380.
- [112] Webster, F. (1995). *Theories of the Information Society*. Routledge.
- [113] Wellman, B. (1979). The Community Question: The Intimate Networks of East Yorkers. *American Journal of Sociology* 84: 1201-1231.
- [114] Whiteley, P. (1980). Review of Manuel Castells' City, Class and Power. Journal of Social Policy 9(1): 135-137.
- [115] Wilenius, M. (1998). A New Globe in the Making: Manuel Castells on the Information Age. Review Essay. *Acta Sociologica* 41: 269-276.
- [116] Wright, E.O. (1997). *Class Counts: Comparative Studies in Class Analysis*. Cambridge University Press.
- [117] Wright, E.O. (2005). Approaches to Class Analysis. Cambridge University Press.
- [118] Van Dijk, J. (1999). The Network Society: Social aspects of new media. Thousand Oaks. Sage.
- [119] Van Dijk, J. (2001). The One-dimensional Network Society of Manuel Castells. A Review Essay. *Chronicle World*. 07-Jan-01, <u>https://www.utwente.nl/bms/vandijk/research/network_society/</u> <u>network_society_plaatje/d_castells_review_castells_net.pdf</u>(accessed September 25, 2015).
- [120] Van Dijk, J. (2010). Review of *Communication Power*, by Manuel Castells. *Communications: The European Journal of Communication Research*, 35(4): 485-500.
- *** Institute of International Studies. (2001). Identity and Change in the Network Society. Conversation with Manuel Castells. Conversations with History: Institute of International Studies, UC Berkeley, interviewed by Harry Kreisler, <u>http://globetrotter.berkeley.edu/people/Castells/castells-con0.html</u> (accessed July 5, 2011).

Programs and Publications

- Upcoming Events...
- Journals...
- Conferences Proceedings...
- Books Collections....

Journals ...

Journal of Advanced Research in Entrepreneurship and New Venture Creation



Editor in Chief PhD ProfessorDomenico NICOLÒ MediterraneanUniversity of Reggio Calabria, Italy

The Journal of Advanced Research in Entrepreneurship and New Venture Creation (JARE-NVC) seeks to promote further ideas and debates on the international, cross-cultural and comparative academic research about entrepreneurs and entrepreneurship. Manuscripts that are suitable for publication in JARE-NVC cover domains such as, but are not limited to these: Entrepreneurship and growth; Entrepreneurship support and policy implications; New firms and regional context; Academic entrepreneurship and technology transfer; International entrepreneurship; Innovation activities in SME; Entrepreneurship; Start-ups development (or growth) strategies; Start-ups competitive strategies; Risk factors for start-ups; Business planning models; Family business; Entrepreneurial values and business success; Smart cities policies and start-ups; Venture capital and Private Equity (or Private equity investors); Venture creation models (or patterns); Start-ups Ecosystems.

Journal of Advanced Research in Entrepreneurship and New Venture Creation, starting with its first issue, will be indexed in <u>RePEC</u>, <u>CEEOL</u>, <u>ProQuest</u>, and <u>EBSCO</u>databases.

Web: <u>http://www.asers.eu/journals/jare_nvc</u>

E-mail: <u>asers@asers.euasers2010@yahoo.co.uk</u>

VOIGINO VI, 10000 I(II) OGININOI 2010

Journal of Advanced Research in Economics and International Business



Editor in Chief: PhD MădălinaCONSTANTINESCU SpiruHaret University, Romania

Co-Editor: PhD Daniele SCHILIRÒ Messina University,Italy

Journal of Advanced Research in Economics and International Business provides a forum where academics and professionals can share the latest developments and advances in the knowledge and practice of Economics and International Business. It aims to foster the exchange of ideas on a range of important international subjects, to provide stimulus for research and the further development of international perspectives, and to publish empirical and applied research on issues relating to Economics and International Business.

Journal of Advanced Research in Economics and International Business, starting with its first issue, will be indexed in <u>RePEC</u>, <u>CEEOL</u>, <u>ProQuest</u>, and <u>EBSCO</u>databases.

Web: http://www.asers.eu/journals/jareib

E-mail:jareib.asers@gmail.com



Editor in Chief: PhD Andy Ştefănescu University of Craiova, Faculty of Economics and Business Administration, Romania Co-Editor: PhD Rajesh K. Pillania Management Development Institute, India

The Journal aims to serve researchers, scholars through prompt publications of significant advances in any branch of management science, and to provide a forum for the reporting and discussion of news and issues concerning management science.

Journal of Advanced Research in Management is indexed in <u>RePEC</u>, <u>IndexCopernicus</u>, <u>ProQuest</u>, <u>CEEOL,CABELL'S Directories</u> and <u>EBSCO</u> databases.

Web: <u>http://www.asers.eu/journals/jarm</u> Email: <u>jarm@asers.eu</u>

Journal of Advanced Research in Law and Economics



Editor in Chief: PhD MădălinaCONSTANTINESCU SpiruHaret University, Romania

Co-Editors: PhD Russell PITTMAN International Technical Assistance Economic Analysis Group Antitrust Division, SUA

Journal of Advanced Research in Law

and Economics provides readers with high quality and empirical research in law and economics. The *Journal* publishes analytical studies on the impact of legal interventions into economic processes by legislators, courts and regulatory agencies. Finally, important developments and topics in the analysis of law and economics will be documented and examined in special issues dedicated to that subject. The *Journal* is edited for readability; lawyers and economists, scholars and specialized practitioners count among its readers.

Journal of Advanced Research in Law and Economics, starting with its first issue, will be indexed in <u>SCOPUS</u>, <u>RePEC</u>, <u>EconLit</u>, <u>CEEOL</u>, <u>ProQuest</u>, <u>CabellDirectories</u> and <u>EBSCO</u> databases.

Web: http://www.asers.eu/Journals/jarle

E-mail: jarle@asers.eu

Journal of Advanced Research in Organizational Psychology



Editor in Chief: PhD PompiliuCONSTANTINESCU

Academy of Economic Studies, Romania andAssociation for Sustainable Education, Research and Science

Co-Editor: AndraMihaelaPALOS

Association for Sustainable Education, Research and Science

The aims and scope of the *Journal of Advanced Research in Organizational Psychology* is to provide its readers with up-to-date studies and current trends in the field of organizational psychology. The *Journal* will host articles dedicated to the study of inner-group psychology and the social dynamics existing in the organization today. The contents of the *Journal* can be useful to students and practitioners alike, as they will provide insight to new theories, thoughts and perspectives with regards to the field of organizational psychology.

Journal of Advanced Research in Organizational Psychology, starting with its first issue, will be indexed in <u>RePEC</u>, and <u>CEEOL</u>databases.

Web: http://www.asers.eu/Journals/jarop

E-mail:jarop@gmail.com

Journal of Advanced Studies in Finance



Editor in Chief: PhD. Laura GAVRILĂ (formerly ŞTEFĂNESCU) SpiruHaret University, Romania

Co-Editor: PhD RajmundMIRDALA Technical University of Kosice, Faculty of Economics,Slovak Republic

The *Journal* aims to publish empirical and theoretical articles which make significant contributions in all areas of finance, such as: asset pricing, corporate finance, banking and market microstructure, but also newly developing fields such as law and finance, behavioral finance, and experimental finance. The *Journal* serves as a focal point for communication and debates for its contributors for the better dissemination of information and knowledge on a global scale.

Journal of Advanced Studies in Finance is indexed in <u>EconLit</u>, <u>IndexCopernicus</u>, <u>RePEC</u>, <u>CEEOL</u>, <u>ProQuest,CabellDirectories</u>and<u>EBSCO</u> databases.

Web: http://www.asers.eu/Journals/jasf

E-mail: jasf@asers.eu

Journal of Environmental Management and Tourism



Editor in Chief: PhD Ramona PÎRVU University of Craiova, Faculty of Economics and Business Administration

Journal of Environmental Management and Tourism will publish original research and seeks to cover a wide range of topics regarding environmental management and engineering, environmental management and health, environmental chemistry, environmental protection technologies (water, air, soil), at-source pollution reduction and waste minimization, energy and environment, modelling, simulation and optimization for environmental protection; environmental biotechnology, environmental education and sustainable development, environmental strategies and policies, etc.

Journal of Environmental Management and Tourism is indexed in <u>RePEC</u>, <u>IndexCopernicus</u>, <u>CEEOL</u>, <u>ProQuest</u>, <u>EBSCO</u>and<u>Cabell Directories</u>databases.

Web: http://www.asers.eu/Journals/jemt

E-mail: jemt@asers.eu

Journal of Research in Educational Sciences



Editor in Chief: PhD Laura UNGUREANU SpiruHaret University, Romania

Assistant Editor: Andra-Mihaela PALOŞ

Association for Sustainable Education, Research and Science

The Journal is designed to promote scholarly

thought in the field of education with the clearly mission

to provide an interdisciplinary forum for discussions and debates about education's most vital issues. We intend to publish papers that contribute to the expanding boundaries of knowledge in education and are focusing on research, theory, current issues and applied practice in this area.

Journal of Research in Educational Sciences is indexed in <u>RePEC</u>, <u>IndexCopernicus</u>, <u>CEEOL</u>, <u>ProQuest,CabellDirectories</u>and<u>EBSCO</u> databases.

Web: http://www.asers.eu/Journals/jres

E-mail: jres@asers.eu

Theoretical and Practical Research in Economic Fields



Editor in Chief: PhD Laura UNGUREANU SpiruHaret University, Romania

Co-Editor: PhD Ivan KITOV Russian Academy of Sciences, **Russia**

Theoretical and Practical Research in Economic Fields publishes original articles in all branches of economics - theoretical and empirical, abstract and applied, providing wide-ranging coverage across the subject area. The *Journal* promotes research that aims to unify the theoretical-quantitative and the empirical-quantitative approach to the economic problems that can be solved through constructive and rigorous thinking.

Theoretical and Practical Research in Economic Fields is indexed in <u>RePEC</u>, <u>EconLit</u>, <u>IndexCopernicus</u>, <u>CEEOL</u>, <u>ProQuest, CabellDirectories</u> and <u>EBSCO</u> databases.

Web: http://www.asers.eu/Journals/tpref

E-mail: tpref@asers.eu

Active Calls for Book Chapters Contributions!



STARTUPS AND STARTUP ECOSYSTEMS. THEORIES, MODELS AND SUCCESSFUL CASES IN THE MEDITERRANEAN AREA

A bookeditedbyPhDProfessor Domenico NICOLÒ, Mediterranean University of Reggio Calabria, Italy <u>domenico.nicolo@unirc.it</u>

Tobepublishedby ASERS Publishing in CD-ROM format with ISBN and DOI.

Deadline for submission30thOctober, 2014 <u>Download Call for Book Chapters</u>: http://www.asers.eu/asers-publishing/books.html

Conferences Proceedings ...



Proceedings of the ASERS First on-line Conference on Global Trends in Finance

Coordinator: PhD. Rajmund MIRDALA

Format: 17cm x 24cm ISBN: 978-606-92386-8-4



Proceedings of the ASERS First on-line Conference on World's Economies in and after Crisis: Challenges, Threats and Opportunities

Coordinator: PhD. Laura ŞTEFĂNESCU

Format: 17cm x 24cm ISBN: 978-606-92386-0-8



Proceedings of the ASERS First on-line Conference on Competitiveness and Economic Development: Challenges, Goals and Means in a Knowledge based Society

Coordinator: PhD. Andy STEFĂNESCU

Format: 17cm x 24cm ISBN: 978-606-92386-4-6

ASERS



Web: www.asers.eu URL: http://www.asers.eu/asers-publishing E-mail:asers@asers.eu ISSN 2068 - 7710 Journal DOI: http://dx.doi.org/10.14505/tpref Journal'sIssue DOI: http://dx.doi.org/10.14505/tpref.v6.1(11).00